palgrave>pivot

16

16

116

(

 $\left(\right)$

110

111

11

NGOS, SOCIAL CAPITAL AND COMMUNITY EMPOWERMENT IN BANGLADESH

M. Rezaul Islam

10

NGOs, Social Capital and Community Empowerment in Bangladesh M. Rezaul Islam

NGOs, Social Capital and Community Empowerment in Bangladesh



M. Rezaul Islam University of Dhaka Dhaka, Bangladesh

ISBN 978-981-10-1746-9 ISBN 978-981-10-1747-6 (eBook) DOI 10.1007/978-981-10-1747-6

Library of Congress Control Number: 2016950715

© The Editor(s) (if applicable) and The Author(s) 2016

This book was advertised with a copyright holder in the name of the publisher in error, whereas the author holds the copyright.

This work is subject to copyright. All rights are solely and exclusively licensed by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use. The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made.

Cover illustration: Mono Circles © John Rawsterne/patternhead.com

Printed on acid-free paper

This Palgrave Macmillan imprint is published by Springer Nature The registered company is Springer Nature Singapore Pte Ltd. The registered company address is: 152 Beach Road, #22-06/08 Gateway East, Singapore 189721, Singapore For (late) Rejia Begum, my mother

ACKNOWLEDGEMENTS

I would like to acknowledge the love and support of my family over the years-my wife, parents and children. It is they, above all other, who have made possible my involvement in such academic work. Far too many people to mention individually assisted my work in so many ways. They all have my sincere gratitude. In particular, I would like to thank to my PhD supervisor Professor William John Morgan (University of Nottingham). I have also got direct and indirect supports from some of my colleagues and international scholars: Professor Alan Rogers (University of East Anglia), Professor P. Sillitoe (University of Durham), Professor D. Lewis (LSE), Professor A. Mia (University of Development Alternative), Professor S. Marshall (University of West Indies), Miss Sharmin Rahman (London Kings College), Dr M. M. Ahmad (AIT), Dr J. Devine and Dr S. White (University of Bath), Professor J. Buckland (University of Winnipeg), Dr K. E. Jackson (University of Auckland), Dr Richard Coll (Vice Chancellor, University of Fiji), Dr Peter Gates and Dr Mark Lymbery (University of Nottingham), Dr M. Emranul Haque (University of Manchester), Dr Wahiduzzaman and Mr Fozle Khoda (University of Dhaka), Dr Dennis Banda (University of Zambia), Professor Md. Ismail Hossain and Mr Shofiqur Rahman Chowdhury (Shahjalal University of Science & Technology), Mr Mohsin Reza (Jagannath University) and Mr Mayeen Uddin (my ex student). My special thanks go to two distinguished scholars: Professor Robert Putnam (Peter and Isabel Malkin Professor of Public Policy at the Harvard University) and Dr Glenn Laverack (University of Southern Denmark).

CONTENTS

1	Introduction: NGOs, Social Capital and Community Empowerment		1	
2	NG	Os in the Development Discourse	7	
	2.1	NGOs: Concepts and Background	8	
	2.2	NGOs for What and for Whom	13	
	2.3	NGOs in Global Development Discourse	14	
	2.4	Conclusions	19	
3	Ban	gladesh and the Emergence of NGOs for Development	21	
	3.1	Country Context: Bangladesh	22	
	3.2	Origin of NGOs in Bangladesh: Socio-Economic and		
		Political Realities	23	
	3.3	NGOs' Activities in Bangladesh: Hopes for Development	29	
	3.4		36	
	3.5	Conclusions	37	
	Notes		37	
4	Globalisation and Indigenous Knowledge: NGOs' Role for			
	Development			
	4.1	Globalisation: Challenges and Prospects for NGOs in		
		Development	40	
	4.2	Indigenous Knowledge: A Local Development Intervention		
		for NGOs	43	

ix

	4.3	Indigenous Knowledge Versus Global Knowledge: NGOs' Nexus Role	46
	4.4	Conclusions	48
	Not		49
=	NC	On' Franch in Special Comital and Community	
5		Os' Facet in Social Capital and Community powerment	51
	5.1	Social Capital: A Widening Field for NGOs' Activities	51
	5.1	Meanings and Aspects of Social Capital	52
	5.2 5.3		55 59
			- 59 - 61
	5.4		01
	5.5		63
	- /	Paradigm for NGOs' Development Activities	
	5.6	Conclusions	69
6	NGOs' Capacity for Social Capital in Bangladesh		
	6.1	Participatory Planning: Create Routes for Social Capital	73
	6.2		
		Mutual Benefits	75
	6.3		
		Examples of Collective Actions	79
	6.4	IGPs Versus Social Trust	81
		Conclusions	84
	Not		85
7	NG	Os and Community Empowerment in Bangladesh	87
`	7.1	Community Participation Versus Community	07
	/.1	Empowerment	88
	7.2	Use of Scientific Tools: Increase Problem Assessment Capacities	90
	7.3		90
	7.5	Quality	91
	7.4		
		Mobilisation	93
	7.5		95
	7.6	Communication and Knowledge Sharing: Strengthen	20
		Links with Outside Agents and Improve Critical	
		Awareness About Needs	98

133

7.7 Decentralisea	l and Autonomous Management	100
7.8 Conclusions	C C	102
Notes		103

8 NGOs' Limitations and Challenges for Social Capital and Community Empowerment: Local Dynamics and Global Pressures

Glo	bal Pressures	105
8.1	Local Peoples' Poor Socio-Economic Conditions	107
8.2	NGOs' Conventional Role: Questions of Accountability	109
8.3	Social Capital Is Meaningless Without the Support of	
	Other Capitals	113
8.4	Social Networks and Empowerment (Politically) as	
	Negative Features for Development	115
8.5	Lack of Social Trust	117
8.6	External Intervenors, Donor Dependency and	
	Organisational Autonomy	120
8.7	Religious Leaders Frustrate Many Initiatives	123
8.8	Local Power Structure, Inaccessible Market and Lack	
	of GOs' Help: Evolving New Initiatives	124
8.9	Centralisation in Decentralised Management	126
8.10	Limitations of Social Capital and Community	
	Empowerment Approaches	127
8.11	Conclusions	130
Not	es	131

References

Index	153	3
Index	153	3

Introduction: NGOs, Social Capital and Community Empowerment

Abstract This introductory chapter provides a brief outline about the capacity of the non-governmental organizations (NGOs) towards social capital and community empowerment. This chapter briefly explains the three concepts used in this book e.g., NGO, social capital and community empowerment, and then explains NGOs' position in social capital and community empowerment. Finally, this chapter provides the chapter outline of this book.

Keywords NGO · Social capital · Community employment

Non-governmental organizations (NGOs) become an important sector for development in the developing countries like Bangladesh. Due to its socio-economic disadvantages and the lack of governmental supports, NGOs' development initiatives are considered as one of the best alternatives for all aspects of community well-being (economic, social, environmental and cultural) in the country. This becomes a very popular initiative, whereby the community members come together to take a collective action and generate solutions to their common problems. It is considered as a 'grass-roots' process by which a community becomes more responsible; organise and plan together; develop healthy options; empower themselves; reduce ignorance, poverty and suffering; create employment and economic opportunities; and achieve social, economic, cultural and environmental goals. In the wake of 'flexible accumulation' and intense market competitions under the conditions of neo-liberal globalisation, the international NGOs are playing a significant role in funding and formulating development frameworks in the development activities in many developing countries like Bangladesh. It is said that almost 90 % of the NGOs' funding is coming from foreign donations. As a result, NGOs and donors increasingly face the twin challenges of demonstrating effectiveness in their work and accountability in their relationships with various stakeholders. Donors, on the one hand, are particularly concerned about accountability of NGOs in the efficient and effective delivery of services. NGOs, on the other hand, are often concerned that accountability to donors can overshadow and overwhelm their accountability to communities and to their own missions.

This book has focused on NGOs' development interventions in two development approaches such as social capital and community empowerment. The term 'social capital' has been the subject of great interest and debate in the development literature. It has been elaborated so widely that it is very difficult to explain its theoretical context briefly. NGOs' contribution for social capital is relatively a new area in development studies. Yet both governments and NGOs' social capital formation in Bangladesh is little known (Mondal 2000; Islam and Morgan 2012a; Islam 2015a, 2015b). The theoretical framework of social capital in this book is based on four core elements provided by Robert Putnam (Putnam 1993, 1995). These elements are collective action, social trust, coordination and cooperation of mutual benefits, and sharing norms and values. On the other hand, the concept of 'community empowerment' is one of growing interest in development discourse. The theoretical framework of community empowerment of this book is based on the nine domains provided by Glenn Laverack (Laverack 2006). Community empowerment is advocated by prominent agents in economic and social development, and especially by NGOs. Empowerment enables people to organise and influence change based on their access to knowledge, to political processes and to financial, social and natural resources.

To consider the terminological explanations, we can see that both concepts are very crucial in NGOs' development initiatives. Much ink has been spilt on the question of NGOs, social capital and community empowerment towards development—albeit separately. However, we still lack a complete and coherent account of the complex nexus between of NGOs, social capital and community empowerment. What is the practicality of NGOs' necessity in developing countries like Bangladesh for development? How are the NGOs' development interventions positioned in the dynamisms of global pressures and local needs? What are the merits of social capital and community empowerment in NGOs' activities? What are the capacities of NGOs in social capital and community empowerment? What is the important role that can be defined of the NGOs' social capital and community empowerment towards development? What challenges and limitations are the NGOs heading? This book provides a comprehensive understanding of NGOs' real capacities with focus on Bangladesh to illustrate a global problem. I believe that while the book focuses on Bangladesh, the implications are international in scope.

As we turn the millennium, a concern is growing again among a new generation of students, but they confront a literature on the NGOs, social capital and community empowerment, which is largely descriptive, dated and does not contain the complex nexus embedded in the flexible accumulation of and powerlessness of the developing countries under neo-liberal globalisation. While teaching courses related to NGOs and social development, we have found that existing books have a few compilations with uneven and disjointed chapters. We have used several books in courses and assembled clippings in course-packs, but the students find themselves wishing to know more about the complex nexus between NGOs, social capital and community empowerment, a more coherent structure with which to understand them. This book will meet the growing needs of this new generation of students, as well as academics, conscious citizens and policymakers.

Chapter 2 provides the background information, for example, country context of Bangladesh, the socio-economic and political realities of the origin of NGOs, their activities and why Bangladesh bothers NGOs. This chapter opens a discussion about NGOs' significant position and the role in development discourse. The massive expansion of the NGO sector in both the north and the south is widely recognised. In the 1980s, the decade of neo-liberalism, the market was seen as the panacea of development discourse and practice. Since the 1990s, there has been recognition that liberalisation alone is an inadequate response to the social and economic development of developing countries. However, NGOs' corporate responsibilities have been widened largely. In many developing and underdeveloped countries, NGOs were considered as an effective

alternative development agent in international development arena. The aim of this chapter is to address the key issues that emerged in the 1980s and 1990s on the theme of development and NGOs; examine theoretical and policy debates on NGOs, states, civil society and democracy; and critically assess the roles and relationships of development NGOs and international organisations.

Chapter 3 provides the role of NGOs for development with globalisation and indigenous knowledge. First, the chapter provides a discussion on the emergence of NGOs for development. It highlights the main parameters of NGOs. The chapter attempts to integrate various relevant debates of NGOs' activities in the socio-economic, political and cultural development paradigms. It is found that the NGOs have abrasive relationships with state and donors, on the one hand, and global versus local development debates, on the other. The chapter shows how the development challenges might be solved through consulting both local and global knowledge approaches, so that the NGOs can achieve participatory-oriented development interventions according to the choices of the rural mass people in Bangladesh.

Chapter 4 presents a comparative discussion of indigenous and global knowledge approaches within the globalisation process. The chapter highlights and explains each of these concepts, and attempts to establish a significant link with a new role of NGOs for development. The main purpose is to find out the gaps, for example, what are the general explanations of these concepts, and which are effective for Bangladesh?

Chapter 5 elucidates the complex intersection between social capital and community empowerment towards NGOs' development initiatives. The chapter provides a solid theoretical discussion about different aspects of development and then examines how social capital and community empowerment are widening where NGOs' role is significant.

Chapter 6 discusses NGOs' capacity for social capital development with the evidence of two NGOs, for example, Practical Action Bangladesh (PAB) and Proshika in Bangladesh with two indigenous communities, for example, blacksmith and goldsmith. This chapter explores NGOs' capacities on the five main elements of social capital such as social network, social trust, sharing norms and values, collective actions and mutual benefits. The main argument is whether the NGOs' capacities were elusive.

Chapter 7 is based on the evidence of the two NGOs, which is linked with Chap. 6. The chapter considers the role of NGOs in terms of their

capacity as agents of community empowerment in Bangladesh. The chapter then investigates the application of community empowerment domains.

Chapter 8, the final chapter of the book, looks at the major limitations and challenges of NGOs' capacities for social capital and community empowerment approaches. This chapter highlights the evidence of two NGOs that are discussed in Chaps. 6 and 7. A growing number of critical assessments suggest that the operational impact of NGOs in socio-economic development is less than claimed. NGOs may pursue a 'service delivery' paradigm, where the provision of services is strictly separated from engagement in the broader polity. In reference to this connection, NGOs become providers of goods to poor 'consumers', rather than 'facilitators' of collective action and empowerment. This growing predominance of service delivery programmes among NGOs is problematic for the goals of empowerment and poverty alleviation. Based on the evidence and literature survey, this chapter shows that the NGOs could not achieve their development targets because of the local dynamics and global pressures. This chapter analyses a number of local dynamisms such as local context, political issues, NGOs' conventional role, NGOs' monolithic development approach, lack of accountability and the global pressures such as lack of local funding and donor dependency, global development frameworks, decentralised management system.

NGOs in the Development Discourse

Abstract This chapter opens a discussion on the significant position of non-governmental organisations (NGOs) and their role in development discourse. The massive expansion of the NGO sector in both the north and the south is widely recognised in the present time. In the 1980s, the decade of neo-liberalism, the market was seen as the panacea of development and the NGOs were to play an increasingly significant role in development discourse and practice. Since the 1990s, there has been recognition that liberalisation alone is an inadequate response to the social and economic development of developing countries. However, NGOs' corporate responsibilities have been widened largely. In many developing and underdeveloped countries, NGOs are considered as an effective alternative development agent in international development arena. The aim of this chapter is to address the key issues that emerged in the 1980s and 1990s on the theme of development and NGOs, examine theoretical and policy debates on NGOs and critically assess the ongoing debate about the benefits of participation in development.

Keywords NGOs \cdot Non-profit organisations \cdot Development discourse \cdot Socio-economic development \cdot Developing countries \cdot Lack of accountability \cdot NGO management

2.1 NGOs: Concepts and Background

The literature gives a certain level of confusion about the definitions of the term non-governmental organisations (NGOs). Willetts (2002) argues that there is no generally accepted definition of an NGO and the term carries different connotations in different circumstances as many diverse types of bodies are now described as being NGOs. Lewis (n.d.) mentions that NGOs' definition remains contested. Ulleberg (2009) argues that it is a nearly impossible task to enumerate the various NGO characteristics when it comes to their aims, strategies, resources, target groups, tools, effectiveness, impact and sustainability. Lewis (n.d.) further mentions that the world of NGOs contains a bewildering variety of labels. While the term is widely used, there are also many other overlapping terms used such as 'non-profit', 'voluntary' and 'civil society' organisations. However, it must be independent from the direct control of any government. According to Turner and Hulme (1997, 200), 'NGOs are generally registered organizations, community groups, professional associations, trade unions, cooperate charity organizations whose aim is to improve the wellbeing of their members and of those areas in which they exist'. It covers a range of organisations within civil society, from political action groups to sports clubs. To consider this, it can be argued that all NGOs can be regarded as civil society organisations though not all civil society organisations are NGOs. NGOs take different forms and play different roles in different continents, with the NGO sector being most developed in Latin America and parts of Asia. Lewis and Kanji (2009) argue that the roots of NGOs are different according to the geographical and historical context. Cleary (1997) argues that although there is contestation of the definition of an NGO, it is widely accepted that these are organisations which pursue activities to relieve the suffering, promote interests of the poor, protect the environment, provide basic social services and undertake community development.

The World Bank defines NGOs as 'private organisations' that pursue activities to relieve suffering, promote the interests of the poor, protect the environment, provide basic social services or undertake community development (Risal 2014). The Norwegian Agency for Development Cooperation (NORAD 2004, 6) defines development-oriented NGOs as organisations that 'attempt to improve social, economic and productive conditions and are found both as small community-based organisations (CBOs) at village and district levels, and as large professional development

agencies at state or national level'. NGO is a non-profit, voluntary citizens' group, which is organised on a local, national or international level. An NGO will not be constituted as a political party; it will be non-profit-making, and it will not be a criminal group, in particular, it will be non-violent (Willetts 2002). The United States Agency for International Development understands NGOs as a wide range of local organisations, particularly in the developing countries, operating various development activities largely with the financial, technical and material assistance of wealthy countries (Schuller 2007).

The nature and structure of NGOs vary considerably. Many NGOs are now recognised as grass-roots organisations (locally based groups) or community-based organisations. Both are active at the national or even the global level (Mostashari 2005). Lewis (n.d.) distinguishes between 'northern NGO' (NNGO) and 'southern NGO' (SNGO). The NNGO refers to organisations whose origins lie in the industrialised countries, while the SNGO refers to organisations from the less developed areas of the world. Lewis further mentions that another key distinction is between membership forms of NGO, such as community-based organisations or people's organisations, and intermediary forms of NGO that work from outside with communities, sometimes termed grass-roots support organisations. There are also numerous examples of bogus NGOs, such as those established as fronts by government (government-organised NGOs) or 'briefcase' NGOs set-up by individuals for personal gain (Lewis, n.d.).

Lewis and Kanji (2009, 10) mentioned NGO as a 'third-sector organisation'. According to them, a third-sector organisation has the following five key characteristics: (1) it is formal, that is, the organisation is institutionalised in that it has regular meetings, office bearers and some organisational permanence; (2) it is private in that it is institutionally separate from government, though it may receive some support from government; (3) it is non-profit distributing, and if a financial surplus is generated it does not accrue to owners or directors (often termed the 'non-distribution constraint'); (4) it is self-governing and therefore able to control and manage its own affairs; and (5) finally, it is voluntary, and even if it does not use volunteer staff as such, there is at least some degree of voluntary participation in the conduct or management of the organisation, such as in the form of a voluntary board of governors. A more common-sense definition focuses instead on the idea that NGOs are organisations concerned with the promotion of social, political or economic change—an agenda that is usually associated with the concept of 'development'.

The term NGO came into currency in 1945 because of the need for the United Nations (UN) to differentiate in its Charter between participation rights for intergovernmental specialised agencies and those for international private organisations. According to the UN, all types of private bodies can be recognised as NGOs, but should be independent from government control, not seeking to challenge governments either as a political party or by a narrow focus on human rights, non-profit-making and non-criminal (Willetts 2002). We can see that due to the weakening of ideological political parties and the retreat of the state from providing social entitlements and services, in response to structural adjustment policies imposed on most Third World countries by the World Bank and the International Monetary Fund (IMF), and under the pressure of neo-liberal reforms (Edwards and Hulme 1995). Some see NGOs as the product of neo-liberal policies, as financially dependent on neo-liberal sources and as directly involved in competing with sociopolitical movements for the allegiance of local leaders and activist communities (Petras 1997). Others see them as mechanisms organised for the 'creation of civil society by external intervention' (Sampson 1996, 121–122). In this case, the conflict between NGO and civil society becomes an important issue in development history. A number of studies emphasise the negative impact of NGOs on social movements (Petras 1997) and explore the impact of what has been termed 'NGOisation' on mobilisation and social action (Jad 2007).

Ebrahim (2001) sees NGOs development from the shifts in development thought, policy and practice over time. He mentions that beginning with the establishment of the Bretton Woods institutions after World War II, the objective of NGOs' activities changed dramatically. The development programmes of governments in the south and of international organisations and aid agencies in the north shifted from an emphasis on large-scale infrastructure, industry and agriculture in the 1950s and 1960s, to a 'basic needs' focus on individuals and families in the 1970s. The NGOs' functions concentrated on the participation, sustainable development and gender equity in the 1980s, more recently, on issues of economic liberalisation and civil society (Fisher 1997). Willetts (2002) and Nelson (2002) explain the origin of NGOs with the role of UN, World Bank and the USA (Boxes 2.1 and 2.2).

Box 2.1 The origin of NGOs with the role of the UN and the USA

The term NGO was not in general currency before the UN was formed. When 132 international NGOs decided to cooperate with each other in 1910, they did so under the label the Union of International Associations. The League of Nations officially referred to its 'liaison with private organisations', while many of these bodies at that time called themselves international institutes, international unions or simply international organisations. The first draft of the UN Charter did not make any mention of maintaining cooperation with private bodies. A variety of groups, mainly but not solely from the USA, lobbied to rectify this at the San Francisco conference, which established the UN in 1945. Not only did they succeed in introducing a provision for strengthening and formalising the relations with private organisations previously maintained by the League, they greatly enhanced the UN's role in economic and social issues and upgraded the status of the Economic and Social Council (ECOSOC) to a 'principal organ' of the UN. To clarify matters, new terminology was introduced to cover ECOSOC's relationship with two types of international organisations. Under Article 70, 'specialised agencies, established by intergovernmental agreement' could 'participate without a vote in its deliberations', while under Article 71 'non-governmental organisations' could have 'suitable arrangements for consultation'. Thus, 'specialised agencies' and 'NGOs' became technical UN jargon. Unlike much UN jargon, the term NGO passed into popular usage, particularly from the early 1970s onwards. Source: Willetts (2002).

Box 2.2 The origin of NGOs with the role of World Bank

Between 1980 and the present, NGOs developed and learned a set of political strategies that are well adapted to influencing development and environmental policy issues at the World Bank. Here, their approach made consistent use of eight political strategies shaped by the World Bank's institutional, professional and political environment. The strategies are built around network advocacy, internationalising local issues by focusing on the World Bank's role in domestic policy, and drawing heavily on the support of the US government to reform and regulate World Bank policy. By the late 1990s, however, many of the same organisations involved in advocacy at the World Bank had become engaged in a broader set of economic policy and trade issues. These issues have required new and different political strategies, forcing NGOs to relearn—even to 'unlearn'—political lessons gained in work on the World Bank. The political and institutional factors that require this political retooling include both the political environment of trade and financial policymaking and the need to relate to growing 'anti-globalisation' social movements whose political approaches differ sharply from those of most NGOs.

Source: Nelson (2002).

Lewis (n.d.) argues that NGOs have existed in various forms for centuries, but they rose to high prominence in international development and increased their numbers dramatically in the 1980s and 1990s. He added that NGOs had been active at the international level since the eighteenth century in Western countries, when national-level issue-based organisations focused on the abolition of the slave trade and movements for peace. By the start of the twentieth century, there were NGO associations promoting their identities and agendas at national and international levels. It is widely accepted that the scope of NGOs working in social development has grown exponentially in the last several decades. Banks and Hulme (2012) classified the trend of NGOs into five different times on the basis of the rising prominence such as until the late 1970s, the late 1970s to 1980s, the late 1990s, 2000s and 2010s (Box 2.3).

Box 2.3 The rising prominence of NGOs

Until the late 1970s: A limited number of small NGOs receiving little external support constitute the NGO sector. Most are northern-based with a southern presence, often based on religious assistance and/or in short-term relief.

Late 1970s to 1980s: 'The NGO decade' takes place amidst the Western pursuit of neo-liberal agendas, with NGOs emerging as a promising development alternative.

Late 1990s: Alongside emergence of the good governance agenda, the first concerns surrounding NGOs take off alongside a focus back on the role of the state.

2000s: A new international aid regime promises greater consultation and focus on non-growth factors. NGOs with their people-centred, rights-based and grass-roots-driven approaches are well suited to continue riding the NGO wave.

2010s: With persistent concerns of NGOs remaining unaddressed and recognition of their limited success in advocacy and empowerment, there is increasing recognition that NGOs are only one sector within broader civil society and they must reorient themselves with their grass roots.

Source: Banks and Hulme (2012).

2.2 NGOs for What and for Whom

Most of the NGOs are working towards the development and wellbeing for the marginalised and disadvantaged people. Hailey and James (2004) argue that NGOs have a social change mission and specifically work with the vulnerable people and marginalised groups who have often been ignored or overlooked by government services and the mainstream social services. Murtaza (2012) states that NGOs were recognised as successful agents over the last two decades in helping improve the lives of marginalised communities in developing countries through programme and advocacy work. Bromideh (2011) thinks that the NGOs have been pivotal in providing food to the most marginalised people. In developing countries, NGOs often provide essential services that in developed countries governmental agencies or institutions would provide. In general, NGOs are also the source and centre of social justice to the marginalised members of society in developing countries or less developed (failed states). Hailey and James (2004) consider NGOs as 'intermediary organisations' which make a bridge between the donors and beneficiaries. This sector constantly responds to multiple constituencies or clients. There is also a new awareness of the important role of leadership in the development of the NGO community and the powerful influence of individual leaders in shaping the destiny of local NGOs.

NGOs sometimes have been criticised robustly due to their commercial attitude. This sector is now aggressively pursuing profit-making business enterprise in the major economic sector. The literature shows that many NGOs in developing countries such as Bangladesh have become rentseeking organisations (Islam and Morgan 2012a) because they charge the highest interest rate (20-30%) on loans given to the poor, while they themselves receive massive financial assistance from governments and foreign donors at rates of 3% (Molla et al. 2008). Due to NGOs' multiple objectives and diversity of work, it is difficult to know precisely how many NGOs are in the world or even in a country. Some authors put the figure at a million organisations, if both formal and informal organisations are included, while the number of registered NGOs receiving international aid is probably closer to 'a few hundred thousand' (Lewis n.d.). The UN estimates that there were about 35,000 large established NGOs in 2000. There are no accurate figures available for resources that NGOs receive from aid, contracts and private donations. In 2004, it was estimated that NGOs were responsible for about US\$23 billion of total aid money or approximately one-third of total Official Development Assistance (ODA) (Riddell 2007, 53).

2.3 NGOs in Global Development Discourse

The role of NGOs in development discourse has been discussed widely. Here, development discourse refers not only to how development is described or talked about, but to also how it is thought about (e.g., its underlying assumptions) and practised (Ebrahim 2001, 80). Perceptions of NGOs in development have been discussed in both positive and negative ways. Some have argued that they are flexible, innovative and efficient vehicles for the delivery of basic services and poverty alleviation, and that they reach poor communities and remote areas at a lower cost than governments (Islam and Morgan 2012a, 2012b; Islam 2014a, 2014b). Vivian (1994) explains NGOs' role as the 'magic bullets' of development. NGOs identify genuine local needs, promote participation and transfer appropriate technologies. Others have argued that most NGOs are started and controlled by charismatic individuals who necessarily limit participatory decision-making (Wood 1997), and that any evidence of NGOs' effectiveness remains weak (Edwards and Hulme 1995). Khandker

(1998) found that in South Asia, Grameen Bank built a worldwide reputation for its work and now operates its activities in a number of other countries. They claim that its credit programmes for poor rural women in Bangladesh attain repayment rates consistently over 90% and that they are entirely self-financed.

Over the past 20 years, there has been a dramatic shift in the provision of basic services in many developing countries (Lewis n.d.). NGOs are now recognised as key third-sector actors on the landscapes of development, human rights, humanitarian action, environment and many other areas of public action. NGOs' social service activities such as services in health care, education and rural credit bring many positive changes among the vulnerable and disadvantaged poor people. This is attributable, in part, to the increasing role of NGOs in development assistance. Whereas 20% of the World Bank-financed projects approved in 1989 included some NGO or CBO involvement, that figure was 47% in 1997. A survey of the World Bank projects approved between 1985 and 1997 found that the institution channelled \$1.3 billion of development assistance through NGOs in seven countries alone (Fruttero and Gauri 2005). NGOs' contributions towards development have been highlighted in some other ways such as building good networking, transforming knowledge, campaigning policy advocacy and aiding from international organisations. Roberts et al. (2005) state that people benefit from the links among NGOs and the transfer of knowledge across these relationships. NGO network 'groups of three or more civil society organizations or NGOs that pursue shared purpose' (Ashman and Sugawata 2013, 394) have become key instruments used by NGOs. On the other hand, NGOs are found as high-profile actors in the field of international development, both as providers of services to vulnerable individuals and communities, and as campaigning policy advocates. NGOs could help scale-up international aid efforts by raising financial resources from private donors. The mobilisation of private donations has attracted little attention in the recent literature, even though NGOs are widely considered to play an increasingly important role in supplementing official aid resources (Lewis and Kanji 2009). For instance, McCleary and Barro (2008) reported that more than 40% of development aid by the USA is channelled through NGOs. From the perspective of individual NGOs, whether additional aid funds can be mobilised mainly depends on how private donations react to the 'marketing' efforts of NGOs. In particular, NGOs attempt to attract higher donations by strengthening their development orientation, reducing

unproductive overheads, better targeting of aid activities and offering options to earmark private donations (Nunnenkamp and Öhler 2012).

Nowadays, NGOs have emerged as an alternative development discourse and contributed to state-provided public services and processes through information provision and tactical support, making services and policies more relevant, responsive and effective. The donor-funded partnership offers much promise in supporting these contributions and improving the relationship between governments and NGOs. NGOs are also active in a wide range of other specialised roles such as democracy building, conflict resolution, human rights work, cultural preservation, environmental activism, policy analysis, research and information provi-sion (Lewis n.d.). Cernea (1988, 8) has argued that NGOs embodied 'a philosophy that recognizes the centrality of people in development policies', and gave them certain 'comparative advantages' over government and public sector. For example, Fruttero and Gauri (2005) stated that with multifaceted development interventions 'Bangladesh Rural Advancement Committee (BRAC) strives to bring about changes in the quality of life of people'. Two Bangladeshi NGOs in their brochures reveal that Grameen Bank provides credit to the poorest of the poor in rural Bangladesh without any collateral. At Grameen Bank, credit is a costeffective weapon to fight poverty and it serves as a catalyst in the overall socio-economic development; and 'poverty reduction programmes undertaken so far have bypassed many of the poorest. In this context, one of BRAC's focuses is the ultra-poor' (Lewis and Kanji 2009).

NGOs play an increasingly important role in the delivery of public services in developing countries, but little systematic evidence is available about their strategic choices (Fruttero and Gauri 2005). The fragile and fragmented character of formal political institutions, coupled with the availability of donor aid, led to the emergence during the 1990s of thousands of NGOs in Bangladesh (Lewis 2008). This sector is particularly high public profile and occupies centre stage in the development landscape of the country (Devine 2003, 228; Islam 2014b). This sector is one of the most sophisticated national development networks in the world. There are nearly 22,000 NGOs operating in the country, and, of these, approximately 1,250 receive foreign assistance (Devine 2003, 229). NGOs operate in more than 78% of the rural villages, and their activities directly benefit 35% of the entire population (Thornton et al. 2000). Most of the NGOs' development activities target the vulnerable and poor people. They offer a wide range of services that are fundamental to these people's daily struggle to survive.

These include credit, education, health, agricultural extension and sanitation. NGOs' objectives are to empower these people in terms of their awareness and participation in development activities and improve their financial and social capitals towards quality of life (Islam 2014b). NGOs in Bangladesh contribute some 3–4% of gross domestic product based on an estimate of the microcredit operations of NGOs. The NGOs' share of total aid to Bangladesh rose from 14% in the first half of the 1990s to 24% in 2006 (World Bank 2006). Funds committed to NGOs have doubled since 2006 (NGO Affairs Bureau of Bangladesh 2009).

NGOs have become very popular to the poor people in Bangladesh. This sector is fostering local participation as they are more locally rooted organisations, and therefore closer to marginalised people than most officials are (Islam and Morgan 2012a; Islam 2014a, 2015b). Poor people are often found to have been bypassed by existing public services since many government agencies face resource shortages and elites (Islam 2014b) often capture their decision-making processes. Many also claimed that NGOs were generally operating at a lower cost due to their use of voluntary community input. However, NGOs are seen as possessing the scope to experiment and innovate with alternative ideas and approaches to development. Some NGOs are bringing a set of new and progressive development that are beginning to capture the imagination of many development activists at this time (Lewis and Kanji 2009).

Despite these benefits, the importance attached to accountability by NGOs varies significantly. In a study of over 600 NGOs globally, Scholte (2003) found that despite their emphasis on it in principle, most NGOs had given little practical attention to the issue. On the other hand, Murtaza (2012, 111) argued that the accountability of NGOs is currently strongest to boards and donors and weakest to peer groups and communities. It also reveals that NGOs can best achieve the twin objectives of enhancing their own performance and protecting themselves from politically motivated attacks by voluntarily developing coordinated, peer-driven and community-focused accountability mechanisms. The issue of accountability in NGOs' activities is seen as very ignorant and NGOs give very little value to their work (Islam 2014b). Lee (2004) identified the following key questions to help develop NGO accountability frameworks: (1) To whom is the NGO accountable for? (3) How is the NGO accountable? Grant and Keohane (2005) identified two sets of entities which are entitled to hold

someone else to account: (1) those entities that have delegated authority or resources to another entity, and (2) those entities that are exposed to the effects of the actions of another entity. These two categories lead to several different types of accountability: (1) administrative (based on managerial oversight), (2) legal (based on contractual relationships and judicial authority), (3) moral (based on ethical values voluntarily accepted by an entity), (4) political (based on constituent relationships) and (5) professional (based on peer relationships with other similar professionals) (Murtaza 2012). One of the contested issues about development NGOs is their contribution to the political participation of the poor. While there is general consensus that in the long run disadvantaged groups can improve their lot only when they get politically organised and obtain a share of state power, there is a clear disagreement over what role development NGOs play in this regard. Many authors, for example, Fisher (1997), agree that NGOs can and do promote the political association of disadvantaged groups in an admirable way. On the other hand, critics see NGOs as instruments for maintaining elite interests and continued subordination (Islam and Morgan 2012a).

As NGOs in Bangladesh and elsewhere have grown in scale and scope, there has been a concomitant growth in debate about NGO operations, performance and accountability (Gauri and Galef 2005; Murtaza 2012). Increased dependency on foreign donors has created doubts over Bangladeshi NGOs' ability to support a civil society acceptable to the Bangladeshi population. In particular, concerns have been raised regarding NGOs' downward accountabilities, principally to beneficiaries, as NGOs are meeting what are seen as unnecessarily onerous upward accountabilities to foreign donors and to governmental regulatory authorities (Harrison 2007; Mir and Bala 2015). Islam (2014b) found that NGOs faced a number of challenges, including lack of understanding about the local context, initiatives to localise global development tools, developing partnership and space for local people in the development processes. They argued that without securing development ownership the NGOs' activities towards community empowerment of vulnerable people would not be possible.

Despite the popular discourse about the empowering role of development NGOs, little evidence can be found of development NGOs addressing the social structural causes of poverty or contributing to the political participation of the poor. Fruttero and Gauri (2005) argued that their analysis does not find strong support for the claim that NGOs were targeting poverty, at least at the level of villages in which they chose to locate. In fact, in most regressions the coefficients on indicators of need (poverty gap, literacy, percentage of landless) are not significant. This is not to suggest that beneficence plays no role in NGO activities. There are countless individuals in Bangladesh, as in other countries, for whom volunteerism and personal sacrifice, a special concern for the poor irrespective of where programmes are located, and the appeal of a higher calling are why they work in NGOs. Ulvila and Hossain (2002) reported that most of these NGOs in Bishwanathpur village in Bangladesh and in Markhu village in Nepal were found to be non-political and developmentoriented in nature. Typically, the development NGOs pay no or little attention to the oppressive power structures or to the promotion of political participation of the poor. This, in effect, easily leads to the establishment of non-governmental bureaucracies favourable to elite interests. Relations between development NGOs and local elites have been depicted as a kind of antagonism from two aspects. First, the local elites may oppose an NGO if the organisation collapses the existing balance of power by supporting elites' clients. Second, the private sector, which traditionally has been controlled by local elites in rural areas, sees in NGOs a market rival (Makita 2009; Islam and Morgan 2012a).

2.4 Conclusions

Today, NGO is a dominant ideology being pushed around the world. With huge diversity in terms of definitions, types and nature of works, NGOs have become an alternative agent in the development discourse. The scaling up of NGOs in the global arena started after World War II and became popular especially in the underdeveloped and developing countries, where the government public services are poor for the mass people. Despite some criticisms such as the lack of accountability, high administrative cost, bureaucracy, lack of people's participation, political involvement, dependency of foreign donation and lack of transparency, NGOs' contributions have become very popular due to their huge positive impacts on social awareness, education, public health, poverty alleviation, disaster mitigation, human rights, women empowerment and rural development.

Bangladesh and the Emergence of NGOs for Development

Abstract NGOs have become an important development agent in Bangladesh. This sector is increasingly becoming important because of claims that they are efficient and effective; they are innovative, flexible, independent and responsive to the problems of poor people at the grassroots level (Bagci 2007). The growth of such NGOs over the past four decades in Bangladesh has given an increasingly important role and has led them to forming a distinctive sector within civil society. Over this time, NGOs have been engaged in all sectors of social life like relief, rehabilitation, health, education, development programmes, peace, human rights, environment, and so on. They have been using finance raised from voluntary, private sources and donor agencies, and managing themselves autonomously at local, national and international levels (Bagci 2007; Gauri and Galef 2005). In the last chapter, we have seen that NGOs vary widely according to size, sector of activity, religious orientation, their functions (service providers, social movements, networks or apex organisations), their relationships to donors, their organisational sophistication and other factors. The effects of 'massive proliferation' are perhaps nowhere more evident than in Bangladesh, which has one of the largest and most sophisticated NGO sectors in the developing world (Gauri and Galef 2005). It is said that over 90 % of villages in the country had at least one NGO (Fruttero and Gauri 2005), and foreign assistance to the country channelled through NGOs has been above 10%. This chapter provides a discussion on the emergence of NGOs for development. It

© The Author(s) 2016 M.R. Islam, NGOs, Social Capital and Community Empowerment in Bangladesh, DOI 10.1007/978-981-10-1747-6_3 also highlights the main parameters of NGOs. The chapter attempts to integrate various relevant debates of NGOs' activities in the socio-economic, political and cultural development paradigms. It was found that the NGOs have abrasive relationships with state and donors, on the one hand, and global versus local development debates, on the other. The chapter shows how the development challenges might be resolved through consulting both local and global knowledge approaches, so that the NGOs can achieve participatory-oriented development interventions according to the choices of the rural people in Bangladesh. The chapter also explains the determinant factors of how NGOs have become homes for Bangladesh and why Bangladesh bothers NGOs.

Keywords Bangladesh \cdot Socio-economic \cdot Development \cdot Poverty alleviation \cdot Microcredit \cdot NGO financing \cdot Foreign donation

3.1 Country Context: Bangladesh

Bangladesh achieved its independence in 1971, but has deep-rooted traditions and cultural practices. Bangladesh took the inspiration for an independent state from the historic language movement of 21 February 1952, which remains a milestone and a red-letter day for the people of Bangladesh, who had fought long and hard to achieve democracy. The People's Republic of Bangladesh has been a parliamentary democracy since 1991. As a developing country, Bangladesh is progressing gradually, though many of the Millennium Development Goals (MDGs) are not yet achieved, partially due to its high population growth rate (World Bank 2005). As the Malthusian theory suggests, the country is struggling with the contest between population control and depleting resources. Largely, due to the limitations of the government initiatives, the NGO is increasingly becoming a necessary sector for development in Bangladesh (Devine 2003, 228; Islam and Morgan 2012a; Islam 2015a).

Etymologically, the word 'Bangladesh' is derived from the cognate 'Vanga', which was first mentioned in the Hindu scripture *Aitareya Aranyaka* (composed between 500 B.C. and A.D. 500). Another school of thought defines the term 'Vanga' as derived from *Bodo* (aborigines of Assam) words 'Bang' and 'la', which connote 'wide plains'. Bangladesh is located in south Asia, between 20°34' and 26°38' north latitude and

between 88°01′ and 92°41′ east longitude. The country is bordered on the east, west and north by India except for a small strip of boundary with Myanmar on the southeast. The Bay of Bengal lies on the south, and the Himalayas are close to its border in the northwest. According to the last Census 2011, the total area is 147,570 km², with a population of 149,772,364 (79% of whom live in the countryside), crisscrossed by numerous rivers including the Padma, Meghna, Jamuna and Brahmaputra, and their innumerable distributaries and tributaries. The country's capital and largest city is Dhaka (population 12,797,394). The commercial capital and largest sea port is Chittagong (3,858,093) and other major cities are Khulna (1,388,425) and Rajshahi (775,495) (GoB 2011). The total divisions are 8, districts 64, upazilas 545, unions 4,543, villages 87,223, households 32,173,630 and city corporations 6.

3.2 Origin of NGOs in Bangladesh: Socio-Economic and Political Realities

The development context of Bangladesh is related to the country's overall socio-economic, cultural and political conditions. Bangladesh got its independence from Pakistan in 1971 with a socialist ideology to free the mass of people from hunger and exploitation, and equal the distribution of power and resources. Bangladesh was ruled by a number of political parties: a parliamentary democracy (1971-1975, 1991-1995), a one-party socialist rule (BAKSAL in 1975), multiparty political party (1975-1981) and military government (1981–1990). The ruling parties were the Bangladesh Awami League (BAL),¹ Bangladesh Nationalist Party (BNP) and Bangladesh Jatio Party (BJP). However, the military governments for a long period ruled Bangladesh. On 11 January 2007, following widespread violence, a caretaker government was appointed to administer the next general election. The country had suffered from extensive corruption, disorder and political violence. The new caretaker government made it a priority to root out corruption from all levels of government. To this end, many notable politicians and officials, along with large numbers of lesser officials and party members, were arrested on corruption charges. The caretaker government held a fair and free election on 29 December 2008. BAL's Sheikh Hasina won the elections with a landslide victory and took the oath of prime minister on 6 January 2009 and is continuing her regime.

The country's development is stated clearly in the state's constitution, and Bangladesh is committed to implement its constitutional procedures. Article 16 states: 'The state shall adopt effective measures to bring about a radical transformation in the rural areas through the promotion of an agricultural revolution, the provision of rural electrification, the development of cottage and other industries, and the improvement of education, communications and public health in those areas so as progressively to remove the disparity in the standards of living between the urban and the rural areas'. By inference, I get a directive to address the disparity in the standards of living. Again, Article 19 (2) states: 'The state shall adopt effective measures to remove social and economic inequality between men and women and to ensure the equitable distribution of wealth among and women and to ensure the equitable distribution of wealth among citizens and of opportunities in order to attain a uniform level of economic development throughout the republic' (GoB 2008, 18). Nevertheless, practically the country failed to implement these constitutional proce-dures. The state policy of socialism was hardly taken seriously and was formally deleted from the country's constitution during the Ershad regime. The operation of the nationalised industries, previously aban-doned by their Pakistani owners, was driven by the objective of enriching a few politically favoured private individuals. The initial process of private context databased of the mationalised the series of private sector development, as pursued during the regime of General Ziaur Rahman, turned out to be an early version of what is now called 'crony capitalism'. Largely, Bangladesh did not see any significant reversal of reforms, even as governments changed, because both the large parties (BAL and BNP) were broadly committed to carrying out the same economic reform agenda (Mahmud et al. 2008).

It is important to emphasise that the economy of Bangladesh is one of the most vulnerable economies in the world, characterised by an extremely high population density, a low resource base, high incidence of natural disasters and persistent sociopolitical instability, especially during the initial years of the state. Moreover, the country inherited a war-ravaged economy after the Independence War. With such extremely adverse initial circumstances, the implications for economic growth were considered extremely unfavourable for Bangladesh. In effect, with extremely bleak development prospects, the predominant theme that persisted during the initial years was one of negative images. Sen et al. (2004, 1) argue that Bangladesh was considered a model of extremities and odds of human existence, an example of a hopeless future, a case of constant fear of some hidden disasters in the making and a permanent cause of liberal conscience and global welfarism. They add that the country was regarded as a 'landscape of disaster' and a 'catalogue of woes'. Such pessimistic appraisals defined a development discourse that conditioned the mind-set of the domestic policymakers as well as the external donors for the subsequent two decades. The 'agrarian pessimists' highlighted the importance of traditional production relations constraining the future developments of productive forces. Similarly, 'persistent authoritarianism' was considered as the ultimate political fate of Bangladesh (Sen et al. 2004, 1).

Bangladesh already had 6 long-term (5 years) plans, 2 medium (2 years), 2 Poverty Reduction Strategy Paper (PRSP) (2005–2008 and 2008–2011) plans and 44 annual (yearly) development plans. The 'poverty' issue has been identified as one of the most important problems in the country. A number of programmes and projects such as agriculture, rural development, primary and mass education, health and family welfare, women and children affairs, and labour and employment are considered for poverty alleviation. The social sectors of the government include the poverty-related projects such as agriculture, education, youth and sports, mass communication, cultural affairs and public administration. It was seen that social development and poverty alleviation projects work as supportive efforts to achieve the planning goals in Bangladesh. The social safety net programme of the present budget is of unique character and very extensive in nature.

The Government of Bangladesh has taken the human factor as the instrument as well as the ultimate objective of development. Since it is being gradually recognised that people must be the focus of development for poverty to be alleviated on a sustainable basis, productive capacity of the poor must be raised. The Fifth Five Year Plan in 1997-2002 stated that the best way to do so is to raise the quality of human capital. Therefore, human resources development has been accepted by the Government of Bangladesh as the most powerful and effective strategy for poverty alleviation. The liability of a large population in this most densely populated country can be transformed into a most valuable asset through human resources development. It is seen that the policies aim at promoting a development process that ensures equitable access to benefits, particularly for the poor and the disadvantaged groups, including women and children and their empowerment. The target of all public policies is to create built-in mechanisms for equitable growth. The efforts are now increasingly directed towards combining resources with social mobilisation, local capacity building, environmental sustainability, gender

equality and with participation in and ownership of development activities. Such actions include both targeted interventions that focus on the needs, interests and rights of the poor and disadvantaged groups as well as structural reforms and administrative reorganisation for good governance to create an enabling environment for their economic, social and political empowerment.

The poverty issue in rural areas is more vulnerable and fragile than in the urban areas. In rural areas, poverty refers to forms of economic, social and psychological deprivation, occurring among people lacking sufficient ownership, control or access to resources for the minimum required level of living. It is widely recognised that a multidimensional, problem-solving mechanism is needed to face this challenge. The Planning Commission of Bangladesh recognises that, despite some progress in poverty alleviation and social sector development in Bangladesh, much remains to be done. The Rural Development and Cooperatives Division of the Government is responsible for planning, implementing, monitoring and evaluating rural development programmes. The Bangladesh Rural Development Board, the Department of Cooperatives and the Academies for Rural Development at Comilla and Bogra play important roles for rural development. The major functions of the rural development agencies include employment and income generation, provision of credit and institutional support to rural target groups, vulnerable group development, formation of rural cooperatives, provision of health care, sanitation, safe drinking water, education, etc.

A number of international organisations, such as the Asian Development Bank (ADB) (2008), Department for International Development (DFID) (2009) and World Bank (2008), together with the Government of Bangladesh (GoB 2007), state that the country is progressing day by day, many businesses are doing well and decades of steady economic growth have helped many people get out of poverty. Many international NGOs, such as DFID, International Labour Organisation (ILO), CARE Bangladesh, ADB and UNESCO, are working with the government sector to stimulate more investment so that it brings economic opportunities for millions of poor people, including the extreme poor. To review the overall development in Bangladesh, the World Bank (2008 in Mahmud et al. 2008) states that Bangladesh has, in recent decades, achieved reasonably rapid economic growth and significant progress in social development indicators such as sustained macroeconomic stability, improved health and education, gender parity in primary and secondary education, and a steady decline in poverty levels. Still the country needs to concentrate to achieve the goals of the MDGs, and Bangladesh is committed to achieving these by 2020. The goals are included in the country's first PRSP, *Unlocking the Potential: National Strategy for Accelerated Poverty Reduction*, and were addressed by the reports of the 12 thematic groups that helped prepare the strategy paper. Both the Medium-term Budgetary Framework and the Annual Development Programme have also been tuned to the MDGs. In this regard, the Government of Bangladesh (Government of Bangladesh (GoB) 2007) states that with 8 years remaining for these goals to be fulfilled, this mid-term review will help guide government policies. Sen et al. (2004, 2) argue that in recent years Bangladesh has graduated into the league of 'medium human development', according to the global ranking of the UNDP.

It is agreed that the overall development in Bangladesh is slow, and a number of explanations are offered for this slow progress. I think the major impediments are the desperate initial conditions after gaining independence, lack of resources, natural disasters, widespread corruption and a record of systemic governance failure. Mahmud (2008) argues that the development strategy and the associated economic environment in Bangladesh since the early 1970s have undergone successive shifts and refinements, often linked with change in the ruling political regime. He adds that it was out of pragmatism that the shift in policy away from socialism to private sector development was initiated within a few years of gaining independence. Now Bangladesh faces the enormous challenge of achieving accelerated economic growth and alleviating the massive poverty that afflicts 40-50%. Ahluwalia and Mahmud (2004) argue that the strategies for meeting this challenge have included a shift away from state bureaucratic controls and industrial autarky towards economic liberalisation and integration with the global economy, on the one hand, and building human capital and empowering the poor, on the other. Policies of substantial budgetary allocations for health and education (admittedly within a small total resource base) were also combined with institutional innovations and public-private partnership to venture to develop a human capital base from the very poor initial conditions. The market-oriented liberalising policy reforms were initiated around the mid-1980s with the support of the IMF and the World Bank, and have been followed through since then in various phases. Although aid conditionality did have an important leveraging role, the sequencing, design and implementation of these reforms had much to do with the political incentives in relation to the economic rationale of such policies. It is generally agreed that economic reforms in Bangladesh have not been matched by progress in building the institutions of political and economic governance (Mahmud et al. 2008).

Despite the history of democratic processes, Bangladesh has its share of the acute syndromes of a fractured polity, bad governance, convulsive society, a dual economy and a political leadership with irreconcilable beliefs, symbols and prejudices. Corruption and poor governance are impeding Bangladesh's efforts to reduce its massive poverty by reducing economic growth and lowering the achievement of social objectives. These can destroy citizens' faith in their government; they also discourage the foreign and domestic investment, which Bangladesh needs so badly, and they undermine the ability of Bangladesh's development partners to sustain their support for the country. The UNDP (2006, 1-2) reported that governance in Bangladesh is on the brink, in large part due to pervasive corruption and élite capture of the state. However, the country is seen by many as mired in a 'crisis of governance' and experiencing 'degraded and deadlocked democracy'. Fifteen years of multiparty rule has not resulted in the consolidation of electoral democracy. The political system has become dysfunctional, institutions have become unresponsive and a systemic lack of respect for and enforcement of the rule of law is generally observed. It is seen that democratically elected governments have fallen far short of delivering on their promises to promote economic prosperity, social justice and attend to the peoples' welfare.

Within these socio-economic and political paradigms, the NGO sector became an inevitable agent for accelerating its socio-economic conditions. The literature shows a number of reasons why the NGO sector flourishes in Bangladesh. The root of the NGOs in Bangladesh is from the colonial period, when private charities and philanthropic groups were formed to establish or maintain schools, hospitals and orphanages. While NGOs existed in the former East Pakistan, the impetus for the birth of the development NGOs in Bangladesh came in the wake of the terrible cyclone of 1970 and the War of Independence in 1971 (ADB 1999, 1). Initially, NGOs were raised at relief and reconstruction. On the other hand, Jackson and Islam (2005, 47–48) claimed that the reason was the shortage of government funds, with Bangladesh increasingly dependent on international donors. The expansion and legitimation of NGOs as influential partners in socio-economic development began largely in the 1980s. According to some sources, the proliferation of NGOs in Bangladesh emerged as one of the mainstream development interventions to address poverty alleviation and female empowerment only since 1990 (Newaz 2003). There are three important factors that encourage NGOs' expansion in development activities: NGOs' success stories, the aid agencies' use of NGOs for development activities and the NGOs' increased funding sources from both the government and foreign donors (Haque 2002). The ADB (1999, 2) reports that the need for NGOs started when inefficiencies in delivery of goods and services led the government, with support from its development partners, to take measures to break monopolies and allow the market mechanism to operate. The vacuum created by the failure of either the government or the market to deliver services was filled by NGOs. Now development (CD) and NGOs' activities are synonymous in Bangladesh.

3.3 NGOs' Activities in Bangladesh: Hopes for Development

In its broadest sense, the term 'NGO' refers to organisations (1) not based in government and (2) not created to earn profit (ADB 1998). On the basis of size, the NGOs can be classified as big, medium and small. From a purposive point of view, NGOs can be single-purpose-based or multiplepurpose-based (Haque 2002, 415). On the other hand, based on source of funding, working coverage and status, NGOs can be classified into three: international, national and local NGOs. There are contradictory opinions regarding how many NGOs are working in Bangladesh. According to the NGO Affairs Bureau (NGOAB) and ADAB, by late 2004 there were 1,882 NGOs registered with the NGOAB, with around 1,100–1,200 receiving foreign funds (World Bank 2006). The NGOAB (2009) lists 2,418 NGOs, but the increase of NGOs in Bangladesh is an issue of concern. The NGO News (2015) reported that now 2,333 NGOs are working in Bangladesh (as on 1 January 2015). The World Bank (2006, 11) states that there are 2,000 NGOs working in Bangladesh, and the bank identifies Bangladesh Rural Advancement Committee (BRAC), Association for Social Advancement (ASA), Grameen Bank and Proshika as big NGOs, which accounted for 85% of all donors' funding to NGOs. They employ 10,000 employees and handle multimillion dollar budgets. These big NGOs have also provided models for the programmes and organisational structures and practices to others.
Fruttero and Gauri (2005) mention that by 2000 more than 90% of rural communities had some NGO presence. Around 13 million, mainly poor women, are now reached through microfinance programmes, some 80% of primary enrolment is provided by NGO schools, and there are nation-wide health and sanitation programmes, where NGOs are involved. The World Bank (2006, 3) shows that one key factor in the rapid growth of the sector has undoubtedly been the availability of foreign aid: grants to NGOs averaged \$160 million annually in the first half of the 1990s, and has grown to about \$238 million annually since then.

Development through NGOs has been a landmark in developing countries such as Bangladesh; their contributions have been defined in many ways, such as 'democratisation of development' (Clark 1991), 'sources of development alternatives', 'vehicle for popular participation' (Farrington et al. 1993), 'advocates for the poor' and 'cost effective and efficient' (Nel et al. 2001; Bebbington 2004). Newaz (2003) describes these innovative approaches as conscientisation, holistic and minimalist. The 'conscientising' approach suggests the poor are systematically disorganised by the economic, social and political relationships. It holds that the only long-term solution to the problems of poverty is to assist the poor in challenging the relationships with increasing awareness and help them become organised so that they can use solidarity as a means of escaping their exploitations-social, economic and political. The 'holistic' approach should be focused on non-formal education, social and political awareness building, and political mobilisation to confront patriarchal power structure, particularly for women, to address the gender subordination. On the other hand, the 'minimalist' approach supports providing credit with minimal training or other supplementary support services to generate income (Ahmad 2006; Newaz 2003).

Most NGOs in Bangladesh provide a strikingly homogenous set of services, with credit dominating. The World Bank (2006) reports a survey of 300 NGO branches in 2003 and shows that, while the total range of NGO interventions is wide, the typical NGO branch provides between three and four services. Around 90% of all NGO branch offices provide credit services, followed by health (56%), sanitation (52%) and education (45%). Islam et al. (2005) conducted a study of NGOs which provide non-formal education. They found that most had included income generation vocational training as a major content area (100%), followed by fisheries, poultry, apiculture, agriculture, animal husbandry, general education and the practice of literacy skills, health and nutrition, social

development (gender equity, cooperative, decision-making and leadership) and cottage industry. Nevertheless, the sector is dominated by some élite NGOs, such as Grameen Bank, BRAC, ASA and Proshika. The impact of microcredit is an increasing income, aiding female empowerment, children's schooling and the health status of the rural poor whilst reducing household vulnerability. These social gains are also associated with the complementary social mobilisation, training and awareness-raising activities that typically go hand in hand.

Several NGOs have established commercial enterprises, whose profit stream is earmarked to support their development programmes. This 'endowment model' is designed to reduce dependence on donors and other outside funding sources, and is an innovative solution also used by foundations and charities elsewhere. A further innovation is the NGOs' role as 'social venture capitalists', developing 'frontier' markets with commercial ventures, such as Grameen Phone and Arong of BRAC. The World Bank (2006) indicates that the profitability of NGO enterprises varies and larger NGOs tend to have the edge. The current market share of most NGO businesses is small, with some notable exceptions. Overall, NGO businesses that are separately incorporated from the parent NGOs' social programmes perform better because separation allows for greater specialisation, more strategic business planning and improved governance.

The NGOs' relationships with the government and donors have been viewed in three ways: legitimacy, financing, and monitoring and supervision. The World Bank (2006) states that the legal framework relating to NGOs is obsolete, with an overabundance of laws and official agencies with limited capacity. The NGOAB is widely recognised as the primary regulator of development NGOs and administering the laws relating to foreign donations in Bangladesh. There are 12 laws for registering and regulating NGOs, but most NGOs register under the Voluntary Social Welfare Agencies (Registration and Control) Ordinance, 1961, administered by the Ministry of Social Welfare (World Bank 2006; Ahmad 2006). Staffing at the NGOAB has barely changed since 1990, even though the number of NGOs the agency is supposed to oversee has tripled. As a result, delays are common, financial reports are not scrutinised adequately and basic data relating to foreign-funded NGOs are not maintained systematically. The microcredit apex body, Palli Karma-Sahayak Foundation, introduced some standardised financial management practices for microfinance programmes, but accounting and auditing practices are

also influenced by donor financial reporting requirements. Such requirements are highly variable and typically focus on the donor-financed project rather than on the whole institution.

Due to its historical dependency on external aid, Bangladesh has one of the strongest local presences of donors, both globally and within South Asia. This large presence has created the need for mechanisms that can help to coordinate the activities of the various donors and the Government of Bangladesh. One such mechanism was the Bangladesh Development Forum (BDF), which functioned as the apex body for dialogue and aided coordination at the international level until 2005, when it was replaced by the first PRSP Implementation Forum. Both the BDF and its successor are comprised of members from the Government of Bangladesh, as well as international and local representatives of donor agencies. The BDF existed under the name of the Bangladesh Aid Group (BAG), also commonly referred to as the Aid Consortium until 1999, when it was renamed the BDF. In the mid-1970s, the World Bank took the initiative to establish the informal Local Consultations Group (LCG) to meet the demand for local aid coordination and donor-Government of Bangladesh dialogue on development issues and priority reforms in Bangladesh. Since its beginnings, the LCG has gone through an evolutionary process, which has considerably enhanced its scope and significance, and it currently functions as the forum for local aid coordination and policy dialogue between the government and donors in Bangladesh. The LCG subgroup on governance functions to promote an active dialogue on good governance in Bangladesh, and to strengthen coordination and cooperation both vis-àvis the government and with the donor.

NGOs and donors increasingly face the twin challenges of demonstrating effectiveness in their work and accountability in their relationships with various stakeholders. Donors, on the one hand, are particularly concerned about accountability of NGOs in the efficient and effective delivery of services. NGOs, on the other hand, are often concerned that accountability to donors can overshadow and overwhelm their accountability to communities and to their own missions (Ebrahim 2004, 4). There are a number of ways that the donors provide direct fund to NGOs in Bangladesh. The most common is funding for specific projects. Where financing needs are large, donor funds may be pooled and a donor– liaison function introduced to coordinate support and reduce transaction costs for the NGO. In some instances, donors have financed the whole range of NGO activities, and in a few of these cases, the institution develops into a different legal entity by the end of the funding period. In recent years, the recognition that partnerships with smaller NGOs carry high transaction costs has increased reliance on wholesale vehicles, in which an agency manages a pool of money for which smaller NGOs compete (World Bank 2006).

It is argued that the main role of government is in modernising the legal and regulatory framework, and ensuring effective oversight of NGO activities. It is important that all parts of government recognise NGOs as allies in development and accept constructive criticism as supportive of a democratic polity. Government should withdraw where NGOs have the clear edge in the delivery of certain services. As the process of decentralisation evolves, local governments can play a more active role in delivering services in partnership with NGOs. Government ought to strengthen monitoring and contracting procedures to improve the effectiveness of aid that is channelled to NGOs. The Government-NGO Coordination Council should be revived to provide a forum for regular government-NGO discussions. Donors should continue to support NGO activities in Bangladesh, both to improve pro-poor service delivery and to promote a broad-based civil society, while ensuring that they do not undermine the accountability of NGOs to Bangladeshi stakeholders.

The NGOs' development interventions became popular and effective for development in Bangladesh for various reasons. Their close links with the community, their quick response to new circumstances, ownership and participation, basic needs, human rights, gender equality, the environment and sustainable development have assigned greater power of NGOs' activities (Haque 2002). The World Bank (2005) states that Bangladesh has nearly achieved the MDGs' goals relating to elimination of gender disparity in schooling opportunities. Dreze (2004) says that Bangladesh is doing better on most social indicators, such as lower infant and maternal mortality rates, higher child immunisation rates, better access to 'improved' water sources and sanitation and higher primary enrolment rates than India. Kahler (2000, 10), through some case studies of NGOs' intervention in developing countries, argues that NGOs' flexibility in programming has allowed for significant innovation. Developing localised, tailor-made responses to technical development problems has forced NGOs to innovate.

Perceptions of NGOs' contributions towards development are mixed. Some believe that NGOs are flexible, innovative and efficient vehicles for the delivery of basic services and for poverty alleviation. The NGOs can reach poor communities and remote areas at lower cost than governments. They can identify genuine local needs, promote participation and transfer appropriate technologies as the 'magic bullets' of development (Gauri and Fruttero 2003). On the other hand, today NGOs are involved in all stages of political processes within the UN, influencing decision-making and policy implementation from the inside. Others have argued that most NGOs are started and controlled by charismatic individuals, who necessarily limit participatory decision-making (Wood 1997), and effectiveness remains weak (Edwards and Hulme 1995). One of the significant problems that NGOs are facing at present is partnership problems with different groups, including GOs versus NGOs and NGOs and funding agencies. For example, regarding GO–NGO partnership, the NGOs' view is that the governments often attempt to maintain control over the devel-opment process initiated by the NGOs and that their operations are under constant threat. The World Bank (2006) commented on the public image of the NGOs' perceived illegitimate shift into for-profit activity, low service quality and limited development impact. But the bank agrees that the Bangladeshi NGOs are building good networks with accountability and trust relationships among service users, politicians or policymakers, and service providers. In South Asia, for instance, Grameen Bank has built a worldwide reputation for its work and now offers development advice and consulting services to a number of other countries (Gauri and Fruttero 2003). It claims that its credit programmes for poor rural women in Bangladesh attain repayment rates consistently over 90% (Khandker et al. 1998) and that they are entirely self-financed.

NGOs' microcredit programmes have been robustly criticised. Haque (2002) argues that large NGOs in Bangladesh are turning into rentseeking institutions because they charge among the highest interest rates (20-30%) on loans given to the poor. These are now aggressively pursuing profit-making business enterprise in major economic sectors. As a result, the original mission of NGOs such as Proshika, Grameen Bank and BRAC to eradicate poverty has become marginalised (Wood 1997). But Newaz (2003) and the Microcredit Summit (1997) state that microcredit to the poor, with a view to achieving equitable and sustainable gains, is the key element for economic and social development in the twenty-first century. Islam and Mia (2007a) find that most of the NGOs have programmes for socio-economic development but a very few of them have innovative elements, for example, in non-formal education linked income generation programme (IGP) for poverty alleviation.

Buckland (1998) argues that a common criticism is that they are smallscale and consequently do not seriously affect widespread poverty. Hedrick-Wong et al. (2001) examine the impact of microcredit on poverty alleviation initiatives of Proshika and found two problems, such as poor knowledge about the use of microcredit and NGOs' imperfect enforcement to utilise this credit. Newaz (2003, 65) argues that NGOs are accountable to the donor, not to the people, and this upward accountability has skewed NGO activities towards a donor-driven agenda for development rather than towards indigenous priorities. As a result, Bangladesh is still a 'test case for development', and its bargaining power runs through development aid. Westergaard and Hossain (2005) claim that misuse of loans and its overlapping are very common in microcredit programmes. Because of their 'donor dependency', their accountability for human services is very loose. They say that recent studies have, however, documented that many women give their loans to their husbands, which causes a big confusion about female empowerment through microcredit programmes.

To consider the aforementioned discussion, the conclusion is threefold: first, the overall development in Bangladesh is managed by the NGOs and their overall development is mixed. But NGOs' development trend is still slow and emphasis is given to the economic issues (e.g. increase of per capita income, employment and microcredit) rather than to sociocultural development (e.g. quality of education, self-reliance, community awareness, peoples' participation in development activities, population control, improvement of nutrition status and human rights). Moreover, NGOs' contributions have been calculated on the basis of quantitative measurement, but their quality of services in terms of social development has been either ignored or taken second place. For example, the rate of child enrolment in school, the number of people under microcredit schemes or the number of people who are taking service from the particular NGO have come to the front of their success. But the quality of education, peoples' sociocultural improvement, self-reliance or sustainability issues have been ignored. Secondly, NGOs' development activities are more donor dependent (e.g. financing, policy issues and development strategies). And thirdly, NGOs' relationships with the government and donors are not reasonably good. Major problems are related with regulation, financing, monitoring and supervision, and NGOs' freedom for working. As a result, NGOs fail to bring satisfactory progress in social development in Bangladesh.

3.4 Why Does Bangladesh Bother NGOs?

As a newly established independent country, Bangladesh is still struggling with a huge population, low per capita income, mass poverty, political unrest, unemployment and underemployment, illiteracy, child labour, malnutrition, corruption and related social problems. Natural disasters occur frequently, with floods massively affecting the poor economy every year, delaying its progress. As development agents, many NGOs (both national, local and international) are working to solve these problems. As stated earlier in this chapter, due to shortages of government funds, Bangladesh is increasingly dependent on international donors. Various studies report that there are disparities regarding the NGOs' contribution for community development. Some of their main weaknesses are low levels of people's participation, less concentration of local resources and lack of resource mobilisation, on the one hand, and donor dependency, on the other. As a result, 'intellectual aid' (a gift of useful knowledge, which is a very different matter from material aid) (Schumacher 1973) is becoming more important for Bangladesh. In this regard, Burns (1996) emphasises the small business and entrepreneurship. Bornstein (2004) argues that social entrepreneurs are demonstrating new approaches to many social ills and new models to create wealth, promote social well-being and restore the environment in the globalised world.

Bangladesh was colonised historically by different nations between 1757 and 1947 by the Portuguese, the Dutch, the French and the British. I argue that the colonialist or early missionary mentality in Bangladesh is still very much alive. There is still a widely held view that anything associated with culture² and hereditary values is pagan, and thus backward, as reflected by the vast number of urban people who feel hesitant, if not somewhat embarrassed, to associate themselves with their own cultural background. It is time to encourage a more informed recognition of this deeply rooted mentality as the product of a particular time and of specific policies in human history, and to acknowledge the limitations it imposes on development as well as its devastating effects on the natural environment. I argue that many socio-economic problems of Bangladesh are related to culture and attitude. The majority of urban youth still pledge to the 'American dream' and, on a minor scale, to the 'urban dream'. The growing trend towards urbanisation is encouraging thousands to discard their indigenous knowledge in the belief that new knowledge and new opportunities are to be found in town. Yet the

realities of mass unemployment, the high costs of urban life and of further education, and the growing pandemic diseases (such as HIV/AIDS) testify that this is not the case. Many end up homeless, jobless and penniless, with neither the traditional skills that sustained their ancestors, nor the specialised and expensive skills required for employment in a modern town. The inevitable result is poverty.

The existing literature shows that there are debates about which kind of knowledge approach favours the progress of developing countries such as Bangladesh. Some turn to indigenous knowledge, some global knowledge, while others believe in the integration of the two. Each has been practised to varying levels and degrees, and Bangladesh already gains from some of these experiences. But the result is not satisfactory (see Barr and Dixon 1998; Howell 2003). In facing up to the challenges of globalisation, there is a great demand for a shift from the traditional paradigm to the new paradigm. In this connection, the development literature in Bangladesh needs to investigate a new knowledge paradigm, which considers the global and local realities. This paradigm would look for maximum benefits, but minimise the disadvantages for community development through NGOs' interventions.

3.5 Conclusions

This chapter has highlighted the main parameters of NGOs. The chapter attempts to integrate various relevant debates of NGOs' activities in the socio-economic, political and cultural development paradigms. It was found that the NGOs have abrasive relationships with state and donors, on the one hand, and global versus local development debates, on the other. The chapter shows how the development challenges might be solved through consulting both knowledge approaches, so that the NGOs can achieve participatory-oriented development interventions according to the choices of the rural mass of people in Bangladesh.

Notes

 BAL, also translated as the Bangladesh Peoples' League, is the mainstream secular political party in Bangladesh and was the political catalyst for Bengali discontent and rebellion in 1971. The League, with co-founder Sheikh Mujibur Rahman as its leader from 1953, was an opposition party in Pakistan and had a moderately socialist ideology. The party's candidates won a majority in the 1970 elections, but the central government in West Pakistan banned the League after war between East and West Pakistan erupted in early 1971. When Bangladesh (formerly East Pakistan) won its independence in late 1971, the party was the nation's dominant political force. The party is now headed by Sheikh Hasina, the daughter of the late Sheikh Mujibur Rahman. In 1981 and again in 1991, it was defeated in a popular election by the BNP. The League, however, went on to win the 1996 parliamentary elections (Encyclopedia 2003).

2. Culture generally refers to patterns of human activity and the symbolic structures that give such activities significance and importance. Cultures can be 'understood as systems of symbols and meanings that even their creators contest, that lack fixed boundaries, that are constantly in flux, and that interact and compete with one another' (Findley and Rothney 2006). Culture can be defined as all the ways of life including arts, beliefs and institutions of a population which are passed down from generation to generation. It has been called 'the way of life' for an entire society. As such, it includes codes of manners, dress, language, religion, rituals, norms of behaviour such as law and morality, and systems of belief as well as the art (Goodall 1986).

Globalisation and Indigenous Knowledge: NGOs' Role for Development

Abstract In recent years, the arena of NGO action has expanded rapidly from local and national settings to the international level due to globalisation. At present, NGOs are facing twin dilemmas in management and operational levels. First, NGO management is a big concern. Most of the NGOs are managed by the external agencies because they mostly depend on foreign funding agencies who decide how the NGO programmes will be managed and organised. Another is NGOs' local pressure because they are working with the local community people who have different demands and needs. But in most cases, they implement the global knowledge-based universal development frameworks. This chapter presents a comparative discussion of indigenous and global knowledge approaches within the globalisation process. The chapter highlights and explains each of these concepts and attempts to establish a significant link with a new role of NGOs for development. This theoretical discussion looks at NGOs' capacities in the paradigms of indigenous and global knowledge approaches. The main purpose is to find out the gaps, for example, what are the general explanations of these concepts, and which are effective for Bangladesh?

Keywords Globalisation · Indigenous knowledge (IK) · Foreign donation · NGO management · Liberalisation · Knowledge debate

4.1 GLOBALISATION: CHALLENGES AND PROSPECTS FOR NGOS IN DEVELOPMENT

Globalisation is not easily understood for its multidimensional nature, asymmetric impact and complexity. It has been characterised by many components and elements which are not easily measured. Ludden (1997) argues that it cannot be understood simply as the result of one dominant force. He adds that the technologies that lower transportation and communications costs do facilitate globalisation, which we can measure as a process by physical movements from place to place, but the globalisation project is neither explained nor mandated by technology. He again says, within moral theories, scientific paradigms, religious passion, cultural mythology and legitimate authorities, people are moving across the old boundaries to the new world. Concretely then, the *globe* in globalisation is a type of human territory, a space that is marked with human aspirations, meanings and power. It is a domain of human control and social order, like any other kind of territory.

The concept 'globalisation' is a dynamic and deliberated word; it means many things to many people. However, globalisation is at the centre of a starkly polarised debate over the major policy issues in the world today. Attempts to define globalisation usually seem clumsy, partial and complex (Sen 2002). Sen argues that it is still not altogether a well-defined concept. A multitude of global interactions are put under the broad heading of globalisation, but it varies from the expansion of intellectual and cultural influences across borders to the enlargement of economic and business relations throughout the world (United Nations 2004b). Different authors often concentrate on very dissimilar aspects of this phenomenon. Globalisation offers many possible meanings, which depend on ideology, context, perspective and location. It is a hotly disputed and contested issue (ILO 2004). Harcourt (n.d.) notes that some global advocates, such as Ohmae (1990, 1995), write enthusiastically about the 'borderless world' and 'the end of the nation state'. Some social scientists prefer a broad, rather unfocused definition, such as the 'movement of people, information, symbols, capital and commodities in global and transactional spaces'.

Most economists understand the concept as free trade and see the modern form of globalisation as part of this process. For instance, Eslake (2000) says, '...globalisation is...simply the logical extension of the tendency towards specialisation and trade that has been going on almost since mankind first walked on the surface of the earth'. Others, like

Friedman (1998), regard globalisation as being not just about trade, but also about the triumph of market forces, technologies and democratic forces throughout the world. Gunter and Hoeven (2004) state that it means gradual integration of economies and societies, driven by new technologies, new economic relationships, and the national and international policies of a wide range of actors, including governments, international organisations, business, labour and civil society.

From the sociological and cultural point of view, Ludden (1997) says the term globalisation refers to human networks of influential interaction which are measured and explained by many factors, including migration, trade, empire, technology and the spread of languages and disparate cultural elements. Mittelman (2000) argues that 'globalisation means an historical transformation in the economy, of livelihoods and modes of existence; in politics, a loss in the degree of control exercised locally ... and in culture, a devaluation of a collectivity's achievements ... Globalisation is emerging as a political response to the expansion of market power ... [It] is a domain of knowledge'. On the other hand, Sen (2002) sees globalisation often as global westernisation. Ottone (1996) says that globalisation is commonly referred to either as 'the knowledge society', 'the information society', 'the communication society', or more generally, 'the post-industrial society'. According to the Free World Academy (2005), globalisation is the integration of democracy, legal reforms, capital, technology and information across national borders, in a way that is creating a single global market and a global village.

Globalisation enhances global markets, technologies, ideas and cohesions. It reduces space and time, and fades borders. It opens the door to opportunities. On the other hand, it increases human insecurity as the spread of global crime, disease and financial volatility outpaces action to tackle them (UNDP 1999). However, it describes not just an increase in the flow of goods, services, images, ideas and people, but a change in the way of production, distribution, consumption, and other defined and undertaken activities.

The literature published by the World Bank and the IMF highlights globalisation as a process of money transformation. For example, Harcourt (n.d.) argues that there are mainly four dimensions in which globalisation appears to human society: trade, investment, organisational change and 'new economy' technology. Dependency and Marxist approaches have emphasised the inequalities as inherent in the global expansion of capitalism. Bello et al. (1998) argue that the capitalism of

Southeast Asia and the nature of globalisation are flawed. They add that increasing liberalisation results is unhappiness for the majority, while enriching a minority and benefiting business in rich countries. Gunter and Hoeven (2004, 8–12) review over 1,200 articles and books on the social impacts of globalisation and find that all factors may have brought some economies and some societies closer together, but they may have also marginalised many countries and individuals. The second aspect of the globalisation process concerns the increasing homogenisation of policies and institutions across the world, for example, trade and capital market liberalisation; the reduction of the welfare state; international agreements on intellectual property rights and the standardisation of policies and behaviours that have promoted globalisation.

Leen (2003) argues that globalisation is neither novel nor new; it is a real danger of cultural homogenisation as the result of contemporary 'Western'-driven models. Current global processes have been shaped by the neo-liberal ideology. It accents the rights and control of the individual by advancing the role of the market place into the economic, political and social sphere (Buckland 2004, 127). Deacon (2000) argues that globalisation is undermining equitable public social provision. On the other hand, many authors argue that globalisation is a new and effective means for development. Sen (2003), for example, argues that globalisation has provided appropriate local steps. In the purest sense, it is not a threat to local cultures—imperialism is. It is seen that developments linked with globalisation have opened up boundless possibilities and new opportunities for human progress and enhanced the quality of life for many people in the developing countries. Local knowledge (LK) is not useful to solve many problems, such as the flood problem in Bangladesh. Some authors have mixed opinions about globalisation. For example, Morgan (2005) argues that the rapid process of globalisation has clearly destroyed and damaged many indigenous and belief systems in recent centuries, though he argued that this is simply an inevitable process of historical change. As Karl Marx says, human beings make their own history, but not always, as they intend and not in circumstances of their own choosing (Layder 1987).

Globalisation provides access to resources to use the political strategies of cooperation, negotiation and contestation, through which rural groups challenge their poverty. Recently, the argument has been raised that due to globalisation the development organisations (like NGOs) are facing varieties of challenges in this new millennium. For example, Buckland (2004, 128, 139) argues that the NGOs (mainly big NGOs) have been encouraged to directly confront policy-level constraints to development and move out of programme implementation. These changes have been significantly influenced by official aid agencies and NNGOs, and complex implications for indigenous NGOs and local communities. He adds that many people, including local poor people in Bangladesh, have benefited by the scaling up of reform-oriented development organisations. But this strategy has not overcome other powerful local, national and global constraints to development that have led to persistent poverty, gender bias and economic change that has not benefited the poor.

However, we understand that globalisation is a dynamic as well as a complex concept. We see that there is much academic discussion about whether globalisation is a real phenomenon or only a myth. The term is widespread; it is seen that the characteristics of globalisation have already been seen at other moments in history. Also, many note that those features include international trade and multinational corporations, which are not as deeply established as they may appear. Globalisation evokes different feelings to different people. Some look at globalisation as 'the panacea'—the elixir that eradicates all poverty, removes disparity and enables the global citizens to enjoy a decent lifestyle—the lure of globalisation. Others view globalisation as 'the evil' that further enslaves them to the rich.

4.2 INDIGENOUS KNOWLEDGE: A LOCAL DEVELOPMENT INTERVENTION FOR NGOS

'Indigenous knowledge' (IK) is an important term in contemporary development studies. Since there are increasing international concerns about the negative impacts of globalisation, IK has been considered a key aspect in sustainable development. It calls for the insertion of local voices and priorities, and promises empowerment through ownership of the process. However, there has been little critical examination of the ways in which IK has been included in the development process (Briggs and Sharp 2004, 661). IK has been used since the dawn of civilisation, but it was first formally recognised as invaluable to sustainable development at the Rio Conference¹: 'Our Common Future' in 1987. Since then, many institutions have shown increasing interest in it and numerous international conferences have taken place concerned exclusively with IK. IK is also known as 'local' or 'traditional' or 'rural peoples' knowledge' and is not easily or simply defined. It is a diverse concept. In anthropology, it refers to the original inhabitants of a specific geographical area, a land which may have been occupied subsequently by migrants or colonists (Morgan 2005). In the 1960s to the late 1970s, the word then took on a more explicitly populist flavour of 'grass-roots politics' and 'folk cultures' in the sense of 'indigenous' as opposed to state of *high culture*² (Scoones and Thompson 1994). In view of the marginalisation process, IK today means 'non-Western' or 'anti-Western' knowledge'. Knowledge is much more than a collection of facts; it relates to the whole system of concepts, beliefs and perceptions that people hold about the world around them. IK includes the way people observe and measure what is around them, how they set about solving problems and how they validate new information.

Many authors and institutions describe IK as a system or body of knowledge, which evolves from a variety of sources within a social process. For example, Johnson (1992) states that IK can be defined as 'a body of knowledge' built up by a group of people through generations of living in close contact with nature. Such knowledge evolves in the local environment, so that it is specifically adapted to the requirements of local people and conditions. It is also creative and experimental, constantly incorporating outside influences and inside innovations to meet new conditions. It is usually a mistake to think of IK as 'old-fashioned', 'backwards', 'static' or 'unchanging'. On the other hand, some authors describe the concept with some components in a specific framework. Others state that IK is based on some specific group of people (called traditional/indigenous or aboriginal people). I think that the first two views are more accepted than the last one. I believe that IK has already been proved as a body of knowledge given its enormous literature and widespread application. The third view explains the concept evolves. My argument is that no definition is complete, but taken together they explain what IK means. I agree with Cheng (2003, 6), who argues that IK is the knowledge that has been tested as valid in a local context, and accumulated by the local community or people. The usefulness and validity of IK varies according to time and social context.

IK covers numerous components and aspects, which make it wide and complex. It includes agriculture and horticulture, astronomy, forestry,

human health, traditional medicines and healing, knowledge of animals, fish and ecological systems, sustainable use of natural resources and the environment, traditional classification systems for living, learning systems and oral traditions, spirituality, symbols, traditional arts and culture, designs, symbols, scientific and ecological methods, crafts, music, dance, songs, stories, foods, medicines and wellness (or disease prevention), and products (Hoppers 2004). Such knowledge systems are cumulative, representing generations of experiences, careful observations, and trialand-error experiments. IK does not exist in a vacuum; it belongs to a community, and access to this knowledge is gained through contact with that community. No one person, authority or social group is the single repository for this knowledge or can claim to know the entire body of knowledge. It is more widely shared locally than specialised scientific knowledge (Sillitoe 2002). With this view, Dr Erica-Irene Daes, Chair-Rapporteur of the UN Working Group on Indigenous Populations, explained in a 1994 report that IK is a complete knowledge system with its own concepts of epistemology, philosophy, and scientific and logical validity (Kooy and Vizina 2006, 6).

IK is locally based and recorded, for example, in the memories of the living and transmitted orally. But it is dynamic and is 'developed to very specific biological, ecological, climatic and socio-economic conditions'. It is a mixture of knowledge created endogenously and acquired from outside, but then absorbed and integrated within the society by 'trial and error'. The broad set of IK includes information of a functional and of an aesthetic character. The processes and products can be used in agriculture or industry, as well as intangibles of cultural value. IK is not static; it evolves and generates new information as a result of improvements or adaptation to changing circumstances. The context and forms of expression of IK vary significantly (Correa 2001). Correa adds that some IK is codified, that is, formalised in some way (e.g. textile designs, ayurveda a traditional medicine). A great part of IK, however, is non-codified or tacit, such as 'folk', 'tribal' or 'indigenous' medicine, which is based on traditional beliefs, norms and practices accumulated through centuries of experiences of trial and error. It is holistic, qualitative and is a cultural property rooted in a social context (Johnson 1992). It is non-formal, dynamic and adaptive (Boven and Morohashi 2002, 12). It is traditional, which is 'a fluid and transferring agent with no real end' when applied to knowledge. Negotiation is a central concept.

4.3 INDIGENOUS KNOWLEDGE VERSUS GLOBAL KNOWLEDGE: NGOS' NEXUS ROLE

IK for development is an emerging issue in social sciences, including development studies. In recent years, this has been considered a significant research topic in higher education. Thaman (2003) argues that the incorporation of the indigenous perspectives in higher education is an imperative because (1) various vibrant indigenous cultures have their own views of the world which deserve recognition and acknowledgement, (2) the academic institutions should recognise the ownership and control of IK by indigenous people rather than by the academy and (3) the academic centres (in the context of specific studies) needed indigenous cultural knowledge to validate and legitimise their work.

A vast amount of literature has been published on the context of knowledge and development. As a result, the knowledge-based development paradigms have divided into a number of groups. This process causes tension among development actors such as NGOs. Despite some significant advantages of global knowledge (GK), IK is still considered as an important knowledge approach for development in countries such as Bangladesh, where a significant portion of the population and their occupations are based on IK. Guhathakurta (2004, 115-122) discusses past and future options of the development paradigms in Bangladesh and argues that the development options have shifted through a number of paradigms. These are from self-reliant autarchic goal to integration with the world economy; from public sector to privatisation; from poverty reduction to efficiency; governance, gender and human rights; and modernisation and growth versus indigenous perspective. The overall socio-economic texture of the society in Bangladesh remains traditional. According to his view, Bangladesh needs to emphasise an indigenous approach for development. This approach has become an important development topic given the failure of top-down, externally driven development interventions. Seeland (2000) calls it an invention of a postmodern development approach. The great value is that it can be a key element of the social capital of the poor and constitutes their main asset in their efforts to gain control of their own lives (Gorjestani 2000). However, this issue becomes more complex in some developing countries, where the socio-economic and cultural situations are comparatively poor, diverse and heterogeneous.

IK in Bangladesh is an integral, valuable resource based on deeprooted traditions. Craft persons have acquired various skills, developed through the wisdom and experience of successive generations. The knowledge, culture, skills and practices based on IK have been woven into community development, most of them being established through a process of trial and error, which underpin their ontological and epistemological basis rather than through scientific research. However, they do have some scientific principles behind them, though such principles may appear somewhat obscure or even immersed in knowledge bases, which are not at present fully recognised as scientific approaches. However, the practitioners and other beneficiaries of this indigenous and traditional knowledge and these technologies are not fully aware of the scientific principles behind them due to illiteracy, ignorance and lack of social education.

Currently, there is a big debate about how development organisations such as NGOs will introduce development interventions, which consider a meeting between IK and GK. These reflect tensions that occur between the global and the local, between the universal and the individual, between tradition and modernity (and now postmodernity), between long-term and short-term considerations, and between the need for competition and concern for genuine quality of opportunity (Delors 1996, 15 in Morgan and Tujman 2000, 56). Delors states:

Culture is steadily being globalised, but as yet only partially. We cannot ignore the promises of globalisation nor its risks, not the least of which is the risk of forgetting the unique character of individual human beings; it is for them to choose their own future and achieve their full potential within the carefully tended wealth of their traditions and their own cultures which, unless we are careful, can be endangered by contemporary developments. (Delors 1996, 15, in Morgan and Tujman 2000, 56)

The most urgent needs in Bangladesh are about reshaping educational systems to foster knowledge and love of local ecological and cultural diversity. The integration and development of the local language is essential for more equitable and effective means of communication and cooperation across the differences (Rhea 2000). The key challenge here lies in the creation of 'a concept of knowledge to which a concept of experience corresponds'. Lewis et al. (2008) suggest a form of representation, which would capture the subjective dimension (universal dimension) of social reality, whilst simultaneously allowing an objective knowledge (true in certain times, places or people) of the world.

The important thing is how to save and hold up the worth of human diversity in the face of the powerful negative potential that globalisation exerts (Rhea 2000). Cheng (2003) argues that the relationship between LK and globalisation in education is dynamic and interactive. Localised globalisation (globalisation based on the local sociocultural, economic and political situation) in education can allow for better utilisation of values for local developments if local creativity and adaptation can be induced in the process of operational change and cultural change. Cheng agrees that this integration is a big challenge for national and international NGOs for community development. Cheng further adds, from a perspective of long-term local and global developments, the integration of both localisation and globalisation would be a preferable choice. So, it is a contemporary question to the NGO providers, policymakers and development workers whether to investigate a new policy framework (which might be called a 'new knowledge paradigm'), where NGOs' capacity for social capital and community empowerment can function to achieve sustainable community development and self-reliance. This new paradigm would focus on 'people' rather than 'interventions' and identifies the 'client', not the 'policy', as an important step in addressing the social, economic and political needs of the people.

4.4 CONCLUSIONS

This chapter highlights different aspects of GK and IK, and looks at the space for NGOs in development activities. The chapter proves that both knowledge interventions are important for NGOs' development activities. In this case, NGOs are facing twin pressures, such as the local pressure from the local people and global pressure from the donor agencies. Globalisation brings a lot of opportunities and development initiatives for the NGOs, which are proved as productive and effective for rural development. On the other hand, NGOs are taking place local peoples' opinions in the decision-making process through local-level planning and local-level participation, which is based on LK. However, it clearly proves that an initiative is emerging, which might resolve this twin pressure for a greater success of the NGOs' activities.

Notes

- 1. Of equal concern to many world citizens is the uncertain status of the IK that reflects many generations of experience and problem solving by thousands of ethnic groups across the globe. Very little of this knowledge has been recorded, yet it represents an immensely valuable database that provides humankind with insights on how numerous communities have interacted with their changing environment including its floral and faunal resources (Warren 1996, 81).
- 2. 'High' refers to those aspects of culture which are most highly valued and esteemed by a given society's political, social, economic and intellectual elite. Opera, yachting and Tom Stoppard are associated with high culture in the USA. Generally, the most powerful members of a society are the ones who have the most influence over cultural meaning systems, and therefore the more powerful classes tend to enjoy the privilege of defining 'high culture' (Carlisle 2006).

NGOs' Facet in Social Capital and Community Empowerment

Abstract Social capital and community empowerment are two important domains for NGOs' development activities. Social capital is the quality and quantity of relationships, networks and norms among people and organisations that facilitate collective action. The objectives are bonding strategies that build trust and cooperation among individuals and within communities; bridging strategies that break down barriers across groups and communities and enable collaborative action on shared objectives; and scaling up strategies that connect communities in collective action for social change and development at the policy and/or systems levels. On the other hand, community empowerment is a broad term that involves social action processes, and individual and collective efforts and outcomes. It refers to individuals, families, organisations and communities gaining control within the social, economic and political contexts of their lives in order to improve equity and their quality of life. The objective of this domain is to promote development by providing motivation, education and proper cognitive, democratic and social skills; increase communities' capacities and resources to bring people together around common goals and interests; increase participation of the communities in decision-making and problem-solving processes; enhance exchange and partnership with local, regional and international communities and groups; and build-up social capital. This chapter elucidates the complex intersection between social capital and community empowerment. The chapter provides a solid theoretical

discussion about different aspects of NGOs' development and then examines how social capital and community empowerment are widening where NGOs' role is significant.

Keywords Social network · Human relation · Social relation · Social capital · Community empowerment · Social development

5.1 Social Capital: A Widening Field for NGOs' Activities

Social capital has become one of the most striking concepts in social sciences over the last decade. This is particularly true of work in development activities. It has been treated as a key feature of phenomena, which includes so many aspects that cannot be described within limited space and time (DeFilippis 2001; Portes 1998; Islam 2015b). Many definitions are now attached to the concept, which lead to justifiable confusion about what constitutes 'social capital' (National Statistics 2005). Morgan (2008, 35) argues that the concept social capital is still relatively vague and underdeveloped as a concept of policy value. It is described as a muddy (Evans 1996), elusive (Kilpatrick et al. 2001), robust (Paldam 2000) and divergent (Fedderke et al. 1999) concept. It has been called social energy, community spirit, social bonds, civic virtue, community networks, social ozone, extended friendships, community life, social resources, informal and formal networks, good neighbourliness and social glue (National Statistics 2005). I would say that it is hard to encapsulate in a single sentence because its elements, nature, theoretical development and measurement are continuously widening over time.

Social capital is a relatively new concept (Bayat 2005, 2), though the term was first used by Lyda J. Hanifan, a state supervisor of a rural school in West Virginia, in 1916, to explain the importance of community development for successful schools (Olate 2003, 9). It was then used in 1950 by a team of Canadian urban sociologists in 1960 by an urban scholar Jane Jacobs, and in 1970 by an economist Glenn Loury (Olate 2003, 9). The concept was developed in the 1980s and 1990s by Pierre Bourdieu (Bourdieu 1986), James Coleman (Coleman 1988, 1990) and Robert Putnam (Putnam 1993, 1995, 2000a). Portes (1998) argues that this concept dates back to Durkheim's work, which emphasise Marx's distinction between an atomised class-in-itself and a mobilised and effective class-for-itself. After the 1990s,

this concept became one of the principal topics to social researchers. Baum (2000) states, before 1981, that the number of journal articles listing social capital as a key word totalled 20, between 1991 and 1995 this has risen to 109, and between 1996 and March 1999 the total was 1,003. I guess the number would now be several thousands.

5.2 Meanings and Aspects of Social Capital

In 1985, Pierre Bourdieu was the first sociologist to use this term in community development as a neo-Marxist model (DeFilippis 2001). Bourdieu (1986) says that the sources of social capital combined the actual and potential resources, concurrent to tenure of a strong network of more or less institutionalised relationships or reciprocal relationships (e.g. membership). His explanation was concurrently both economic and a set of power relations that comprised a variety of dominions and social interactions generally thought of as non-economic (DeFilippis 2001).

The American social scientist James Coleman (1988) brought social capital into the mainstream. He argues:

Social capital is defined by its function. It is not a single entity but a variety of different entities, with two elements in common: they all consist of some aspect of social structures, and they facilitate certain actions of actors...within the structure. (Coleman 1988)

Robert Putnam's redefinition of social capital adds new dimensions. Putnam, who may be the most important author on the subject, defines

'Social capital is features of social life-networks, norms, and trust-that enable participants to act together more effectively to pursue shared objectives'. The norms and reciprocal networks that make collective action possible are themselves contingent on an existing foundation of social trust and solidarity. (Putnam 1993, 1995, 2000)

The most important characteristics of Putnam's definition are (1) social capital is twisted by individuals or groups of people in regions, communities, cities, countries or continents; (2) it becomes mainly a normatively good thing and is given credit for (a) promoting good, democratic government and (b) generating and sustaining economic growth and development; (3) it can be simultaneously a 'private good' and a 'public

good'; (4) it refers to the norms and networks of civil society that lubricate cooperative action among both citizens and their institutions; (5) network of trust and voluntary associations are 'win-win' sets of relationships in which everyone involved benefits; and (6) it was based on scientific investigation (DeFilippis 2001).

Kilpatrick et al. (2001) intend to categorise all of the major definitions of social capital into two main groups that can be called 'collective benefit' and 'individual benefit'. They say that the definitions by Putnam and Narayan are collective benefit groups and Coleman and Bourdieu's are individual benefit groups. Like DeFilippis (2001), I think that there are two key components of Bourdieu's work, which have been lost in current discussions of social capital. First, he almost envisages capital and power as synonymous, and he differentiates between the social networks that an individual is embedded in, and out of which social capital precipitates (or emerges), and the outcomes of those social relationships. Secondly, social networks should not simply be associated with the products of those social relationships that might be intense, but are unable to generate resources because of lack of access.

Coleman stated different sets of actions, outcomes and relationships as social capital. Social capital for him is inherently functional, and it is whatever allows people or institutions to act. DeFilippis (2001) argues that this definition is normatively and morally neutral. That is, neither desirable nor undesirable; it simply allows actions to take place by providing the needed resources. Putnam's definitions are also criticised for their lack of potential utility for community development as limited support comes from either theories of political economy or the empirical realities; it is quantitative (e.g. more or less) not qualitative (better or worse) change. But I argue Putnam's explanation is wide and based on empirical evidence, where he explains how and where this concept develops at individual and group level in the community.

Based on existing literature, Vitásková and Mateju (2006) sum up two significantly different approaches to the conceptualisation of social capital. The most dominant stream defines social capital primarily as an attribute of societies, as an inherent characteristic of social environment based on the high degree of interpersonal and institutional trust facilitating people's cooperation. The other stream defines social capital in terms of mutually beneficial exchanges based on social connections and informal networks allowing individuals to achieve their own particular goals. The former approach prevails in 'Western' countries, while the latter prevails in the study of social change in post-communist societies, where social capital drawing from interpersonal trust seems to be rather low.

The use of the concept 'social capital' is very wide, and it is widening across the disciplines over time; it refers to norms, 'neighbourliness' and networks that enable people to act collectively (Narayan and Woolcock 2000). It includes institutions, relationships, attitudes and values that govern interactions among people and contribute to economic and social development (Grootaert and Bastelaer 2001). It is called the relationships among actors (individuals, groups and/or organisations), which create a capacity to act for mutual benefits or a common purpose. However, the core elements of social capital are interaction, trust, cooperative ventures, agreements, norms and reputations, emotive behaviour, networks, sanctions and long-term relationships, the local commons, dispositions, belief, economic performance, conformism and contagion, and culture.

Social capital is now considered as one of the most important assets of society. It is an accumulation of social, psychological, cultural, cognitive, institutional and related assets that increase the amount or probability of mutually beneficial cooperative behaviour. It can be formed as structural (roles, rules, precedents, procedures and networks) and cognitive/normative (arising from mental process and resulting ideas, culture and ideology, norms, values, attitudes and beliefs that contribute to cooperative behaviour and mutually beneficial collective action). Putnam (1998) makes a further distinction amongst these dimensions, highlighting a difference between informal and formal networks. Informal networks include the relationships that people have with their families, partners, friends and neighbours; whereas formal networks include relationships at work, within community groups and churches, and with formal bodies, such as businesses and governments. Woolcock (2001) specifically classifies three types of social capital: bonds, bridges and linkage. Bonding social capital refers to relations among family members, close friends and neighbours. Bridging social capital refers to more distant friends, associates and colleagues, in general people who share similar demographic characteristics. If bridging social capital refers to the horizontal dimension, linking social capital refers to the vertical dimension. Linking social capital is defined as the capacity to leverage resources, ideas and information from formal institutions beyond the community. Carroll (2001) classifies social capital into six domains: family and kinship connections, community networks, cross-sectoral linkages, institutional policy framework, sociopolitical (state-civil society) relations, and social norms and values.

In general, capital can be categorised into six headings, such as human, social, financial, cultural, physical and natural/environmental. The function of each is to create wealth and human well-being. Human capital includes knowledge, skills, competencies and attributes embodied in individuals, which facilitate the creation of personal, social and economic wellbeing. It also includes motivation, moral behaviour and attitudes. In this regard, Putnam (2000b) notes that human capital refers to individuals, whereas social capital refers to connections among individuals and the social networks, and the norms of reciprocity that arise from them. It arises from family, schools, local communities, firms and national or subnational administrative units and other institutions. On the other hand, Bourdieu (1986) argues that social capital is never completely independent, like physical and human capital, because of its homogeneity and multiple effects. Coleman (1988) suggests that, like other forms of capital, social capital is productive, but it can be depleted if it is not renewed. In addition, Putnam (2000b) also suggests that the more people work together the more social capital is produced, and the less people work together the more community stocks of social capital are depleted. Schuller (2000) argues that cultural capital is more academic. It refers to the credentials and cultural assets embodied in individuals and their families. It has been used in two constructing directions: the reproduction of social hierarchy, as elite families endow their children with the cultural capital, which enables them to succeed in maintaining their élite position. It is also used to explain how to manage and use education to move from non-élite positions into élite positions.

There is a big debate whether social capital can produce benefit and productivity like physical and human capital. In this perspective, Grootaert and Bastelaer (2001) argue that the key attribute of capital is that it is an accumulated stock from which a stream of benefits flows. Here social capital is an asset, which characterises actual capital, and it is more than just a set of social organisations or social values. They again state that the social capital studies and the empirical literature elsewhere document that social capital can directly enhance output and lead to higher productivity of other resources, such as human and physical capital. I believe that the scope of social capital and cultural capital is broad and covers a lot of components that do not fall under the conventional categories of physical and human capital. There are a large number of opinions whether social capital is a measurable concept. For example, Grootaert and Bastelaer (2001) argue that social capital is difficult, if not impossible, to measure directly like human capital. For empirical purposes, the use of proxy indicators is necessary. They show an example that years of education and years of work experience have a long tradition as proxies for human capital and have proven their value in numerous empirical studies. They also argue that social capital indicators differ both geographically and sectorally. For example, measures of membership in associations were found to be a relevant indicator in Indonesia, Kenya and countries of the Andean region, but not in India and Russia, where informal networks are more important (Grootaert and Bastelaer 2001).

At present, a wide range of literature has been published on measurement of social capital where different authors use different indicators. A brief description of international attempts is shown in Table 5.1. However, it is seen that most of the authors use a number of aspects for this measurement. But a number of elements, such as network, trust,

Authors and year	Indicators of measurement
Coleman 1988	Children's educational attainment included personal, family and community dimensions
Robinson 1997	Family relationships, as these relations are the basis of all other relationships
Onyx and Bullen 2000	Attitude, trust, participation in the local community, reciprocity, personal empowerment, diversity/openness, relation within the workplace, attitude to government and demographic information
Grootaert 1998	Horizontal associations (e.g. local institutions, kin homogeneity, trust, community organisations, etc.) Civil and political society (e.g. civil liberty, political rights, democracy, corruption, etc.) Social integration (e.g. social mobility, divorce rate, unemployment, etc.)
	Legal and governance aspects (e.g. bureaucracy, court system, etc.)
Cox 1998	Unobtrusive measures, measures of attention and wider interest in the society, sociability and engagement, attitude measures—how people express their feelings, expectations and the multidimensional picture

 Table 5.1
 Measurement of social capital

(continued)

Authors and year	Indicators of measurement
Narayan and Cassidy 2001	Structure Vs norms, sources Vs outcomes, form Vs function (formal and informal groups, networks, etc.) and narrow Vs broad (social institutions)
Krishna and Shrader 1999	Structural (membership and involvement of organisations, network mutual support, collective actions and conflict resolution)
Putnam 2000	Cognitive (solidarity, trust, reciprocity and cooperation) A wide range of services (housing, day care, family health clinics), their responsiveness and involvement to public, private, voluntary, civic and cultural services (e.g. postal, telephone, club, sports, etc.) and social trust
Green et al. 2000	Neighbourhood, network geography, reciprocal help and trust, trust (system and personal), civic engagement and efficiency, health, lifestyle, economy
Cote and Healy 2001	Coverage of key dimensions (networks, values and norms) and should be balanced between the attitudinal/subjective and the behavioural (such as cultural context for measure behaviour and attitude)
National Statistics 2005	Participation, social engagement, commitment Control, self-efficiency Perception of community-level structures or characteristics Social interaction, social networks, social support Trust, reciprocity, social cohesion
Spellerberg 2001	 Behaviour (giving to strangers, participation in voluntary organisations, participation in voluntary organisations, participation in informal networks, wider interest in the society, community participation and compliance with rules and norms) Attitude and belief (beliefs about self, attitude towards others, trust and reciprocity, attitude to government and others, social institutions, values and norms, and outlook for and confidence in the future) Population groups (demographic, family, cultural, employment and communication) Organisations (number, type, size of 'membership', structure,
Krishna 2003	capital and networks, and mode of operation) Membership in labour-sharing groups, dealing with crop disease, dealing with natural disaster, trust, solidarity and reciprocity
Stimson et al. 2003	Qualitative assessment (awareness of community events, perceptions of the community, community activities and perception of what residents want/do not want in their community) A structural survey approach (informal structures, formal structures, informal norms, formal norms)

Table 5.1 (continued)

Source: Adapted by author from different sources.

membership, relationships/association, social integration, attitude, social support and social norms/values, are very common. I argue that the authors use different scientific methods and statistical tools (e.g. correlation, regression), but most did not state the limitations of their studies, as I believe that there were some limitations. One significant limitation is the question of the reliability and validity of the indicators, such as the data collection instruments and methods (Paldam 2000; Ostrom and Ahn 2003), and generalisation of this measurement (Portes and Landolt 1996). Each of those measures has merits in the specific context in which it was used. Due to the strong *contextual* nature of social capital, it is unlikely that it will ever be possible to identify a few 'good' indicators that can be used everywhere as universal indicators for such measurement.

The aforementioned discussion shows that social capital has received increased attention in the literature and has been studied at multiple levels, including the individual (Burt 1992), organisational (Nahapiet and Ghoshal 1998) and societal (Putnam 1993). At the individual level, social capital has been defined as the resources embedded in one's relationships with others. At the organisational level, it has been defined as the value to an organisation in terms of the relationships formed by its members for the purpose of engaging in collective action (Freel 2000). Social capital is an attribute of individuals and their relationships that enhances their ability to solve collective action problems (Ostrom and Ahn 2003). The role of social capital has also been examined on a more macro-level in terms of its impact on the well-being of regions or societies (Bourdieu 1986; Coleman 1990; Putnam 1993). We see that the concept has been robustly criticised both for its original assumptions and theoretical foundation, as well as for the acceleration of its uptake and practice. I argue that much of this criticism is made based on ideological position, limited theoretical review and restricted empirical examination.

5.3 Social Capital, NGOs and the Development Debate

I stated earlier in this chapter that the term 'social capital' has been the subject of great interest and debate in the development literature. As a result, it has been elaborated so widely that it is very difficult to explain its theoretical context briefly. Evans (1996) suggests the importance of a more regionalised social capital to overcome locally confined solidarities in order to increase representative bargaining power and access to

information. There is a general agreement that it is something that exists between individuals and organisations. This 'something' emerges from connections between entities and is further developed through trust, mutual understanding and mutual actions, based on shared norms and values (Kay 2005). In a study of 13 development programmes in Asia and Africa, Brown and Ashman (1996) found that higher levels of social capital in terms of the number and quality of grass-roots organisations, and the degree of intersectoral links between public and non-governmental actors, resulted in more effective programme design, implementation and significant long-term change. They concluded that the creation and strengthening of social capital is essential to building intersectoral collaboration in order to mobilise local resources and direct energies towards future problem solving.

Newton (2001, 206) states that voluntary associations are alleged to create the cross-cutting ties and social networks that bind society together. They create social bonds between like-minded people and bridges between different social groups, wherever there is an overlap. Abom (2004) argues that NGOs can play a role in encouraging social capital between the community and the state by standing with grass-roots community groups by helping to broker a 'place at the table'. In the process, opportunities are provided for collective action that may allow people to exercise their own power.

Abom (2004) states that some scholars argue that social capital is not inherently linked to noble ends, such as democracy, equitable economic development and inclusion that Putnam and others have attributed to it. Rather it has been seen that nepotism, exclusivity and divisions among families and communities may result from strong forms of social capital. NGOs seem to impinge on social capital both positively and negatively, depending on their intervention approaches. For example, as observed, some NGOs limit or obstruct social capital formation by providing direct services. Abom (2004) argues that this strategy, implemented mainly by Northern-led and Northern-funded NGOs, aims to reduce the immediate suffering of the 'poor' and meet their short-term visible needs in the hope that the 'poor' may get themselves back on their feet to 'escape poverty'. Such NGOs glimpse their role as the 'doer', placing low expectations of participation and commitment on beneficiaries.

It is agreed that this approach does provide for the human and physical relief and development of many 'poor' people. But many argue that this non-participatory approach minimises the wider social capitalbuilding processes, primarily by contributing to people's mentality of dependence on others, perpetuating top-down values and acting as a barrier to more fundamental structural change. Moser's (1996) investigation of urban squatter communities finds that communities served by top-down delivery systems, including NGO welfare programmes, had less developed community-based organisations. Buckland's (1998, 239) case study of some prominent NGOs in Bangladesh highlights how advances made by highly technical income-generation projects came at the expense of community self-reliance and the 'ability to communicate and promote organisation among the rural poor'. On the other hand, Dowla (2006) finds that the microcredit of Grameen Bank in Bangladesh had created and cultivated a substantial level of social capital by creating trust, norms and networks.

5.4 Community Empowerment

The concept of 'empowerment' is a buzzword. Empowerment is a process whereby individuals struggle to reduce personal powerlessness and dependency by having increased control over their lives (Lord and Hutchison 1997). It is highly advocated by prominent development actors and mainly NGOs (Bodja 2006). As a broad concept, it represents more than a simple increase in income level or access to material resources (Islam and Mia 2007, 15). Empowerment enables people to organise and influence change on the basis of their access to knowledge, to political processes and to financial, social and natural resources (Bodja 2006). The concept has been defined in the literature in two ways: individual empowerment and community empowerment, but there is no universally accepted definition. It means different things to different people, and it varies across countries and cultures. It provides rebalancing of the structure of power within a community by addressing control of resources, control of decisions, control of information and physical domination (INTRAC 2005). It helps to expand the assets and capabilities of the poor people to participate in, negotiate with, influence, control and hold accountable the institutions that affect their lives. The basic elements of empowerment are rights, resources and support to information, inclusion/participation, accountability and local organisational capacity (Narayan 2002; Islam and Morgan 2012a).

Now the discourse of 'community empowerment' has become the common denominator of action for development agencies in a

global world (Tembo 2004; Islam and Morgan 2012a). The concept was used in development in the 1990s to replace community participa-tion (Rifkin 2003). Rifkin uses the CHOICE (C=capacity building, H=human rights, O=organisational sustainability, I=institutional accountability, C=contribution and E=enabling environment) theory provided by the Nobel laureate Amartya Sen (Sen 1999). He shows how health equity can be achieved through community empowerment. The emergence of empowerment as a development topic is associated with the overall shift to alternative development as policies that are derived from mainstream development. Empowerment as an 'alternative development approach' is now approaching, which allows social aspects of development and local communities to enhance participation through using local resources. Laverack (2006) reviews a number of authors' 'domains' approaches of community empowerment, and then he develops nine components of these domains. This subchapter considers these nine domains, such as improving participation; developing local leadership; increasing problem assessment capacities; critical awareness of people's needs; building of organisational structures; improving resource mobilisation; strengthening links to other organi-sations and people; creating an equitable relationship with outside agents; and increasing control over programme management. These domains influence and allow individuals and groups to become better organised and mobilise themselves towards community change. Laverack states that there is considerable similarity between the domains. He adds, however, that there is no consensus on which particular combination is the most effective for community empowerment. Instead, the domains provide a guide to community development practitioners in their planning, application and evaluation of community empowerment in a programme context.

Laverack says, although these domains have been used by practitioners for many years, their purposeful integration in a programme context is a more recent idea. The role of the practitioner in using an empowerment approach has been conventionally concerned with facilitating the movement of people along a continuum. These domains give a more precise way of developing strategies for progression along the empowerment continuum. Laverack argues that this domain approach is very flexible and can lead to individuals and groups developing empowerment strategies to better identify their problems and solution through collective action.

NGOs have certain characteristics and advantages, which allow using those empowerment elements successfully compared to state implementing agencies. Wuyts et al. (1992, 134) argue that the empowerment approaches can work successfully with the NGOs' advantages, such as acceptance and relation with marginalised groups, organising participation, quick and flexible response, and cost-effectiveness. Even with such features, there are a number of arguments that suggest that NGOs perform differently to the aforementioned assumptions. Bodja (2006, 20) contends that NGOs are accused of hijacking genuinely needed development aid by well-established élites. These élites are also seen as new structures to perpetuate existing systems of neo-patrimonialism in poor communities. NGOs, mainly international ones, have expanded their structure that they have lost their efficiency in terms of delivering cheaper and quicker service. Moreover, these NGOs are often blamed on the grounds of misunderstanding local conditions and hiring amateurs with less skill (Bodja 2006, 20), which interrupt the target of community empowerment.

5.5 Social Capital and Community Empowerment: New Paradigm for NGOs' Development Activities

Over the last three decades, three major schools of thought have dominated research on underdevelopment, namely growth model or stage theory, modernisation theories and dependency model or world system perspective. These 'borrowed concepts' and 'conventional development theories' and practices characteristically have come from outside. In such schemes, knowledge of and links with the recipients are virtually minimal. These not only imply ineffective 'transmission of development' but also indicate little recognition of the process of internal generation. VanSant (2003) describes that there are three ways by which the NGOs deliver their activities. First, relief and welfare—involves primarily the delivery of inputs, such as humanitarian assistance. Secondly, small-scale local development—involves service delivery and building of some local capacities for self-help. And thirdly, sustainable development systems—involve grass-roots mobilisation and policy advocacy.

Due to globalisation, the concept of development regarding the role of NGOs needs to consider some important indicators. The former Secretary General of the UN Kofi Annan (2001 in VanSant 2003) says, 'Today's real borders are not between nations, but between powerful and powerless, free and fettered, privileged and humiliated'. I might add that

development is often the ticket across these borders. Networking as a part of social capital is an aspect of globalisation and is a major strategic device for NGOs. Networks contribute to adaptability and problem-solving strategies. Networks can more swiftly discover and adapt new techniques. Networking, as John Clark (in VanSant 2003) notes, does not mean working only with like-minded groups; it means building partnerships to tackle issues that would be impossible without particular strategic alliances. In this connection, some new concepts, such as community empowerment, social network, social capital, self-reliance and sustainable development, have become important components and objectives of development. It is because of this that the change of NGOs' role, from relief to development (Davis 2006, 4), and from development to sustainable development, is now getting more attention. This approach is now overemphasised with a growing recognition of the role of non-government development organisations. In the context of globalisation, NGOs need to focus on expanding space for poor people (about 80 % of developing countries) to engage with emerging forms of states and markets for their livelihood advantages. This implies creating space, where the poor can effectively negotiate the global discourses and frameworks of development from their particular experiences and images of reality (based on local knowledge) (Tembo 2004). Within this discussion, I argue that the NGOs' interventions are now needed to reshape from the conventional approach. In this connection, there are three important development indicators which help to reshape the content of NGOs' development activities, where a new definition is emergent. These are:

- the great involvement of NGOs in development sectors in community level as a part of globalisation,
- the emerging of social capital with huge prospectus and
- NGOs' community empowerment movement towards sustainable development and self-reliance.

Within livelihood assets and institutional processes, 'knowledge' (including information) is the key indicator, which helps to achieve community empowerment. It is accepted that knowledge, in all its forms, is a means to the attainment of economic self-determination and self-reliance. Every aspect of human activity, in terms of the use of knowledge, is becoming a vital component of socio-economic and political interaction. This system evolves through human interactions, as well as with nature. The NGOs are the key actors, which provide knowledge as their development interventions to the community people. Here, the main interaction happens between community people and development agencies (NGOs) (Tembo 2004). The NGOs' role for developing countries, such as Bangladesh, is more important, when a significant number of populations are out of this knowledge system. The most important thing is that knowledge can be considered as powerful when it can solve the practical problems of the community. Simply, this knowledge is called social capital, which can be invested directly for community well-being. It is accepted that the ultimate purpose of this knowledge, which is disseminated as interventions by NGOs for developing countries like Bangladesh, is to generate social capital and community empowerment for social development. Which knowledge approach should be used, either local or global or an integrated approach, is important for such a development. However, this is a debateable agenda, and globalisation makes this complex, where the NGOs' role becomes challenging.

Many research papers have been published on NGOs' role in sustainable development, where NGOs' capacity for social capital with IK's role has been criticised meticulously due to globalisation. Mathias (1995) claims that NGOs still appear to make little use of IK, though it is considered a valuable resource for social development. Roue (2003, 536) even finds that the fundamentalism of some NGOs does more harm than good to indigenous populations. Tembo (2004) calls it 'an actor-oriented development approach', which has failed to create a proper negotiation between global knowledge and local experience. Donor dependency is one of the ubiquitous criticisms of NGOs' activities, which is considered a big threat for indigenous initiatives (Devine 1996). Briggs (2005) argues that much effort is expended on designing and disseminating 'solutions', but too little time is spent on understanding the problem. Sillitoe (2000) claims that the confusion between scientific and indigenous knowledge in Bangladesh occurs in NGO level because of misunderstanding the IK agenda and the attitude of scientists. In keeping with these views, I have seen that there are three different strategies where the NGOs use knowledge approaches, such as 'patent', 'transfer' and 'mixing'. Each of these three strategies has been practised in Bangladesh, but the positive community change has still not been achieved. The third one gave some comparatively positive outcomes in many countries, but still there are few examples of where it has done so successfully. Stephens (2000, 11) shows how a new knowledge paradigm can be developed through integration of IK and GK. My argument is that it

is ideally possible, but the application of this new knowledge would be difficult to achieve its desired success. For example, Barr and Dixon (1998) show how the farmers' and fishers' local knowledge failed to integrate with the scientific knowledge into natural resources in Bangladesh. They argue that there were considerable intellectual difficulties and practical problems with this integration. However, there are some examples where this integration gave a positive outcome. For example, Howell (2003, 8) shows that this combination can offer a more appropriate, reliable and comprehensive early warning indicators of cyclones for vulnerable rural people in Bangladesh.

Very few research-based documents are available in Bangladesh. Khan et al. (2000) argue that the research and documentation of IK are still at an embryonic stage in Bangladesh. They provide a list of IK-related literature, but most of them are in forestry, agriculture, flood and irrigation, livestock, health and management. None of those is methodological or purely general development issues. However, some progress was found and some countries achieved some success, but these initiatives have been criticised. Moreover, the distribution of resources, power relations, sustainability of development, stand of culture and public health status is questionable.

Let us look at the literature, which considers how a new knowledge framework is possible, what the constraints are and how we can overcome them. For example, regarding rural development through agriculture, Briggs and Sharp (2004) show that this new approach should be matched without any challenges of its content, structure or value system of this view with modern science. One of the probable gaps is scientific language, which might be inadequate for describing the complex and changing experiences of farmers and other actors in rural development. Briggs et al. (2007, 247) examine the implications of the local environment knowledge in two Bedouin communities in Egypt and find that the traditional IK-based environmental knowledge cannot work successfully. Instead, they recommended considering it with economic and sociocultural realities for sustainable development strategies. Tembo (2004) identifies these as 'ideological', 'language' and 'image conflict'. As a result, 'knowledge', other than those derived from formal science, is still observed doubtfully by many in the development community. It is feasible to be incorporated into development practices where indigenous technical solutions are straightforward and uncontroversial. But I argue that this attempt may not be possible for developing countries such as
Bangladesh, where sociocultural variations are enormous. In this connection, Craig et al. (2000) claim that lack of scaling out local capacity building and empowerment policies for 'scaling up' to the larger regional or national level could fail. Because of the lack of local roots, the large-scale plans can be consent from above, but 'participation' cannot be imposed from above. It develops differently at different paces, depending upon local context.

I argue that an ideal policy framework should be developed using both knowledge systems with consideration of local context, local resources and capital, local peoples' capacity, local institutional capacity and local people's needs. It should be developed through authentic people's participation, the community's ownership and within an environment where a joint/partnership initiative can run as a learning process. It is a new form of knowledge approach, which is the principal thrust of the research investigation. I argue that it is neither possible by patent nor transfer—neither possible by mixing nor integrating. I think that mixing is a way in which emphasis is given on the basis of both. It offers neither sustainability nor self-reliance.

Social capital has become an important analytical concept and policy tool within development (Bebbington 1997). From an analytical perspective, several researchers have tested whether social capital does underpin successful economic development and/or poverty alleviation programmes. Kay (2005) agrees with the importance of social capital for community development and considers it as 'social energy'. He states some of the results to emerge from a 3-year research project (CONSCISE Project) in Western Europe. Kay looked at the contribution of social capital and found that social capital made a great contribution towards community development. He found it as 'glue' (binds a community) and 'grease' (enables things to happen smoothly), which enhanced community mobilisation, capacity building, planning, partnership, community renewal, place management, social entrepreneurship and sustainability (Kelly and Caputo 2005; Ingamells 2006). In this way, social capital has been linked to collective norms, values and relationships reflecting the involvement of human individuals in 'a common life based on family and community'. Here development is generally defined as a social learning process, which serves to empower individuals and to involve them as citizens in collective activities aimed at socio-economic development (McClenaghan 2000). Mohan and Stokke (2000, 247) argue that development based on local and social

capital approach has moved away from holistic theorisation towards more localised, empirical and inductive approaches. A more liberal and populist approach to local empowerment centres on such an open-ended interpretation of participation described as a 'paradigm shift'.

As a non-threatening language of trust, networks, reciprocity and associations, social capital binds communities together and ensures both political and economic progress (Mohan and Stokke 2000). Social capital is the density of formal and informal institutions, which links other kinds of inputs into development and brings them centre stage. Importantly, these analysts perceive social capital as a vital local endowment leading to local development. At the level of macro-policy, the World Bank sees social capital underpinning a wide range of developmental initiatives (Fox 1997). The broad objective of social capital is to secure an asset-based community, which is a relatively new approach for participatory development. This process helps to join up environmental, economic, social, demographic, technological, political and other issues by empowering communities. In this regard, social capital becomes an important analytical concept and policy tool within development (Bebbington 1997; Brown and Ashman 1996). It is a vehicle which strengthens the bond with social networks. And the NGOs' role in this context is crucial. I argue that NGOs can bridge the gap between the government and the community, organise poor people, take collective action, fight for peoples' rights and represent the interests of the community members, facilitate the supply of inputs into the management process, mediate between people and the wider political party, networking, information dissemination and policy reform. For increasing social networks and community participation, the NGOs should respect the community's indigenous contribution as manifested in their knowledge, skills and potential.

Empowerment is also considered as an important outcome of community development (Barnes et al. 1997). Kamata (2000) argues that after the failure of top-down and blueprint development model 'participatory development' has been a frequently discussed topic for the new model of 'people-centred' development. The concepts of empowerment and IK have been introduced into the context of the developing world and are becoming widespread. Decentralisation (Brinkerhoff and Azfar 2006), citizen participation (Eklund 1999), downward accountability (Kilby 2006) and capacity enhancement need assessment (McNeil and Woolcock 2004) under community empowerment are now considered important aspects for community development. Stubbs (2006) shows in his case study in Croatia that alongside concepts of globalisation and neo-liberalisation, a third process, that of NGOisation, needs to be addressed for successful development. Rifkin (2003) presents the progression of the concept of participation to the present term of empowerment, and the links among empowerment, equity and health outcomes. I stated earlier in this section that recently Laverack (2006) provided two case study examples of how community empowerment can be better conceptualised, planned and applied within a programme context by using a 'domains' approach in community development practices.

5.6 Conclusions

The chapter discusses different aspects of social capital and community empowerment with the role of NGOs for development activities. Despite many contradictions and criticisms, it is clear that indigenous and global knowledge systems have been considered as powerful tools for development. There are many aspects of the approaches of social capital and community empowerment where NGOs can play a significant role for sustainable development. I argue that there are still many opportunities for the indigenous NGOs where the application of these approaches can be justified, and these should be researched. This is also a concern that many NGOs still cannot reach the local poor people properly. On the other hand, globalisation brings many prospects as well as tensions, which may reconsider the process with NGOs' interventions. It appears that a qualitative investigation would be useful to formulate appropriate locallevel planning for reaching local people and their development. However, I do re-emphasise three important aspects from the aforementioned discussion. First, the NGOs' role towards social capital and community empowerment is significant. Secondly, globalisation creates a number of opportunities for development, which we should consider according to the local conditions rather than applying straightforward, otherwise this approach may destroy many IK-based resources. Thirdly, it is a crucial policy issue that NGOs' development activities should follow an appropriate policy as an effective contribution towards socio-economic development.

NGOs' Capacity for Social Capital in Bangladesh

Abstract This chapter discusses NGOs' capacity for social capital development with the evidence of two NGOs: Practical Action Bangladesh (PAB) and Proshika working with two smith communities such as blacksmiths and goldsmiths located at Mostafapur Bazar, Faridpur, and Mirpur (1) Market, Dhaka, Bangladesh, respectively. The findings looked at the two programmes: the Markets and Livelihoods Programme (of PAB) and the Small Economic Enterprise Development (of Proshika) programme. Data were collected through a qualitative case study, where a multimethod data collection procedure, such as semi-structured interview, social mapping, participant observation, in-depth case study, focus group discussions and documentation survey, was employed. The Practical Action (formerly Intermediate Technology Development Group) was established by E. F. Schumacher, the author of the book Small Is Beautiful (1973). It started working in Bangladesh in the early 1980s in response to requests from a number of prominent national NGOs. In 1990, a Bangladesh country office was formally established with its head office in Dhaka. From 1997, PAB began implementing large-scale programmes in the technology areas of food production, agro-processing, small enterprise development and light engineering. The vision of the organisation is 'a world free of poverty and injustice in which technology is used to the benefit of all'. The mission aims 'to eradicate poverty in developing countries by developing and using technology, and by demonstrating results, sharing knowledge and influencing others'. On the other

hand, Proshika is one of the largest NGOs of Bangladesh, which started out in a few villages of Dhaka and Comilla districts in Bangladesh in 1975, although the organisation formally took its first step in 1976. The central ethos of Proshika is human development and empowerment of the poor, who gradually stand tall to achieve freedom from poverty by themselves. The working areas of Proshika cover most parts of Bangladesh. The main activities of Proshika are people's organisation building, human development training, practical skill development training, universal education, urban poor development, development policy analysis and advocacy, employment and income generation, small economic enterprise development, organic agriculture, livestock development, fisheries development, social forestry, irrigation and tilling technology service, sericulture development support communication, people's culture, disaster management and preparedness, policies for risk and vulnerability management, integrated multisectoral women's development, computer in development, assistance to other organisations, research and demonstration project, and legal aid services.

Keywords Proshika · Practical Action Bangladesh (PAB) · Poverty alleviation · NGOs · Social capital · Social network · Mutual benefits · Enterprise development

NGOs' capacity for social capital is a relatively new area in development studies (Islam and Morgan 2012a, 2012b; Islam 2015a, 2015b). This approach is gaining increasing attention within the sustainable development paradigm, where the NGOs' microcredit versus macrocredit debate reaches its peak (Islam and Morgan 2012a). The literature on social capital suggests that NGOs' capacities are not limited to providing financial help to poor people, but also to creating capital available in the community, so that a community can be empowered, on the one hand, and the community people can use this capital for their development, on the other (Islam and Morgan 2012a). The local institutions are good sources of social capital. Putnam (1993, 1994) and Evans (1996) argued that the elements of social capital are features of social organisations, which facilitate, coordinate and distribute mutual benefits for the community people. In their recent studies, Mondal (2000), Huda et al. (2005), Mannel and Chowdhury (2005), Kabeer (2003) and Purvez (2003, 2005) find that

Bangladeshi NGOs are creating significant levels of social capital. Yet Mondal found that both governments' and the NGOs' social capital formation in Bangladesh is little known. This section considers NGOs' capacities according to the four fundamental components of social capital, namely collective action (interaction between individuals, groups, neighbours, communities and agencies), social trust, coordination and cooperation of mutual benefits, and sharing norms and values.

There were a number of areas, where both NGOs worked to create social capital. With the diverse information and opinions supplied by all groups of stakeholders (NGO staff members, community leaders and smiths), I present the common and general working areas of both NGOs. These included survey and PRA, group formation, family approach, information sharing, meeting and consultation, participatory plan, good staff behaviour, training, programme coordination, market chain, market map, link and work with local GOs-NGOs and community leaders, organisational reputation, jobs and income generation programmes (IGPs), cultural activities and social festivals, observation, communication, counselling and advocacy, monitoring and supervision. I highlight all of these areas, using some broader headings with some examples of NGOs' interventions, where these were frequently used. I acknowledge that it is very difficult to point out individually which NGOs' intervention creates a particular component of social capital, as one component is interrelated with others and therefore problematic to summarise.

6.1 PARTICIPATORY PLANNING: CREATE ROUTES FOR SOCIAL CAPITAL

Both NGOs constructed a certain level of social capital. Both NGOs followed the 'participatory planning' (PP) under the 'bottom-up approach', which is considered a scientific approach for social development. The advantages of this PP were that it was a process framed through consulting a number of groups, including themselves, and validating diverse groups' opinions and needs. Along these lines, the blacksmiths and goldsmiths found ways to exchange their views or create social networks. These networks increased opportunities for these poor people to improve their ability to work together, providing a safety net for social survival through being part of a group (Purvez 2005, 95). It was more than just public participation, but rather a spectrum of processes, such as involvement, engagement, negotiation, pre-mediation and mediation.

This route enabled these communities to plan activities by using their own traditional human and financial capital. This PP was a valuable tool for integration between local and other institutions, including programmes of NGOs. It established links between different levels of society and government administration, such as the community, Union Parishad (UP) and with the district. This kind of plan also improved peoples' motivation, learning and self-realisation, feelings of ownership, self-esteem and assessment capacities; these were helpful to identify problems and solutions. It generated constructive collaboration among stakeholders, who might not be used to working together, who often came from different backgrounds and might have different values and interests.

The data found that the collaborative collective action was motivated not only by self-interest, but also through mutual trust, which strongly promoted participatory development, social capital and empowerment through horizontal alliances (Mondal 2000). The NGOs' first linkage starts with the community members, community leaders and local institutions through a community survey. The field staff members took opinions from the community people and from other groups and institutions, such as UP, Thana, district, civil society and local GOs-NGOs. The staff members said that both NGOs provided their supports to a group rather than towards to the individuals, who had a business, business knowledge and experience, and cash capital. This participatory plan gave opportunity to the staff members to explain the reasons and the impacts of participation to the programmes. At the end, most people viewed broader participation as positive because this plan gave them a voice beyond the community leaders. The NGOs continued to share values and norms with blacksmiths and goldsmiths through their continuous supervision and monitoring services, though they had finished their particular project with them.

By using this participatory approach, both NGOs were successful in achieving the socio-economic improvement, reducing programme costs and ensuring that needs were met. Moreover, the NGOs created social capital by forming new norms of trust and association, largely between themselves and the smiths. However, the networks and formations of trust were mediated by the agency practitioners, and therefore created new bonds of dependency. The blacksmiths and goldsmiths were not fully clear about the objectives and goals of this PP. They found this plan complex and doubtful. In this regard, Buckland (1998, 237) argued that successful NGOs need to be both

lauded for the successful extension of social capital, but also implored to work at further localising it. In addition, one of the problems was that the communities in Bangladesh are not homogeneous. There were huge gaps in terms of social roles, power relations and social positions between and within different groups of the communities, such as local leaders, general community people, civil society and local administration. This argument was true in both NGOs as their approach demonstrated 'group capacity building' for collective action as the most difficult to accomplish, but was 'at the heart of the development process'. A group capacity was 'the ability to act together, not once, but consistently' to 'get ahead collectively'. Nevertheless, both NGOs were rarely evaluated on their group capacity building, largely due to measurement difficulties. However, the whole process of participatory plans created many rooms, where different elements of social capital were found.

6.2 Channel of Communications: Sharing Values, Norms and Mutual Benefits

There were a number of working areas where both NGOs developed channels of communications based on IK and GK interventions, including interpersonal contact, business networks, consultations, face-to-face meetings, sharing values, visits, training, monitoring and supervision. In all cases, the NGOs established this communication as the way of horizontal relationships with smiths, community leaders, local institutions and with their colleagues. The NGOs introduced new knowledge and technologies via traditional and informal channels of communication through interpersonal contacts, including family, friends and colleagues. The business networks began through such interpersonal contacts, which provided social support and self-confidence. A firm's trade network, concerning links between smiths and users of these goods and services, and its knowledge networks focused on the flow of information and exchange of knowledge (Malecki 1998). Such kinds of business projects were transpired through three types of networks, each serving a different function for the entrepreneur: networks for service and assistance, for example, to solve a specific problem; networks for information and structuring, especially for knowledge about whom to contact for a specific purpose; and networks for entrepreneurship and product development. However, I found that local entrepreneurs' networks and local traditional trust-based

relationships were critical, and it could not function as well as the NGO workers expected. In the absence of trust, communication was less frequent and full cooperation did not occur (Malecki 1998).

In the office environment, both NGOs followed a horizontal type of communication pattern, where everybody referred to each other as bhai (brother) apa (sister) among Muslims or dada/didi¹ among Hindus, as it was their office decorum. This norm was followed from the top to bottom. Both NGO staff members said that they had very good relations with their colleagues, and that they were helpful in exchanging information, knowledge and experiences. Both NGOs had face-to-face consultation meetings, where the smiths shared their knowledge and experiences with each other and benefited from this group environment. The staff members generally found these group activities to be dynamic processes for sharing information, updating technologies and improving skills among the smiths. However, I found that the smiths had difficulty with this kind of formal communication. Many of them could not understand the meaning and importance of such communication due to their poor education, and they partially failed to share their norms and values. Though within a group, an individual had the chance to develop both human and social capital by observing and practising group rules, group principles, values and norms, leadership and management skills. For instance, Proshika formed approximately 18,417 Group Federations² up to December 2007 all over Bangladesh (Proshika 2008). These federations brought people together in cooperation on mutually beneficial projects, thus enhancing communication, reducing objectification of others and increasing interdependence and trust. As a part of the institutional involvement, both NGOs permitted the smiths to use them as referees as many blacksmiths and goldsmiths got loans from NGOs, such as Proshika, ASA and Grameen Bank.

The frequency of face-to-face interaction through monitoring and supervision and group meeting among smiths and staff members generated common social norms ('social glue') and created predictable behavioural patterns (Svendsen and Svendsen 2004, 28). This process gave a sense of 'belonging' to the organisation, establishing trust amongst them in helping to develop a stake in organisational goals. Significantly, the meetings demonstrated an enormous amount of 'social energy' (Mannel and Chowdhury 2005, 1). The decision-making, reporting and internal learning operated as closely intertwined processes within the organisation. The minutes of meetings, periodic activity reports and reports of visits across the organisation were maintained in written form. This practice created a constant flow of information throughout the organisation, both vertically, from centre to field and back to the centre, and horizontally, across subcentres, area and divisional levels. In this way, social capital defined as people's ability to cooperate enhanced economic growth in society (Svendsen and Tinngard Svendsen 2004, 31).

Both NGOs had different events for improving collegial relations. For example, the Gana Unnavan Procesta (GUP), the partner NGO of PAB, held a meeting called *shakal shava* (morning meeting) before starting their actual office work, where they openly discussed their office matters. I found this a good practice. Accordingly, SEED had a weekly staff meeting, where they did the same. Their monthly area staff meeting at head office was another event, where all area officers shared their observations, thinking and experiences in terms of what they were able to do, what they could not do and what they would aim to do in future. Then the head of the area officers passed this information to their area office colleagues. The relationships of shared values were not simply contractual, but based on a deeper collective commitment to social change. NGOs, donors and communities shared this vision for positive change through discussions on what constituted a 'successful' development intervention (Ebrahim 2004). Both NGOs claimed that they had some environs where they shared their agency values and norms with other agents. The Human Resource Policy Manual of PAB (Practical Action Bangladesh (PAB) 2005, 15) denotes that Proshika has a human resource information system which ensures that 'the right person is in the right job'. The meetings and workshops helped to share the concerns, ideas, learning processes and experiences within the two or groups of employees. These benefited staff members not only by reducing the gaps, but also by improving and developing the employees' level of knowledge, teamwork, attitude, common objectives, values, ownership, participation and thereby ensuring appropriate conditions to achieve the organisational mission and goals.

Sharing IK with others, so to speak, meant that knowledge that was shared was respected. Sometimes this required a lot of care and attention. As Cajete (2000) pointed out, modes of indigenous experience are evident, not only through collective cultural expressions of art, stories, rituals and technology, but also through the more subtle and intimate expressions of individual acts of respect, care, words and feelings that are continually extended. In this connection, I found that the smiths' as well as the NGOs' practical knowledge was highly valued, stored and connected,

and used for other projects carefully. For example, the PAB staff members said that because of experience PAB's local resource collection techniques, service delivery strategies and management quality had improved. On the other hand, the smiths were more cooperative; they felt freer to share their experiences for their common benefit.

Both NGOs also encouraged effective partnerships with other local institutions, NGOs, GOs and donors to share their processes and values in searching for 'collective actions'. Both NGOs' partnership with donors was helpful in developing their practices, where both parties benefited from each other. With this connection in mind, Dave Brown (Brown 1994; in Ebrahim 2004) emphasises the *Mutual Accountability Model*³ to make partnerships with NGOs and donors, which is, perhaps, the most challenging to develop, but also the most suitable for creating 'partnerships' among organisations. It is helpful in establishing a joint commitment among actors to a shared vision and goals in bridging an understanding from multiple perspectives, in building a collective capacity for social and institutional learning.

PAB's 'exposure visit' was one of the more successful communication events, where the local blacksmiths met urban blacksmiths and businessmen. In this way, the smiths increased their social networks, business and human skills, got new knowledge and renovated their existing knowledge and skills. They knew the urban market, urban industry and urban shops, and got exposure to urban life and society. For example, a blacksmith got new knowledge and skills from other blacksmiths. It was an opportunity where local blacksmiths were able to meet the urban wholesalers and obtain orders. On the other hand, a wholesaler was able to find the smiths and bought products at comparatively cheaper prices. This channel created a new 'business route', where the blacksmith coordinated his traditional skills with global knowledge and demands, and this IK could be integrated smoothly with GK. In this way, they found 'a new way of life and culture'. The PAB thought that it had significant social impact for sharing understanding about the urban community and new knowledge rather than just to build links with the urban community. This process helped to improve confidence and sometimes their mobility encouraged them to migrate to the urban community.

Both NGO staff members and smiths said that training interventions were helpful in creating human and social capital, such as improving skills, management, personal confidence and overall social awareness. In general, training is an exposure-changing programme for improving skills and working experience. It introduced a new thrust of knowledge and increased a certain level of self-confidence. It found that this exposure could spread among other colleagues through 'cascade training'. Poston (1994) conducted a study in Sub-Saharan Africa and found that training systems, sharing local knowledge and traditional apprenticeship systems were more appropriate and contributed to a community's capacity. It was helpful to build up networks with the local smiths in connection with community leaders, local governments, local NGOs and cooperatives, and local business communities.

6.3 Relation Matrix, Market Channel and Market Map: Examples of Collective Actions

Both NGOs used the 'market/business channel', which was considered as an important social network. It enabled the smiths to enter into a competitive market, where they benefited through collective actions. They built up circles with different community people, community leaders, business partners, wholesalers, designers and suppliers. They had an opportunity to justify their existing skills. For example, when a blacksmith entered a market channel, then he needed to know where he got his raw materials (such as coal, iron, steel and electricity). Then he needed to build up relationships with customers, local markets and suppliers, where they sold their products. Sometimes, as I stated earlier, they made contacts with the wholesalers, who bought their products and then sold these in the urban markets. This process was helpful to generate opportunities for gaining social, human and financial capital. This is because, when a smith travels the market map, s/he has to follow a number of channels at different levels, which helps to acquire such capital.

PAB observed that markets matter to the poor. It was made increasingly clear that in tackling rural poverty market-related issues, including access to information, institutions, linkages and trade rules, were vital considerations. Failure to address these issues meant that the benefits of other developments threaten to bypass the rural poor. In considering this, PAB adopted market chains for mapping the market. The market chains comprised of economic actors, who produced and transacted a particular product, as it moved from primary producer to consumer. They included smallholders and larger-scale smiths, input supplies, traders, processors, transporters, wholesalers and retailers (Albu and Griffith 2005). The map included three components, such as market

chain actors and their linkages, enabling business environment factors, and business and extension service providers. This was a key route, which different groups could travel and access new knowledge bases and experiences. The aim was to identify inefficiencies, inequalities and losses, which could be remedied. The smiths could find the inequitable relationships between actors, and then help stakeholders realise mutual benefits by improving the system. The second component helped to find the 'enabling environment' factors. This knowledge could help determine avenues and opportunities for realistic action objectives, lobbying and policy entrepreneurship. The third component is the range of services that could potentially add input supplies, market information, financial services, transport services, quality assurance-monitoring and accreditation as well as technical expertise and business advice; veterinary services; and support for product development and diversification. The mapping 'services' identify particular service needs and their locations within the market chain in order to get an overall picture of the opportunities to improve market chain efficiency or equity. This mapping was a 'precursor' to subsequently assessing the most appropriate mechanisms for delivery of services, in terms of outreach, sustainability and costeffectiveness.

The market map proved a very useful way to visually represent and concisely correspond with knowledge concerning specific market chain actors, operations, contexts and needs to different stakeholders. The participatory market chain analysis (PMCA) is a key approach in operating the Market Map framework-changing the conceptual framework into a practical tool, which facilitated efficiency, improved coordination, stimulated innovation and reinforced trust within the market chain. This process of exchange worked based on bonds and trust (Purvez and Ahmed 2003). Actual and perceived disparities of power within the market chain could, however, get in the way of participatory analysis. In practice, the blacksmiths of the Mostafapur Bazar were not much aware of this market map, though they agreed that they understood that they were better organised regarding their trade than previously. The staff members stated that they could not organise them under this formal market map system, but they believed that the blacksmiths benefited from this market map in many ways. For instance, it facilitated the blacksmiths' sharing of information and reduced transaction costs. One solution was for facilitators to orientate weaker participants in advance so that they understood their role in these events and had realistic expectations of PMCA process outcomes.

Therefore, it proved that skills, knowledge and organisational capability through the market chain helped to earn greater income, more reliably, and sell products and increase profit in existing and new markets.

6.4 IGPs Versus Social Trust

At present, the IGP is an important indicator for sustainable development with a view to making people more self-reliant and their livelihoods more sustainable. One of the common and major interventions of Bangladeshi NGOs is to create IGPs (Islam et al. 2005, 17). Bangladeshi NGOs have been successful in this area, which results in socio-economic improvement. Both NGOs generated a significant number of income activities. The success of these programmes was related to the growth of social capital, in the form of new norms of trust and association, largely between the agency, practitioner and the participant (Buckland 1998, 237). Trust is at the very heart of the problem of social order and is crucial to the conduct of everyday life (Cook 2005, 6). Large-scale social exchange is not likely to occur without formal agreement unless firm social bonds, rooted in trust, have been established. Trust also brings all sorts of good things, from willingness to get involved in communities, to higher rates of economic growth by job creation (Putnam 1993, 1995; Knack and Keefer 1997). Due to the high unemployment rate in Bangladesh, we found that these IGPs helped to create 'trust' among smiths and NGOs. Trust is considered here as 'the chicken soup' of social life. It was observed that a particular solution was tailored depending on local circumstances to build other types of trust. The NGO staff members said that people often trust them more than they did members of their own community; it indicated that families would tell an NGO worker about a problem before telling a friend or neighbour (Abom 2004, 350). When trust was built up between actors or agents, it led to beneficial action through mutual cooperation (Purvez and Ahmed 2005).

The NGOs provided income facilities both by recruitment in the organisations and IGPs in the communities. These had a wide range of impacts on financial, human and social capital in the society. Proshika had 7,138 staff members in June 2007. It covered 24,181 villages, 2,109 slums, 1,919 unions (rural) and 328 wards (urban), 234 *Upazilas* and 59 districts until December 2007. Total primary group members were 2,804,287; households 2,146,985; poverty-free household 1,236,074; and beneficiaries 11,808,421. It is noted that it created 10.51 million

employment/self-employment facilities and it had 20.52 million human development training and 1.16 million practical skill development training participants (Proshika 2008). On the other hand, the PAB's reducing vulnerability programme successfully trained 5,265 beneficiaries, Markets and Livelihoods Programme 155 and infrastructure services programme 75 in 2006, and they claimed that they created a good number of income generation facilities in Bangladesh. An enterprise development enabled the poor beneficiaries to come out of poverty and settle on a sustainable basis. Chaudhury (2006) says that, on average, the beneficiaries of Markets and Sustainable Livelihoods (MSL) (PAB) were able to more than double their income within 3-4 years after receiving training and other business-related services from the project. A majority of the beneficiaries expanded their existing business/enterprises (71%), and more than a quarter of them started new businesses (26%). Through conducting a survey of 39 entrepreneurs in Dhaka City, Small Economic Enterprise Development (SEED) (2006) reported that after receiving interventions from SEED, the smiths' monthly average income was nearly doubled, and socially they were in a good position.

From the aforementioned discussion, this chapter found that the notion of social capital has been articulated as a concept, which is homogeneous and recognises primarily 'positive' social norms, values and practices. The fact is that the communities in both NGOs were rarely discrete entities, and one had greater power to define and legitimise meaning, values, norms and practices than another. Bangladesh and other poor countries are immersed in multidimensional relationships, whether these are hierarchical or reciprocal (Wood 2005). I observed that access to community membership and power relations was structured intra- and intercommunity conflict and competition. I can consider the statement of McClenaghan (2000, 571), who suggests that Putnam refers to the crucial 'bridging function' of social capital, but gives little attention to inequalities in the distribution of income or other economic, physical, environmental and social advantages, which structure relations in the field of community life. This finding can be compared with Bayat (2005, 10), who stated that the lack of consideration of power as an important shaping feature of social capital might lead to horizontal associations, while neglecting vertical relationship. Clearly, the traditional associational ties and forms of closure to which Putnam refers may well contribute to the reproduction of these advantages and, by implication, the reproduction of disadvantage and exclusion. Nevertheless, I observed that the social inequalities in terms of loan distribution, distribution of wealth and

resources, power relations, social control and decision-making in both communities made it difficult to get the benefits of social networks. This process clearly reinforced the very exclusion that social development activists in disadvantaged smith communities were struggling against to those of groups that are more powerful. Sometimes the social network or relationship pattern was seen more or less linked with interest-gaining processes, which sometimes stopped the natural flow of social relationships.

The chapter found that social capital is the fabric of society. However, the chapter supports the argument that the networks trust and shared values of social capital brought human values, knowledge, skills, expertise of blacksmiths and goldsmiths from their daily communication between and within their neighbours. These appear to be the raw material of communities as Coleman (1988) mentions. The blacksmiths and goldsmiths used their social networks, norms and values, and collective actions through their everyday interactions and contributed to economic and social development. These interactions were considered as 'bottom-up' phenomena, which were derived from their social connections and networks, such as trust, mutual understanding and norms of action. I found that, though there were some group conflicts among the local leaders, but their social connections and networks helped to build up a common sense of 'civic' responsibility that integrated them more than a collection of individuals. As a result, both communities improved 'we' feelings attitude and they understood the positive aspect of group interactions. I believe that, without a degree of common identification with forms of governance, cultural norms and social rules, it is difficult to imagine a functioning society as mentioned by Grootaert (1998) and McClenaghan (2000).

However, the contributions of social capital were found in the study as follows:

- The norms and values of blacksmiths and goldsmiths were the form of the basis of their social identity as they earned these distinct characteristics through their daily practices. The community people recognised them in this way.
- The social interactions and networks provided them the instruments for productive output, which started their skills learning process to product marketing. This process brings social mobility, alliance, social adjustment and adaptation capacities, and improved their leadership quality.

• The social trust and cooperation offered them an indispensable unity in the social context to ensure that collaborative efforts were integrated in a sustainable way, where social cohesion and harmony were mutually beneficial and important for their ultimate output.

In networks-based resources (networks referring to Coleman's 'structure of relations'), social capital was seen as significant in an economic development to the smith communities on marketing networks. The external resources, such as loans, training, counselling, monitoring and supervision, were essential for such kinds of small enterprises. However, cooperation was most important to link with one another that the NGO provided to the smith communities. These kinds of enterprises run through cooperation in a number of market channels. Indeed, the constituent elements, such as trust, norms and networks, that they achieved from their communities were important to maintain their day-to-day life. This finding is supported by Malecki (1998). Buckland (1998, 241) asserts that trusts, norms and networks are essential for achieving indigenous efforts at community development. Local social cohesion and mobilisation ensured the vibrancy and success of both local community and broader state development projects. The local cohesion and mobilisation were viewed as the willingness and capacity to cooperate and engage in collective action for mutual benefit. Like Woolcock (2001), this chapter found both bonding (between close family members, friends and neighbours) and bridging (between more distant associates and colleagues) social capital, but it was observed that social capital was not created in a direct way; rather within the market map, the local producers used the metaphor of various kinds of commodities from consumer goods to durable goods. Then social capital took time to build and hence represented an investment for future return, such as forging relationships or membership in associations, but differed from other investments in that the process itself could provide satisfaction.

6.5 Conclusions

The chapter has identified some critical aspects of NGOs' interventions, where social capital approach is more feasible among the smith communities in Bangladesh. The analysis shows that the NGOs were facing some contradictions to create social capital because of the low level of education of their beneficiaries, poor socio-economic and cultural conditions,

non-cooperation from community leaders, lack of government initiatives and NGOs' donor dependency. I observed that the NGOs partially failed to create a congenial atmosphere in the market, where the smiths could sell their products. In some cases, the NGOs could not build up effective networks with the negotiation with other community people, GOs' initiatives, community leaders and with the external donors. In addition, the evidence shows that NGOs' political networks had developed a negative trend towards human services. More or less both NGOs had money-laundering tendencies, and they had leadership and management crises, and donor dependency. It is important that both NGOs recognised the spontaneous participation problem from the community people and community leaders. In this connection, NGOs' social trust with community people and GOs became hard-nosed. Still we have seen that the NGOs' contribution in social capital development was incredible in some areas, for example, participation of the smiths in the development activities, social networking and sharing mutual benefits, collective action, leadership and income generation. However, the overall assessment of NGOs' capacities was convincing in terms of community awareness, sharing values and norms, introducing new technologies, increasing human rights and problem assessment capacities, responding to financial crisis, encouraging participation and mobilising local resources.

Notes

- 1. Bhai/apa and dada/didi translates as brother and sister (with the former referring to a Muslim dialect of Bangla and the latter referring to a Hindu dialect of Bangla) who themselves think that they are equal to each other who hold the same power with their different responsibilities. In one sense, it is a relation where 'power relations' are minimised.
- 2. The federation formed a group of people which is formed by Proshika. It created this federation from a union/ward level to district levels. It has formed 18,244 group federations (14,899 village federations, 1,863 slum federations, 1,316 union federations, 136 upazila federations and 30 urban area federations) till June 2007 (Proshika 2007).
- 3. In which multiple actors seek to build shared values and goals as a basis for joint action and where there are compacts of mutual responsibility rather than formal contracts.

NGOs and Community Empowerment in Bangladesh

Abstract This chapter discusses NGOs' capacity for community empowerment with the evidence of two NGOs in Bangladesh, which were discussed in Chap. 6. The concept of 'empowerment' is one of growing interest in development discourse. It is a process through which individuals struggle to reduce personal powerlessness and dependency and increase control over the circumstances in which they live their lives. It is advocated by prominent agents in economic and social development and especially by NGOs. As a broad concept, it represents more than a simple increase in income level or access to material resources. Empowerment enables people to organise and influence change based on their access to knowledge, to political processes and to financial, social and natural resources. This chapter considers the nine domains of community empowerment suggested by Laverack (2006). These are improving participation, developing local leadership, increasing problem assessment capacities, critical awareness of people's needs, building of organisational structures, improving resource mobilisation, strengthening links to other organisations and people, creating equitable relationships with outside agents and increasing control over management. The evidence investigated NGOs' capacities for these components in the context of competing IK and GK approaches. The evidence found a number of areas of NGOs' capacities for community empowerment and gathered opinions through different sources, including field data.

Keywords Community empowerment \cdot Community participation \cdot Problem assessment \cdot Local leadership \cdot Resource mobilisation \cdot Social enterprise

7.1 Community Participation Versus Community Empowerment

Community participation in all its forms has become an increasingly important aspect of urban and rural policy in both north and south (Lyons et al. 2001, 1233). Lyons et al. (2001, 1233) find that participation has a significant effect on empowerment for sustainable development, which Friedmann (1996) calls 'socially sustainable conditions'. The discourse of participation and empowerment has become the common denominator of action for development agencies in a global world (Tembo 2004, 1025; Islam, Siti Hajar and Haris 2013) The literature suggests the terms 'participation' and 'empowerment' are used interchangeably (Rifkin 2003). Participation is now considered an important component for securing peoples' decision-making and equitable opportunities. It must not be seen as a short-term or casual involvement of people. It is a 'social experience shared by individuals and groups, who live in economic and social relations to each other in a society' (Malki 2006).

Participation is a prerequisite to collective action (Mondal 2000, 463) and an integral element of economic improvement and social change efforts (Bowen 2008, 65). The community workers should work on assessing community feelings through their active participation and then transform these into constructive community action plans. The participatory plan (PP) of the NGOs was most useful where the smiths (blacksmiths and goldsmiths) improved their confidence, traditional attitude, experience and skills. It was helpful to justify and verify their thinking within institutional arrangements. Both NGOs believed that more participation on such kinds of institutional arrangement decreased individual fears, apprehensions and limitations, on the one hand, and increased their 'social mobility' towards social empowerment, on the other.

The staff members of both NGOs said that they began this participation through their community survey, where they investigated all types of social networks and encouraged the active participation of local people. The staff members also said that this kind of participation transpired

extensively through a 'market chain' (as stated earlier in this chapter), where all types of social networks were established through 'collective actions' to achieve greater community empowerment. It was perceived as 'genuine' or 'real participation', which took place as now the populace had a voice in defining or diagnosing its problems. Therefore, development had a solid chance of meeting the real needs of the population. It was such participation which was started, or at least sanctioned, by the interested 'non-élite' population, such as blacksmiths and goldsmiths at the beginning of their decisions. It matured into a 'social force' exercising a critical mass of participating communities 'enabled to enter into spheres of decision or action' beyond their immediate problem-solving (Hamyaran 2006). In this sense, participation was seen as a 'stable, continuous and dynamic process', rather than 'infrequent, occasional and broken' within some contexts and not others to keep it in compliance with the preferences of the powerful actors. It gave smiths, especially the 'hitherto excluded', the opportunity to manage their own affairs, influence public decisions and participate in activities that affected their lives.

PAB arranged the exposure visit, where a blacksmith had a chance to talk directly with other blacksmiths, brokers, wholesalers and businessmen (both urban and rural), designers and suppliers. Many blacksmiths said that their exposure visits and participation in different blacksmith enterprises in Dhaka improved their confidence and working attitude. They were more interested and felt more confident in exploring Dhaka further on those occasions. This gave them more freedom and democratic rights as it was formed by the people and for the people. It was also essential for the repositioning of political power in favour of disadvantaged groups and for social and economic development. However, the staff members of PAB agreed that they were unsuccessful in some cases as they could not bring out the smiths from the exploitation of local élites, such as wholesalers. Such participation was helpful to minimise the exercise of power. The participation inevitably became simply 'a concrete manifestation of a broader political process of democratisation' (Dudley 1993, 160). In Bangladesh, NGOs created many examples of working with poor people, where such kinds of participation reached a practice of 'authentic development' accomplished by 'passive subjects'. This ultimately turned smiths into 'active subjects' that made their own dignity, social identity and social position into their livelihood struggle.

7.2 Use of Scientific Tools: Increase Problem Assessment Capacities

Under the 'bottom-up' participatory development, the improvement of problem assessment capacities among the poor people is now a crucial area for community empowerment. It is more important in developing countries such as Bangladesh, where a significant part of the population is illiterate, and they are not fully conscious about their problems. However, their opinions and decisions are important, regarding the sources and causes of problems, impacts and constructive solutions. The NGOs should have the capability to assess these in the local context.

The tools the NGOs use are important to assess local people's attitudes, culture, norms and values, and political situation. Both NGOs used a number of scientific tools, such as community surveys, participatory market chain analysis, market assessments, PP approach, rapid market surveys (RMS), participatory rural appraisal (PRA), training, technology fairs and also monitoring and supervision as part of their assessment methods. The NGOs developed these scientific tools through the 'trial-and-error' process. These tools were particularly useful to new entrepreneurs to increase their assessment capacity. The RMS was used to assess the present market situation. The smiths found this an easy and useful method, where they could record all necessary information. For example, the milk marketing (for milk processors) of MLP (PAB) included the following information: the name of the milk processor and address, daily receiving amount, form of milk (e.g. solid content), processed items, source/purchasing point, suppliers, milk price (Tk/lit), payment, requirement and possibilities of sale and extra requirement of milk. It was helpful to assess the total market situation, including current supply, price and extra requirement of quantities of milk.

PAB offered some other scientific tools to the smiths, such as learning log sheets, relationship matrixes, need assessment tools and the market actors' behaviour assessment tools. In this way, PAB could assess the most useful resources through identifying specific actions, contributed team success and actions for better in future. The relation matrix was useful to assess the possible outcomes in favour of producer groups, where all possible related parameters/indicators were identified. This matrix was able to locate the current and future baselines based on possible parameters. It shows how the milk producers and milkman relationship was developed in terms of information sharing, quality control, price

settlement and improvement of trust. It was also helpful in improving social networks and social trust among the different groups of the community.

Proshika developed a number of scientific techniques and measurements for proper performance assessment of the smiths, staff members and trainers. These included the methodologies to develop good trainers, strategies for speaking, indicators of good trainers, questionnaires of producers' personal characteristics, producers' evaluation form and indicators of training demands, principles of communication, learning strategies, techniques of memorising capacities, experienced-based learning circle, visual aids of training and principles of group learning. The SEED staff members followed the SWOT (strengths, weaknesses, opportunities and threats) strategies and PRA, which they used to provide statements about the 'needs' and 'problems' on the running projects. In order to do this, they had to complete the weekly monitoring schedule form and submit the loan monitoring report. They also had to make a report on programme performance and problem identification to the area coordinator. SEED used to gather experiences from their other projects, such as microcredit.

Through considering the appropriateness and evaluating the impact of these tools, I found that many of the smiths and the staff members could not own the global knowledge-based assessment tools. The main contradiction was that those tools were not developed through considering the local context; rather the NGOs used them directly, which they developed from the global development interventions. PAB used many tools, such as the market chain analysis, PRA, development frameworks (such as sustainable livelihood framework from DFID), which were to some extent help-ful to increase their problem assessment capacities. SEED had links with many donor agencies, such as DFID, ADB, World Bank and IMF, and some reputed research and educational institutions, such as the University of Bath and the University of East Anglia.

7.3 GROUP WORK PRACTICE: INCREASE LEADERSHIP EXERCISE AND QUALITY

Leadership is an important determinant for organisational performance. The key elements of leadership are vision, innovation, decisiveness and a strong people orientation (VanSant 2003). It is one of the keys to realistic assessment of problems and opportunities, establishment of priorities and

the marshalling of internal and external resources to address these priorities. In effective institutions, leadership does not reside only at the top; elements of it are evident at various levels of the organisation. The leaders, their selection, their training and skills, and the balance between their economic and traditional influence in the community are important considerations in this leadership pattern (Jones and Laverack 2003).

Both NGOs observed that most of the communities in Bangladesh suffer from leadership-related problems. First, the NGO staff members developed leadership for two reasons: one was that they were working with group-based enterprises, where a leader had a vital role to maintain and sustain these group activities; and secondly, the NGO workers wanted to share their organisational values and norms with a leader, who would be able to disseminate and share these values and practices with his/her group members. The staff members found that the reputable leaders are better than the appointed leaders. This kind of leadership can be compared with Gramsci's¹ 'organic' intellectuals, who are developed through their own social processes. This study found that these kinds of leaders were developed in a group through market channel and business plan, involved people with different existing associations and clubs, and made links with different social networks.

PAB developed local leadership through making partnerships and building forums, such as producer group forum, private service provider forum and business development service (BDS) provider forum. As a lead NGO, PAB provided financial, managerial and technological supports to the partner NGOs. For instance, GUP worked with the blacksmiths at Mostafapur Bazar under this partnership. The PAB staff members claimed that at the local level these sorts of forums played significant roles towards exchanging information and knowledge. PAB had 23 types of producer groups with 455 members with association of partner NGOs. On the other hand, as stated earlier, by December 2007, Proshika had established 220 area development centres (ADCs), 149,776 peoples' organisations and 18,417 group federations (Proshika 2008), which were maintained with proper organisational rules and regulations, principles and values. These were helpful for developing leadership and leadership qualities among the smiths. Proshika also formed business associations. These associations had a number of activities regarding local leadership. The staff members and the local community leaders confirmed that a good number of their leaders participated in the local administration elections, and many of them were elected as UP chairmen, ward members and ward commissioners.

Both NGO staff members claimed that they had leadership training, which they served throughout their programmes. Significantly, it was a fact that I did not find any kind of training manual on leadership training, nor did the community people agree with this opinion except business training. This finding was established by the office documents and observation methods. The staff members of GUP (partner NGO of PAB) stated that leaders could never be made through formal training, rather so it actually develops naturally in the society through practising and dealing with their group activities. Gramsci argues that the 'organic' intellectuals are genuinely popular, more conscious and much committed in a constant struggle to achieve both domination and hegemony (Morgan 1987). Some staff members of both NGOs agreed that they had no specific leadership training or training manual, but they always included leadership concepts in their business training.

Although I am in agreement with such observations, I found a contradiction between the opinions that the stakeholders provided during focus group discussions (FGDs) and case study sessions. The NGO staff members said, 'We have a wide range of leadership training', which was a false statement by these NGOs. The NGOs also needed to provide management training, which would be helpful for improving leadership as both groups of traders were involved in their business management. Both NGOs claimed that they provided business and management training to all of their service users. However, I found that only a very few of the smiths got such kind of training practically.

7.4 More Social Networks and Capital: Improve Resource Mobilisation

The ability of the community to mobilise resources, both from within (land, property, authority), and the ability to negotiate resources from beyond itself (training, finances, equipment)are important factors towards capacity building (Jones and Laverack 2003). One of the major functions of the NGOs was to improve resource² mobilisation. Network structure(s) can create enabling conditions for the mobilisation of social capital within communities and for people to feel empowered (Dale et al. 2008, 143). As stated earlier in this chapter, social networks helped many of the smiths by providing work opportunities, financial support, justice and social security (Purvez 2005).

Intense networks of interaction also probably widen the participants' sense of self; developing the 'I' into the 'we' or enhancing the participants' 'taste' for collective benefits (Putnam 1995). Zain and Ng (2006) argue that network relationships elicit and inspire firms' initial internationalisation intention, internationalisation pace and pattern; influence firms' market selection decision and entry mode decision; access to additional relationships and established channels and to local market knowledge. These help firms in lowering costs and minimising risk of internationalisation, and constrain firms' future scope and market opportunity. It was therefore seen that resource mobilisation was helpful to utilise many social, financial, natural, cultural and human capitals.

Both indigenous NGOs were working to develop local resources through coordination of social networks and integration of knowledge systems. The NGO staff members believed that due to the shortage of financial (e.g. donations) and physical capital (e.g. technology, technological tools) they needed to acquire capital from the donors, but they were also working to mobilise local human, social and natural capital. Both NGOs shared knowledge through a number of activities, such as exposure visits, training, and supervision and monitoring. It was seen that both NGOs used many local resources, such as local trainers (e.g. cascade training) and local raw materials.

Both NGOs had relations with other organisations as a part of their programme and experience-sharing scheme. The NGO staff members confirmed that they encouraged and helped their beneficiaries to link with other organisations. They invited the local people, administration and institutions to their different events, such as training periods, and community survey. These NGOs shared their opinions through discussions, meetings and workshops. They had also training exchange programmes with many organisations and encouraged smiths to attend the local fairs, where they involved the local administration and institutions. Through partnership programmes, PAB made excellent links with many local small NGOs, such as GUP.

The PAB file documents showed that the MSL worked with a number of organisations as a part of this partnership scheme between April 2006 and March 2007. These organisations were Bangladesh Agricultural University, Bangladesh Agricultural Research Institute, Wheat Research Centre, Rural Development Academy, Vocational Training Institute, Palli Karma Shahyak Foundation, CARITAS Bangladesh, Parbatya Boudha Mission Bangladesh and Art Institute of the University of Dhaka. They also participated in the 'local market development forum' and shared experience with information,

communication and technology (ICT)-based organisations. The SEED staff members claimed that they had good relations with many market- and business-level GOs and NGOs.

The MLP programme of PAB emphasised capacity building, partnership development and collaboration between GOs and NGOs. This programme provided a significant number of credits, inputs and amount technical assistance, which were mobilised and channelled to the project beneficiaries. For example, a number of partner NGOs had started an enterprise development loan with MLP, where Tk. 21,787,500.00 (£213602) was credited to 1011 small enterprises. Mondal (2000, 34) found that under the saving scheme of Proshika the development workers encouraged their target people to save a small amount of money from their everyday income to face their crises and further investment for more income. I found that about half of their income or profit was used on human development or placed in savings. According to a SEED office data sheet, their total investment was Tk. 177,756,941 (£17,42710), net profit Tk. 66,71,440 (£65,406) and self-sufficiency was 162 from July 2006 to June 2007.

7.5 Social Enterprise: NGOs' Holistic Approach

The concept of 'social enterprise' is a recent feature of NGOs' activities. Freer Spreckley first developed the term 'social enterprise' in 1981. The social enterprises are businesses with social objectives (Cheung 2006), whose surpluses are primarily reinvested to achieve social objectives or used in the community, rather than being driven to maximise profit for shareholders and owners. It is a means for non-profit agencies to maximise their mission-related performance through the development of new ventures or by reorganising activities to improve operational efficiency (Gray et al. 2003). The first Bangladeshi Nobel Laureate (in Peace) Professor Mohammad Yunus introduced this concept with his NGO Grameen Bank, which has now become a new dimension of social development in the NGO sector.

In NGOs' activities, the term includes privately owned ventures that have a very strong blended financial and socially responsible return on investment. It looks into how locally available traditional resources can be a viable policy option for societal transformation. It also encourages improving a continuous engagement from local communities, on the one hand, and secures sense of ownership to them, on the other. It makes stronger community development capabilities, maximising social impacts and improving vision for organisational sustainability. I think that there are many reasons why Bangladeshi NGOs should follow this social enterprise approach for their projects, such as with blacksmiths and goldsmiths. The criticisms of the NGOs' microcredit programmes, such as commercial attitude and poor people's exploitation, furnished the issue as a hot development agenda. It is now proved that the microcredit is not a kind of programme to achieve a comprehensive development strategy for safe sustainable human development as it targets a particular area of development.

It was observed that it secured neither individual development nor that of the community as whole, where social enterprise can be effective to achieve broader development, as it attempts to secure interactions with many social networks, on the one hand, and its impact is more holistic and comprehensive, on the other. This strategy empowers the local people to generate their own capital in a sustainable manner. This approach can work significantly, where many local NGOs are facing funding crises and where NGO–state relationships are often fraught with tension. The representatives of state institutions may see NGOs as a threat and are often intent on keeping a tight eye on their activities, especially those perceived as being potentially profitable (Phillips 2005). This approach can be used as an alternative mode for empowering citizens. Cheung (2006) finds this an innovative NGO approach (in meeting new needs), which creates huge job facilities, enhances integration into labour markets, becomes selfreliant and increases the density of community network and cross-sector partnerships.

The positive side of NGOs' activities as social enterprise is that this works through an effective partnership. The aforementioned description illustrates that it is one kind of joint venture where the NGO is a partner that works with the government and other business sectors. One of the merits of this partnership is to take social responsibility and accountability from the multiple agents as it does not work with one wing. The best value of this approach is that it shares all earning benefits with the smiths, where they own this enterprise. It improves their dedication, sacrifice, care and attention to the enterprise. I mentioned earlier in this chapter that SEED established a partnership with the business sectors, social enterprise agencies and government. The goal of SEED was to promote sustainable income growth amongst the micro and small entrepreneurs, and create employment opportunities for the poor. Until June 2006, the SEED disbursed Tk. 725.86 million (£6.60 million) among 5,640 entrepreneurs. Up to December 2006, a total of 13,441 entrepreneurs (40 % female) were organised at 76 Proshika ADCs, and Tk. 742 million (£6.74 million) loans were disbursed among them. A total of 43,095 new jobs (54% full time and 46% casual) were created (Proshika 2008). Both formal and non-formal training was given to around 2,000 entrepreneurs. The business counselling, technological and marketing extension supports were provided to them. It included some other supportive activities, including social scheme, relief and rehabilitation. This social scheme provided a number of facilities, such as scholarships, further loans, medical and job facilities, for SEED members. This was more comprehensive, holistic and accountable for a sustainable community. One SEED staff member compared its social accountability with the bank sector in Bangladesh:

Banks never go to the community; banks do not provide business plan, training, motivational and counselling supports to the community; they don't have social schemes; rather SEED goes to the community and it builds up social networks with different sectors.

As a socially responsible organisation, Proshika widened its experience and skills base by carrying out relief and rehabilitation activities during the floods of 1984, 1987, 1988, 1995, 1996, 1998, 2002, 2004 and 2005; the cyclones of 1985, 1991 and 1997; the tornadoes of 1989, 1996, 2005 and 2006; and the $monga^3$ of 2005. The floods of 1988 and 1998 and the cyclone of 1991 were colossal and imposed unprecedented damage. During the 1988 flood, Proshika carried out an extensive relief and rehabilitation programme at a cost of Tk. 71.90 million (£0.65 million). The relief and rehabilitation programme carried out during the 1998 flood was even more extensive, involving a cost of Tk. 192.30 million (£1.75 million) (Proshika 2008). The activities conducted in response to these disasters were food and baby food distribution, running health camps, repairing tube wells and latrines, vaccinating cattle and poultry and providing fodder for them, loan disbursement, seed distribution, etc. With such long and successful experience of carrying out disaster rehabilitation, Proshika played a significant role in mitigating the sufferings of the poor people affected by the 2007 cyclone and restoring their livelihood opportunities.

It was seen that both NGOs' enterprise schemes offered the prospect of a greater equity of economic power and a more sustainable society by combining market efficiency with social and environmental justice. These were social mission driven organisations, which dealt with goods or services for a social purpose. The necessity to deliver on financial, social and environmental performance targets was often referred to as having a 'triple bottom line'. Here, the aim was to generate profit to further their social and environmental goals, which helped to consider NGOs as 'social firm' encompassing trusts, cooperatives, associations and institutional belonging. These social objectives helped to tie together the power of local communities and fabricated a wide range of goods and services. The benefits of social enterprise included enhanced livelihoods and employability for the marginalised groups of blacksmiths and goldsmiths; production of reasonable cost, good quality, socially useful and environmentally sustainable, mobilisation of public and private resources to advance social justice and economic opportunity (Jackson 2005). Those brought a change towards a significant new dimension of non-profit performance.

7.6 Communication and Knowledge Sharing: Strengthen Links with Outside Agents and Improve Critical Awareness About Needs

Empowerment has been seen as the transfer of control over decisions and resources to communities or organisations. It is also seen to lie in improving a community's negotiation with external agents. It does not work as a facilitator of rights in general, but as a realisation of rights to enable greater control over livelihood resources (Lyons et al. 2001, 1234). The evidence shows that the NGOs used mainly four types of communications channels for improvement of critical awareness about the community needs. These were aid channel and knowledge transfer, partnerships and links with like-minded GOs–NGOs, PP and media, including website connection.

Both NGO staff members explained that, due to the shortage of financial capital and lack of government help, they had to depend on external agents, such as donors, as an 'aid channel'. But they confirmed that this 'aid channel' was helpful to build-up their capacity, particularly the management skills, new technology and innovation, and development policy between the north and south. The NGO staff members said that the donors had diverse experience working with different countries and with different communities. These networks were frequently international in reach for the purpose of fostering social change (Bebbington 2004). They were called 'transactional networks' (Radcliff 2001), where both NGOs were embedded. However,

this study found that the NGOs faced two types of contradictions. One is 'how such transferred global knowledge able to consider the local issues in Bangladesh?' The second is 'how the NGOs overcome the problem as it is difficult to talk glibly of northern and southern NGOs as discrete entities?

The evidence shows that the outcome was a notion of empowerment, which focused on the intellectual and social assets of communities or organisations. Though the NGOs stated that their relationships with outside agents including donors were horizontal, it was in fact practically vertical. Many authors consider this as 'patron–client relations' (PCR).⁴ Because of power relations, the donors and NGOs' relationships were as 'providers-recipients' under the certain terms and conditions. Through searching financial sources, I found that this PCR was less in Proshika than PAB; this is because Proshika already increased a substantial amount of funding through their microcredit and savings schemes. About 70 % of their fund now comes from its own budget. But PAB's total fund comes from central office and they have no microcredit and saving schemes.

It is necessary for the NGO to have precise knowledge and understanding of local people and their problems. This approach challenged the conventional development approach in which development strategies were imposed from the top. Laverack (2006) argued that rather than using a formal education approach the practitioners should help the people by working with them in small groups, analysing why some people are facing bad conditions and others not, why some people are living in poor conditions and others not, and what local, state and national actions might remedy their particular circumstances. This kind of notion helped them to increase their level of critical awareness. This was called 'critical thinking' or 'critical consciousness' (Jones and Laverack 2003).

PAB has partnership programmes through shared responsibility to effect change, mobilise resources and exchange experiences (Jones and Laverack 2003) with local NGOs, such as GUP. The MSL programme worked with 70 rural NGOs and 200 private 'like-minded' service providers together to ensure 12,646 rural smiths' (e.g. smallholder farmers, non-farm enterprises and rural potters) access to skills, credit, information and technology, and market linkage until 2006. More than 20 government offices in project areas supported the smiths by either improving their own training curriculum or providing training, credit and other material support (Chaudhury 2006). The NGO staff members claimed that their collaboration with other NGOs, GOs and smiths was established successfully.

Both NGO staff members claimed that they followed the PP approach through PRA, FGD, dialogue, campaign and rally, and they successfully shared knowledge with the community people. The NGOs also knew how they would develop their work according to the needs of the smiths. This plan included local culture and practice, peoples' choice, needs and demands, and norms and values. It was a 'learning process', where an organisation learned and found the suitable interventions. Both NGOs took initiatives to find out the real 'mentors' in the local community from whom the smiths learned about their products. The NGOs also had continuous interaction with the smiths through monitoring and supervision via mobile phone. In an evaluation of the MSL project, it was found that the programme had been able to change the real-life story of many rural smiths by facilitating such technology. I found that this kind of technology was used in market assessment Chaudhury (2006, 64) argued that the MSL programme of PAB for poverty alleviation through the adoption and implementation of the BDS approach was successful. The staff members described this as 'knowledge flow', which was documented and shared with other local communities across the country.

Both NGOs had individual websites, which were accessible to the beneficiary groups. PAB reported that during the reporting period of May 2006, 16 articles were published on different technologies of MSL programme in the most popular national daily newspaper named *The Prothom Alo* which were read by 100,000 readers. The Bangladesh Radio broadcasted ten programmes on the small businesses. By assessing letters, phone calls and discussion, the radio authority decided that at least 20,000 rural listeners benefited from this broadcasting. The private and national TV media provided coverage on the rural technology exhibitions, product fairs and training activities on several occasions. On the other hand, Proshika made over 245 audios and 819 participatory videos produced on democracy, human rights, gender equality, good governance and awareness about environmental protection.

7.7 Decentralised and Autonomous Management

At present, 'empowering organisational structures' in terms of management sufficiency, control over management and 'ensuring mutual support groups' (Laverack 2006) are important elements of NGOs' work. Many authors state that, in an increasing global world, whether a local NGO is using financial resources from its own funds is an indication of its move towards 'self-reliance' and relieving donor dependency. From this point of view, I found that both NGOs had donor dependency for their funding, though I stated earlier that Proshika already created a substantial amount of its own funding.

Both NGOs had independent advisory and management boards, and they were adequate to control through autonomous and decentralised management. The MLP of PAB was managed by the head office located at Dhaka, linked with their main office in the UK. The head office of SEED of Proshika was located in Dhaka. According to the opinions of area office, they had full autonomy to work, and their central office did not impose any administrative decisions on them. The allocation of funds was through the head office. The head office took all working decisions according to the needs and demands of the local communities. Both NGOs had more organic relationships at community level; they seemed the most appropriate service delivery partners and were invited more and more by multilateral organisations to implement these programmes. However, I suggest that their organisational structures did not fit the new large-scale requirements; their accountability systems were not matching either and they might end up in difficult political situations with both governments and the populations they serve.

I reviewed the organisations' capacity development, followed by a model provided by Horton et al. (2003). They show how an organisation can achieve its organisational capacity through a number of internal and external resources gradually. These include monitoring the external environment, reviewing the organisation's strategies, identifying capacity needs and plans, negotiating external supports, implementing and managing the capacity development processes, and monitoring and evaluating the process. Jones and Laverack (2003) mention that an empowered organisational structure is able to establish coordination mechanisms in each village to allow the representation of different social groups and leaders from existing village institutions and organisations. It is envisaged that this would facilitate the project to develop a more consequential relationship with people in the villages and facilitate the project to organise project resources more effectively. This would entail a wider range of people in decision-making exercises. In the light of this, I found that the NGOs were partially successful in improving the first two, but failed with the third. They both developed some subsectors.

For example, the subsector of PAB arranged 16 group meetings with two partner NGOs in Dinajpur District in February to April in 2007, where a number marketing issues such as the vaccination for profitable cow rearing, the necessary care for pregnant and milking cows and the experience of the quality milk production were shared among group members. This meeting was important for improving group functioning, but the question was about the improvement of decision-making. The group did not have the autonomy to decide what sort of help was needed, but received the interventions as the MLP decided and provided. So, the study found that the PCR still existed. On the other hand, SEED also had links with other mutual peer institutions, such as the Bangladesh Women Enterprise, ASA and BRAC, and their attempts for community empowerment were comparatively successful. As Fatmi and Islam (2001, 253) argued, Proshika's model for social development and poverty alleviation emphasises total empowerment. Proshika therefore followed an approach of 'walking on two legs'.

7.8 Conclusions

To sum up, although the NGOs showed potential as agents for community empowerment in Bangladesh, their actual contribution in developing local leadership, creating empowering organisational structures and increasing people's control over management was weak. I found that both NGOs followed the so-called PP and tried to consider the community feelings through their active participation, and then integrated these in community planning. But, practically, they faced contradiction as they could not include many marginal and ultra-poor within their PP. The PP could not achieve 'real' or 'genuine' participation as a substantial number of bureaucratic and top-down methods existed. The NGOs used a number of scientific tools, such as the community survey, participatory market chain analysis, market assessment, RMS, PRA, training, technology fair and monitoring and supervision, as part of their assessment. Nevertheless, many of the local producers and NGO staff members claimed that they could not persist with such interventions as they could not own them effectively. I found that a good number of staff members believed that in many cases they failed to fit with the local context and with local producers' needs and feelings.

This chapter argued that the reputational leaders/the 'organic intellectuals', who informally serve the community, had a better chance of building community capacity. It was found that both NGOs believed that leadership is not a process which could be developed by training; rather it should be developed through usual group activities, such as a PP, market

channel and business plan, involving people with different existing associations and linking with different social networks. But this initiative was blocked by some of the local socio-economic and political processes, such as cultural lag, irrational attitudes, political influence, lack of GOs' help and non-cooperation of the local power structure. On the other hand, NGOs' communication and knowledge-sharing practices strengthened linking capacity with outside agents and improved critical awareness among the local producers about their needs. It was found that both NGOs used aid channel, knowledge transfer, partnership with like-minded GOs-NGOs, PP and media, including website connection, under these interventions. The chapter showed that decentralisation and autonomous management systems were significant aspects of community empowerment. I found some contradictions, such as self-dependency versus donor dependency and centralised versus decentralised management systems. It argued that without decentralised and autonomous management systems the NGOs could not take any kind of decision independently, which would be in support of community empowerment. NGOs' social enterprise concept was proved an effective intervention for community empowerment. It was considered to be a holistic approach, which could open many windows to coordinate and accommodate many social networks through partnerships. The chapter illustrated that NGOs' recent trend towards social enterprise was convincing, but the NGOs faced some contradictions, which included the sociocultural barriers (e.g. self-interest attitude, cultural lag and rigidity of traditional practice, financial insecurity) to such new social technology.

Notes

- Antonio Gramsci (1891–1937) was the leading Marxist theorist produced by Western Europe since Marx and Engels themselves. By 1917, Gramsci was an influential member of the Italian Socialist Party, and in 1921 he became a leader of the new Italian Communist Party (Morgan 1987, 301).
- 2. This is something that can be used for support or help and is a humancentred concept. Resources refer to things, such as money, property, assets, wealth, capital or goods, which have economic and social value. In order for something to be considered a resource therefore, it must be perceived to have some attributed value.
- 3. *Monga* is seasonal food insecurity in ecologically vulnerable and economically weak parts of northwestern Bangladesh, primarily caused by an

employment and income deficit before *aman* (one kind of paddy) is harvested. It mainly affects those rural poor, who have an undiversified income that is directly or indirectly based on agriculture (Zug 2006, 2).

4. PCR is a mutually obligatory arrangement between an individual who has authority, social status, wealth or some other personal resource (the patron) and another person who benefits from his or her support or influence (the client) (Anthromorphemics, n.d.).
NGOs' Limitations and Challenges for Social Capital and Community Empowerment: Local Dynamics and Global Pressures

Abstract This chapter looks at the major limitations and challenges of NGOs' capacities for social capital and community empowerment approaches. This chapter highlights the evidence of the two NGOs-Practical Action Bangladesh and Proshika-that we discussed in Chaps. 6 and 7. A growing number of critical assessments suggest that the operational impact of NGOs in development activities was less than claimed. NGOs may pursue a 'service delivery' paradigm, where the provision of services is strictly separated from engagement in the broader polity. In reference to this connection, NGOs become providers of goods to the poor 'consumers' rather than the 'facilitators' of collective action and empowerment (Rahman 2006, 452). This growing predominance of service delivery programmes among NGOs is problematic for the goals of empowerment and poverty alleviation. Based on the evidence and literature, this chapter shows that the NGOs could not archive their development targets because of the local dynamics and global pressures. There are a number of local dynamisms such as local context, political issues, NGOs' conventional role, NGOs' monolithic development approach and lack of accountability. A number of global pressures were noted such as the lack of local funding and donor dependency, global development frameworks and decentralised management system.

Keywords Local dynamics · Global pressures · Financial vulnerability · accountability · Centralised management · Donor dependency

The evidence shows a complex notion in highlighting these approaches individually rather than as combined; it was because the same kind of limitation was accountable to different elements of both approaches. The summary of the findings from both NGOs is presented in Table 8.1. It is envisaged that there were some common limitations which were seen in both approaches. The NGOs individually found a number of distinct limitations. It was also observed that some limitations were aggravated because of NGOs' weaknesses (e.g. a monolithic approach and the exclusion of ordinary people, unhealthy empowerment and malpractices of power, and management problem). Some limitations were related to the country's poor socio-economic and cultural conditions (e.g. lack of social trust, role local power structures and negative role of religious leaders), and the rest were created by the government and donor agencies (e.g. continuing debates

NGOs	Barriers for social capital	Barriers for community empowerment
Both NGOs (common barriers)	Illiteracy, social ignorance, financial crisis of the smiths, lack of confidence to the new ideas, marginality and isolation, class conflicts, leadership problems, power relations, business secrecy, 'knowledge poverty', shortages of financial and other capital, lack of social trust, non-cooperation of local leaders and GOs–NGOs, negative role of brokers, lack of control over the market, exclusion of ultra-poor, reluctance to work with GK, lack of sharing attitude, negative donors role, NGOs' commercial attitude and short period of project, overlapping of projects, <i>chatabaji</i> , partnership problem, natural disaster and management problem	Lack of social education, social ignorance, traditional habits, financial crisis of the smiths, lack of unity and psychological limitations, reluctance to use GK, lack of knowledge ownership, role of brokers and other exploiting groups, non-cooperation of local leaders and GOs–NGOs, lack of government initiatives, inaccessible local markets, shortage of financial and other capitals, lack of social trust, class conflicts, power relations, <i>chatabaji</i> , lack of decentralised management, external intervenors and donor dependency, lack of update knowledge, unavailability of local resources, shortages of training resources and high market competition

(continued)

NGOs	Barriers for social capital	Barriers for community empowerment
MLP (PAB)	Demoralisation and problems of knowledge transfer	Demoralisation, lack of communication, problem of knowledge transfer and project discontinuation
SEED (Proshika)	Political involvement, political pressure, non-cooperation from the community, lack of dedication and low qualification of the staff members	Lack of dedication and low qualification of the staff members, bad relations with the government, low staff salary, high staff turnover, malfunctioning of associations and negative role of religious leaders

Table 8.1 (continued)

Source: Author based on field data.

between IK and GK, donor dependency, inaccessible market and lack of government initiative, and challenges of knowledge transfer).

8.1 LOCAL PEOPLES' POOR SOCIO-ECONOMIC CONDITIONS

In Bangladesh, the general observation is that the rich people have better access to loans, savings and credit than the poor have. Furthermore, even when financial services were extended, the poor might have difficulty taking advantage of them due to feelings of powerlessness because of socio-economic inequalities. There were two sets of issues regarding sustainability. Other obstacles to sustainability were location or remoteness and social constraints, for example, 'illiteracy and female seclusion' (Johnson and Rogaly 1997). I found that those in remote areas had a lower survival and sustainability rate than those in urban areas (Vargas 2001, 18). The evidence shows that due to disadvantaged socio-economic conditions the gap regarding expectations between smiths and NGOs was high. They did not get even a single local person's participation in some of their workshops and meetings. One blacksmith expressed his opinion:

We are labourers, we are illiterate, and so our opinions are valueless. We only know we need Tk.200.00 (\pounds 1.80) daily to live. We are all satisfied with those (NGOs) who will help us to get that.

Some PAB staff members said that many smiths, who were hand to mouth, asked what kind of financial help (such as loans) the NGOs would provide when there was any dialogue with them. They found that many of them left the room rapidly when they knew that they would offer training facilities. The NGO staff members observed that the smiths suffered from 'knowledge poverty'. The staff members stated that they could not take many scientific initiatives due to local peoples' lack of awareness, conservative attitude and social ignorance. One goldsmith stated:

NGOs' loan is not a prestigious matter. It has been recognised that the people who take loans from Proshika are considered as poor.

As a result, the NGOs could not consider many development messages and advance knowledge in their development services. Most smiths could not follow the global knowledge-based guidelines and training instructions. They were also slow in understanding and processing information (Kusumahadi 2002, 4). They were rigid and afraid to share their own local knowledge and experience as they thought they might lose their customers and markets if they did so. One goldsmith stated:

Design is the most valuable part of a goldsmith's work, but nobody wants to share this design as it is their personal and business advantage.

The smiths' financial vulnerability was one of the prime reasons to maintain robust links both within and outside the village (Cahn 2008). Many smiths reported that they were afraid that they would be unable to face the challenges of global competition. The NGOs observed that none of the smiths could take the risks of GK because of high cost, safety and security. It is because of their financial inability, on the one hand, and lack of scientific knowledge, on the other. The goldsmiths felt such kinds of difficulty as they were expecting to introduce a new and scientific carat measuring system to be introduced to determine quality and price. One gold businessman said:

Initially the businessman will lose money while adjusting to this new system. Many workers will lose their job. Jobs will be more challenging and competitive. It is true that lack of social education had limited the space, where they were located. One SEED staff member said:

The goldsmiths do not need to use high technologies as they are not educated to do so. Those take more cost and risks. They are not aware of the benefits of those and many do not feel comfortable accepting those as they find more competition there.

Both NGO staff members said that much of their innovative thinking was not being implemented because of such non-cooperation of the smiths. The staff members found this problem to be a 'vicious circle' and, at times, it took a long time to overcome this barrier.

8.2 NGOs' Conventional Role: Questions of Accountability

The NGOs' conventional role towards smiths in terms of development has been extensively debated. Bebbington (2005, 937) noted that NGOs' interventions became biased towards the less poor. It was observed that the NGOs were concentrated in urban areas, where replication of their activities was frequent, but many of the bottom people were excluded from NGOs' target. NGOs' bureaucratic approach could not ensure a 'sustainable community'; rather, they further marginalised many local people, such as blacksmiths and goldsmiths. Significantly, the exclusion of many poor people in programme objectives is quite a common feature of NGOs' activities in Bangladesh. The realities of those who were poor and marginalised are often ignored or misread (Chambers 2004). The notion of social exclusion focused on inadequate social participation, lack of social integration and lack of power (Hunter 2004).

Notably, microcredit-based NGOs such as Proshika did not provide loan support to those goldsmiths, who were not guaranteed to repay such loans. In addition to this, the replication of a programme in a community had a strong negative effect on the flow of programmes of the same sector or run by the same NGO (Kusumahadi 2002). As a result, many of the extreme poor remained beyond their services. Recent research has also criticised NGOs for lacking the capacity to involve the 'ultra-poor', and the poorest villages and neediest communities. NGOs' rigid practices kept away such vulnerable communities (Johnson and Rogaly 1997; Buckland 2004, 135, 139). Proshika's social and economic empowerment process usually sidelines the vulnerable population. One gold businessman said:

SEED selects the people to give loan support who are able to return their loan safely. I cannot find any difference between the normal bank and NGO bank. Both are same. But I think they should provide loan support to them who actually need it.

PAB worked with people's existing employment rather than focusing support on new employment. Some NGO staff members, community leaders and smiths raised questions about NGOs' short-term development approach. This approach failed to sufficiently mobilise indigenous social and political capital that would build, or rebuild, community capacity and ensure sustainability of impact (Buckland 1998).

The NGOs also observed that the indigenous people, such as blacksmiths and goldsmiths, were suffering 'minority' and 'isolation' problems, and their trust of NGOs was marginal. The smiths thought that, due to the expansion of large- and medium-sized industries through globalisation, they might disappear in the near future. There was a clear lack of confidence in their reluctance to face this global competition as their education and quality of production were not sufficient. They were also afraid of the future as they were not getting any kind of GOs' and NGOs' supports, and national and international networks towards their working capital, and modern tools and technologies. The national and international involving technology-based firms were much stronger than the local ones (Hendry et al. 2000).

The finding concludes that, in the 'winners and losers' game of globalisation, more and more people were excluded from the benefits of this so-called progress of development. Only those who could sell their labour or services as a commodity in an increasingly globalised economic system could survive; the majority who could not were left out. The local languages, oral histories and cultural traditions were also threatened. Equity and redistribution were increasingly recognised as the 'missing link'. Devine (2003) found that the 'ultra' poor had been completely marginalised in the NGO budgets in Bangladesh. It was not simply a lack of available resources; it was also fundamentally an issue of unequal power relations in which the poor were permanently marginalised and vulnerable, dependent as they were on local élites (Wood 2003).

The policy documents of both NGOs were well written, and the NGOs had high-flying goals and objectives, and set out interesting programmes. The documents described how these would be achieved within the time framework. Notably, I gathered very good impressions from their meetings with the head office staff members in FGD sessions, who explained how they were helping the smiths. At this point, I would like to emphasise a couple of interesting arguments: the first is about NGOs' written policy, which was just a policy by which they caught the attention of the donors. However, there was a big gap between their written policy and real implementation at the field level. One Proshika staff member stated:

There is a gap between policy and implementation because of financial dependency. Most of the NGOs are project oriented. When final funding comes it is too small for the purposes, then the NGOs change their target and the donors again shorten their funding allocation during final allocation. Then the NGOs again shorten their project target.

Many smiths and community leaders raised questions about the ultimate controversial objectives of NGOs. A number of smiths and community leaders argued that the NGOs were the 'new money laundering' agencies. They stated that the NGOs were using donations to 'make money for themselves'. Many said that they were confused, whether or not the NGOs used more than 10 % of their resources at the field level, but they had expensive furniture, high salaries, high consultancy fees, etc. Edwards (1999, 367) found the same finding when he reviewed the cost-effectiveness of two NGOs in Bangladesh because of high overheads (a large number of staff and buildings). In this regard, Stubbs (2006, 13) argues that the most active 'high-profile NGOs' are more concerned with personal gain rather than with the wider cause of poverty elimination. In addition, some NGO staff members' attitudes were negative about their own value. They could not believe that their help would contribute a great deal of benefit for the smiths. A field worker of PAB gave this opinion. Smiths also claimed that the staff members came to them just to take their kisti (instalment) and they gave them nothing, which eventually benefited them.

NGOs' development approach was 'monolithic'. The Markets and Sustainable Livelihoods (MSL) of PAB just provided training and advocacy, and SEED of Proshika provided the microenterprise loans, training and advocacy. I argue that a credit does not necessarily help the poor to accumulate assets, improve productivity, escape poverty and improve empowerment (Cosgrove 2002, 62; Nawaz 2004, 170). It is true that microfinance programmes and institutions are increasingly important in development strategies, but knowledge about their impact is partial and contested (Hulme 2000) and promotes lively debate. There were some other concerns within microfinances, such as the distribution and usage. Nawaz (2004) conducted a study in Comilla district in Bangladesh on the microfinance clients of three NGOs-Grameen Bank, ASA and BRACand they found that microcredit did not reach two-thirds of the households, and in particular did not reach the bottom layer of extremely poor households. In addition to this, a significant amount of the microcredit was provided to non-poor and non-targeted households. Importantly, I observed that the microfinance does not 'automatically' empower, just as with other interventions, such as formal education and political quotas that seek to bring about a radical structural transformation that true empowerment entails (Kabeer 2005). This approach can foster sustainable development if, and only if, it is integrated in a view of development that links the social, economic and environmental dimensions (Vargas 2001).

The accountability of NGOs, particularly their 'downward accountability'¹ to their beneficiaries, affected NGOs' effectiveness in the process of empowerment for the poor and marginalised people in developing countries (Kilby 2006) such as Bangladesh. This debate is well travelled, and much less concern is given to the NGO's broad values and its effect. There was a considerable gap between 'the talk and the walk' (Keystone 2006). This notion of accountability created practical knowledge gaps. While, in principle, the donors generally assigned a high value to it, in practice the ways in which they managed their grants and investments did not support it. Here, some areas such as common planning tools, reporting formats and information systems did not capture the quality of accountability in relationships between NGOs and their constituents, nor did they actively enable learning and improvement (Keystone 2006). It was seen in the study that the microcredit demanded 'a type and quality of relationship that actually limited poor people's room to manoeuvre'. Moreover, NGOs were replacing traditional informal credit sources (Davis 2006, 12-13).

Informal discussions with the smiths, who were neither the members of SEED nor took a loan from any NGO, suggested that enterprises required larger amounts to be loaned, which the local NGOs found risky, and many of the local NGOs did not have the capacity to provide such enterprise

development loans to the small enterprises. The government and private commercial banks also suffered from bureaucratic practices and required collateral, which the small entrepreneurs in most cases were unable to provide (Chaudhury 2006). In reviewing the conditions of a SEED loan, I found that the SEED provided loans on specific terms and conditions, which the goldsmiths found hard, complex and unaffordable. They calculated that, including all, the interest rate became up to 38%, which a smith had to pay from his first month of the instalment. It was also found from the list provided by SEED (2006) that the flat rate of other NGOs (such as BRAC 15 % and Micro Industries Development Assistance and Services (MIDAS) 16 %) and other government and private banks (Janata Bank 12.5 %, Agrani Bank 14 % and Prime Bank Ltd 15 %) was comparatively low.

The goldsmiths had serious problems formulating loan procedures; for example, to complete nine pages of an application form with information on personal, business, property, business appraisal and profit. SEED required ten types of documents, such as the original property or tenant agreement, two copies of a passport size photograph, letter (with two witnesses) and one copy of a passport size photograph from guarantor, citizenship certificate or two pages photocopy of passport, bank certificate mentioning current account of the business, photocopy of last month's electricity bill, photocopy of last month's rent payment receipt, photocopy of trade licence and two revenue stamps of Tk. 4 and four non-judicial stamps of Tk. 150 (£1.36) each (two by the name of shop, one by the name of producer and one guarantor). In many cases, the goldsmiths had difficulties producing such kind of documents.

8.3 Social Capital Is Meaningless Without the Support of Other Capitals

Social capital is meaningless without the support of other capitals (Islam and Morgan 2012a; Islam 2015a). Social capital alone cannot build the social economy and develop communities; it has to be used in conjunction with other forms of capital, such as financial, human, environmental and cultural capital (Kay 2005, 168). The recent literature has also shown that possessing physical and human capital is not in itself adequate to ensure project success (Nel et al. 2001). Even local knowledge alone was not enough anywhere, but perhaps particularly in rural areas (Malecki 1998).

Both NGOs had limitations of resources as they could not provide additional support for local people, dependent upon their own assets, skills and enterprise. This study found that the smiths had little capital, very limited access to credit, very little power in the market and rarely received support from the formal institutions. Proshika's social programme, a small part of the budget, was the 'glue' that held community groups and the whole programme together (Yildiz et al. 2003). But recently the cuts in the social programme of this NGO had been too deep. Compared to the first three years of phase VI, the social programmes had declined by 90 % for the training and health infrastructure components, and by 40 % for universal education (Table 8.2).

I found that the smiths were keen to participate in those projects, where their immediate financial benefits were obvious. The overall picture was that the smiths' satisfaction level was comparatively poor

Social components	Achievementfirst 9 months compared to Annual Work Plan (2002–2003)	Achievement compared to phase VI targets (1999–2004)
Organisational bui	lding	
Area expansion	Low: 27%	High: 139% for village/slum expansion
Primary groups	Low: 40%	High: 86%
Federations	Very low: 6–28%	Medium: 66–74%
Training and popu	lar theatre	
Human development	Low: 25–40%	High: 53-144%
Practical skills	Low: 13–37%	Very low to very high: 22–128%
Cultural programmes <i>Health</i>	Very low: 7–27%	High: 81–99%
Health training	Medium to very high: 54% to >500%	High: 73% to >300%
Health infrastructure Universal education	Very low to medium: 13–51%	Low to high: 15–100%
NFP school and learners	New schools: 0%	High: 83%

Table 8.2 Social component and achievement of Proshika

(continued)

Social components	Achievementfirst 9 months compared to Annual Work Plan (2002–2003)	Achievement compared to phase VI targets (1999–2004)
Adult literacy centres and learners	New ALCs: 8%	Medium: 56–61%
Post-literacy centres	New PLCs: 0%	Medium: 49% (first four years)

Table 8.2 (continued)

Source: Adapted by author from Yildiz et al. (2003).

with the long-term social development interventions. Allegedly, the NGOs neither gave advice nor helped to link with other institutions nor were they cooperative in linking with other people, such as business leaders, moneylenders and local administrators. In addition, some smiths argued that they had problems when they said that they were the client of a particular NGO. This situation was worse for the smiths of SEED as they felt that the NGOs should link with the government bank, where they had no access for loan facilities, though they were paying all kinds of taxes to the government. The community leaders also added that the role of NGOs should be that of the 'negotiator'. They also believed that their low amount of microenterprise loan would not be very helpful for their overall development without providing other supports.

8.4 Social Networks and Empowerment (Politically) as Negative Features for Development

The theoretical notion of social network and empowerment is to produce positive outcomes for development. For example, Lin (2000, 787) argued that the poorer social networks mean poor social capital and resources. Purvez (2005) found in his study that the economically poor people in Bangladesh had poor social networks. I agree with these findings and add that, recently, social networks have been a negative feature in some cases in Bangladesh. This observation needs to be located within the country's distinct sociocultural and political conditions (Islam 2015a). Depending

on various causes, social networks might allow the generation of positive or negative social capital. This also depends on the level of power decentralisation in a political system within new institutional and social capital approaches (Svendsen and Svendsen 2004, 173, 175). It is still observed that in most developing nations the power of NGOs has increased in recent years in relation to the state or government.

In the case of Bangladesh, the NGOs have become a formidable force, affecting the political and economic domains, especially the power and legitimacy of the government (Haque 2002, 411). Bendana (2006) calls this the 'NGO-isaton' of movements and politics. In addition, there is a growing trend, in alliance with foreign donors, for the influence of NGOs to expand to the extent that they now compete with the government for scarce foreign assistance at the national level and for sharing political power at the local level. To enter and be involved with the political networks as a result of empowerment of NGOs was not always helpful; rather, these were used for self-interest, where the community people were affected seriously. Here empowerment was seen as 'any process by which people's control (collective or individual) over their lives is increased'. So empowerment was not a neutral force; rather, it was a 'discursive construct, with its assignment meaning resulting from the exercise of power' (Lyons et al. 2001, 1234). The head of the MSL argued that this politically empowered network deformed new feudalism. He stated that:

[The] new power structure is now polarised in and affiliated by party based political networks in the community. This party based politics are spread out so strongly from grassroots to international levels. All parties have organisations everywhere. They are holding power. There is no power at the hands of educated and conscious neutral people such as teachers, social workers, and young generations.

I found that many local NGOs' executive committee members had political involvement, which ranged from the UP chairman to the ministry level. The local NGOs are now part of the local power structure (Hossain 2006). Proshika was accused of embezzling funds for developing NGOs during the BAL regime from 1996 to mid-2001. It was alleged that the Proshika chairperson (who was also the head of the Association of Development Agencies in Bangladesh (ADAB))² maintained a direct connection with the party in power and tried to

influence the ADAB in favour of his political will (Hossain 2006). Hossain (2006) found that, out of 30 local NGOs in the Rajbari District in Bangladesh, only 2 NGOs had been exploiting opportunities through their networks and membership with the ADAB, which had good connections to national NGOs and NGO leaders at the centre. Even the membership of local NGOs depends on how deeply the personal relationship and political line were being maintained. NGOs' politics appeared factionalised along political lines, and conflicts were confirmed to power, leadership and resources. The focus on power and trust naturally brings up the issue of inequality: Can trust exist among those who are unequal in power? (Cook 2005, 7).

It is also argued that the private power becomes public power when it controls voting behaviour (Islam 1974, 5). Many smiths and community leaders said that the staff and members of Proshika were used in the alliance against the existing government. In this condition, NGOs were not considered as autonomous, rather than it was known as the extension of the existing power structure (Hossain 2006). The NGO staff members said that they often faced extraordinary challenges both at a personal and organisational level. They reported that they had to work long hours with limited resources in uncertain and volatile political and economic circumstances in order to help the most marginalised and disadvantaged members of their communities (Hailey 2006).

8.5 LACK OF SOCIAL TRUST

In Bangladesh, the lack of 'social trust' has become a crucial barrier for NGOs' work (Islam 2015a). I mentioned earlier in this chapter that a number of networks, such as NGO staff members, community people, local leaders, local institutions and administration, funding agents, the government, and civil society, were involved within NGOs' initiative through collective actions and partnership contacts. I found that the 'trust' is absent within and between these networks. For example, one community leader of PAB said:

I have no faith in PAB. What are they doing for the blacksmiths? I did not find anything there. I heard that they provided some training, but they did not give any kind of capital to utilise their training. It is worthless.

Due to lack of trust, NGOs' social enterprise concept could not function well. It was one of the barriers to achieve mutual benefits. One black-smith's opinion was:

Why would I share my business with others? It is my business and I want to get all money as I earn. Why would I give and share my earnings and experience to others? It is my property. I don't think that everybody has trust in each other. Sir, please do not do anything else like that. If you do so I am not with you. We will protest that.

I found that the trust between the cross-networks, such as NGOs with donors or NGOs with community people or community leaders/religious leaders or civil society and vice versa, was low. The trust between NGOs and smiths was limited to the 'give and take' policy as many smiths reported that their relation with NGOs was just for 'loan purposes'. There was neither any mental or social attraction, nor any kind of commitment, other than the financial. I argue that such exchange of trust is so slight and ephemeral that it could not give any guarantee to achieve any kind of long-term social change towards achieving community development. The smiths never believed that Proshika would do something for them in crisis periods, such as chatabaji (money collected by someone illegally), business loss, leadership crisis, local conflict and control over the local market. The goldsmiths of SEED found that they did nothing when they suffered from those problems. In response to the question about the involvement with SEED, one goldsmith answered that, like other NGOs, Proshika is a 'new moneylender' (Hossain 2006; Davis 2006), and it was helpful to get some loan from them. His comment was:

When there is no way to get loan from elsewhere then they take loans from SEED.

This kind of 'NGO dependency' is the dark side of social capital (Huda et al. 2005). On the other hand, the blacksmiths were not significantly happier whether or not PAB would provide them other supports, such as further training, loan incentives and help during leadership crisis, as they found nothing before.

The community leaders of both NGOs said that they were not happy to say that one NGO's relation with another or relevant organisations was good. They observed that the NGOs' relation in regards to knowledge and experience sharing was not cooperative; rather, some were hostile to each other. One community leader of SEED said:

I don't find one NGO's relation with another is cooperative. Rather I find there is a cold war among them. I am saying this to see their overlapping activities, commercial motive and exclusion the 'ultra' poor from their activities. I heard from my village neighbours that many NGO staff members caution that they will withdraw their support if they go to another NGO.

The lack of trust from donors is a fundamental problem in the NGO sector. There remains a huge gap between donors and NGOs as to what they expect from each other. It is because some NGOs, such as Proshika, were trying to compromise between the Western knowledge and pressures, which, because of financial supports, became complex. Wallace and Chapman (2003) referred to an experience in Africa, which indicated that in spite of a commitment to participation and bottom-up approaches there was a recognised need for sustainability. They also found that the focus was still on 'us' solving problems for 'them' (be they individuals, households, communities or governments) and little trust was given to agencies.

I found that NGOs' political involvement and corruption limited their capacities. Due to these kinds of involvement, many local people, community leaders, local administrators and the government could not trust Proshika as this NGO was patronised by a political party. It was observed that many local people, who had no political involvement or were of specific political ideologies, withdrew their membership from Proshika, and many social agents became their rivals. NGOs' involvement in corruption was another negative image where people's general faith was completely hampered. The news of corruption, with NGOs taking money from the local poor peoples' savings, is frequently cited in Bangladeshi newspapers. In a study of 20 NGOs, the Transparency International Bangladesh (2007) found that many NGOs were registered with the government in exchange for bribes and some of them had existence only on paper, and the government officials assisted them in the process. This study did not find any proof of such kind of corruption in relation to these two NGOs but, in general, the local peoples' trust for the NGOs weakens the overall perception to the NGOs' activities. There is no regulatory mechanism in Bangladesh for ensuring local peoples' savings and safety of deposits (Jackson and Islam 2005).

8.6 EXTERNAL INTERVENORS, DONOR DEPENDENCY AND ORGANISATIONAL AUTONOMY

The majority of Bangladeshi NGOs are largely designed, funded and managed externally (Hashemi 1996; Buckland 1998). In connection to this, the government-donor relations in Bangladesh (also stated in Chap. 2) are shaped by a history of donor dependence and reactions against it (Green and Curtis 2005). There were a number of issues were important, where NGOs' community empowerment became challenging. Essentially, NGOs' donor dependency became a matter of power relations (Wallace and Chapman 2003). This suggests that NGOs shared an international community development culture reflecting their own national cultural norms.

With this, the local-level planning, organisational accountability, autonomy and social trust became problematic to apply so-called universal knowledge. The 'accountability' problem is traditionally concerned with the unequal relationships within the aid industry between donors, northern and southern NGOs (Lewis 2007; Edwards and Hulme 1995). It is because the donors bring universal values, selfcolonisation and elitism, individualism and anxiety. The NGOs often pretend to represent 'fashionable' and universally acceptable development ideas, knowledge and skills. As a result, enormous pressure gets put on the rural poor to comply with certain 'universal conditions'. These universal conditions are considered as inappropriate or invalid, where less consideration is given to its validity (Wood 2007) for the local context. It creates dual dependency (Ferguson 1994; Escobar 1995). It is directive, not facilitative (Garilao 1987, 119). Ahmad (2006) calls this 'donorship', rather than partnership. The Bangladeshi NGOs find this inappropriate. This is because there is a fundamental gap between the socio-economic conditions of developed countries and the developing world. This kind of 'imported knowledge' does not always fit well with the national development priorities and development systems. Such kinds of aid-based development practices have long been a barrier to sustainable technology. Schumacher (1973) believes that foreign aid is able to play only a limited role in bringing about sustained economic development. Such substantial input of foreign aid is doing much damage to the spirit of self-respect and self-reliance; its loss is greater than its gains. It creates a 'development' gap, which does not encourage innovative practices. In connection to this point, one PAB staff member stated:

Sometimes the donors do not want to extend their projects or increase the funding for those though it needs to continue these for its local community demands. It is a big problem that we cannot provide any follow-up services for them. We all know that these social development projects need to serve continuously for a longer period for achieving actual outputs. But they are bound to stop those at a 'half-done' stage because of donors' discontinuation of funding supports.

Development is not apolitical, and 'the process of organising and empowering communities and poverty groups is in itself a political act' (Garilao 1987, 119). This de-politicisation (see Ferguson 1994; Escobar 1995) of NGO development efforts, experienced in Bangladesh over the last 20 years, is part of a broader global trend in NGO policies (Edwards and Hulme 1996; Fisher 1997; Hulme and Edwards 1997; Kamat 2004). As a result, the NGOs were increasingly shaped by the Western-dominated international development discourse (Rahman 2006, 456–457). However, this explanation can also be seen as 'one-sided' (Rahman 2006, 456), 'marginalised', question of 'sustainability' (Grootaert 1998; Khan 2006, 174) since this Western development model ignored the important influence of local conditions that the NGOs face as their constraints and policy choices. Within this discourse, do aid agencies have the right to be the 'voices of the poor' and to decide correct approaches for humankind and to speak on behalf of grass-roots' communities such as blacksmiths and goldsmiths? This study found that this Western model could not explicitly promote participation, but created confusion among smiths versus NGO workers, due to the lack of social trust.

This study conjectured that around the aid process there was insufficient dialogue or an attempt at dialogue that would sometimes be totally unheard. Partnership relations in NGOs' activities in Bangladesh are inhibited by the instrumentality that each side brings to the equation: donors seeking conformity with current reform prescriptions or conditionality clauses, recipients seeking least (political) cost thresholds (Green and Curtis 2005). In this condition, both parties felt constrained in being more responsive and accountable to their superiors than to others. Like Green and Curtis (2005), I believe that under these circumstances improved donor coordination in any form may be conceived more as a threat than an opportunity by the national government. In addition to this, the South Asian NGOs have tensions regarding cutting-off funding supports from the donors. Globalisation, new (and ever changing) trade agreements and aspects of the emerging new international political order all contain perceived threats.

Due to donors and external pressures, NGOs' organisational autonomy regarding low managerial capacity (Kusumahadi 2002) and inconsistent with the operations (Hulme and Edwards 1997) is now an issue. Both NGOs had mixed opinions about this. Some NGO staff members observed that there were some barriers to increase control over programme management, where a certain level of external/donor pressure existed. Some staff members said that they still had shortages of funds and had to depend on donors. I found that PAB took funds from a number of donor agencies, such as DFID, ETC Foundation and KATALYST during 2001-2006. They faced some problems from the donor agencies, such as project discontinuation and lack of flexibility. Some PAB staff members said that their central management was not decentralised enough to implement some local peoples' urgent demands, such as loan supports, equipment supply and supports during natural disaster. It was seen that both NGOs were based on the new public management (NPM)³ approach. Both NGOs' staff members believed that too much donor dependency was bad for an organisation. An organisational self-assessment can facilitate some serious realisations to do with investments in learning versus investments in doing, policy awareness versus policy influence, insulation versus influence and independence versus partnership. A foreign partner may damage a local NGO's credibility and effectiveness, especially as a leading voice in the policy arena.

However, in crisis states capacity is often limited, which increases the risk of corruption (Larbi 1999). This was particularly true for Proshika as I stated earlier in this chapter. Moreover, the public-sector reforms were externally driven by the donor conditions and timetables. The overambitious nature and the demand for quick results failed to take account of weak institutional and management capacities (Larbi 1999). These kinds of capacities can use a new technocratic language, whilst failing to deal with political problems and contradictions arising between the situation of people in developing societies and the fashionable neo-liberal ideology (Arce 2003, 855). The study found this kind of evidence in both NGOs' interventions. However, I argue that the NGOs

need to develop partnerships with consideration of fair measures and accountability. It is necessary for funding organisations and NGOs to put themselves in each other's shoes in order to understand better their mutual constraints.

8.7 Religious Leaders Frustrate Many Initiatives

The overall attitude of the religious leaders to the NGOs was very negative; this situation was worse in Proshika than PAB. Traditionally, the religious leaders are hostile to NGOs' activities such as those of Proshika. There were three main causes, for example, political involvement with the progressive political party BAL, using donations from Western country sources and partnerships with them, and taking 'interest' through microcredit for women's empowerment. According to the Islamic belief, interest, in the financial sense, is prohibited. In addition, some extremist interpretations of Islam prohibit some forms of women's empowerment, which Proshika was advocating. This problem was greater in rural areas than in urban. The extent of this problem was not great among the goldsmiths, but it was still problematic among the businessmen who are Muslims. According to a staff member of SEED:

The Mullah (religious leaders) groups are our main enemy; they want to demolish Proshika's activities. We know that some Mullahs are committed and that they will continue their jihad (a holy war) until they win against us.

As a result, many federations and groups of Proshika could not progress their work. Proshika had 14,000 dysfunctional groups. Still it was not clear how many groups and federations were really functioning. More than half of the groups and federations had collapsed because of organised propaganda and harassment by the extremist religious groups and gangs orchestrated by political leaders. This situation was aggravated by the competition from credit-providing NGOs and the Islamic Bank, and controversial allegations against Proshika. This complex combination of realities had resulted in anxiety among some members, decreased attendance in meetings, and regular savings and withdrawal of savings in some areas of Proshika (Yildiz et al. 2003).

8.8 LOCAL POWER STRUCTURE, INACCESSIBLE MARKET AND LACK OF GOS' HELP: EVOLVING NEW INITIATIVES

The local power structure, mainly élites, is a threat in both advanced societies and developing countries (Malki 2006). I found that the local power structures changed and there evolved new forms of power structures through a slow transformation. Edwards et al. (1999, 120) described the civil society as a 'magic bullet'. The changing power structures often conformed to or used approaches utilising civil society (Edwards et al. 1999, 120), middle class (Hossain 2006) and local NGO workers (Hossain 2006; Rahman 2006). However, I found that such power structures were still part of the so-called élite class and not ready to lend itself to building positive social networks and community empowerment. They had political links and worked against traditional NGOs (Hossain 2006, 236–237; Rahman 2006, 452). The local government bodies continue to be dominated by élites (Khan 2001).

The chairman of a local NGO—Jana Unnoyan Bahumukhi Shamabai Shamity (Multi-cooperative Association for Peoples' Development) at Mostafapur Bazar of PAB-was a local political leader of BAL. This finding was supported by Hossain's (2006) study. He conducted a study of two local NGOs (Voluntary Family Welfare Association and Karmajibi Kallan Sangstha (Workers' Welfare Association)) in the Rajbari District in Bangladesh and found that the executive directors of both NGOs had links with the main political parties. It is a relatively recent norm that local power structures are more dynamic, diverse with higher occurrences of bargaining. Internally, these power structures seek to build power bases in the village through pursuing a 'politics of reputation'. Externally, they are active with a broad range of strategies for securing and maintaining external relationships, such as 'NGO-ing', party political manoeuvring and securing patronage from upazila-level administration. As a result, the decision-making process was skewed towards those with more power and resources (Edwards et al. 1999, 120).

This repeated reorganisation of local government structures has not only weakened the power and authority of local government; it has also undermined the legitimacy of local government as an institution for expressing the voice of the citizenry. It has been observed that the local government had become simply another element of the élite contest for political power (Khan 2001). Combined with increasing corruption in local government and in the provision of social services at the local level, the rural poor had come to see political participation at the local government level as irrelevant, relying instead on kinship and patronage networks for assistance and support (Rahman 2006). In this situation, the local leaders thought that either the NGOs were trying to do something for the local people or the NGOs were taking over the leadership role they hold. As a result, their role with and control over local people weakened. In relation to this, local money launderers, most of them community leaders, thought that Proshika was a big threat to their traditional business, so they viewed NGOs as the opposition. Due to oppression and deprivation, they could not provide opportunities to develop individual capacity (Sen 1999; Rifkin 2003, 169).

I mentioned earlier that the 'syndicate' controls the business sector in Bangladesh, where a group of a few wealthy individuals are controlling this. The smiths were unable to access the wider market. They were stocking all products and controlling the market prices. It was seen that the NGOs failed to work with the market channels as they had outlets through the market map as social networks. The smiths were bound to sell their products at a low price to those brokers; then the wholesalers buy from them and in turn the consumers buy at high prices. Sometimes, the consumers buy faulty products as the brokers and wholesalers sometimes adulterate the products for their own profit. This process destroyed social trust among smiths and local consumers, while smiths had rival relations with brokers, wholesalers and vice versa.

I found that the local wholesalers (known as rich and local leaders and financially rich people) persuaded the blacksmiths (financially poor and uneducated) that the market was not able to pay them a high price for their products and such kinds of undermining and discouragement deterred them. Because of this demoralisation, the poor class remained poor and its transformation was very slow.

The relationship of policy makers (government), funding authority (donors) and implementing organisations (NGOs) in Bangladesh was not cooperative. As a result, many projects could not minimise political and institutional limits and pressure of the policymakers, which resonated with their ideological assumptions. The findings of the research can also be compared with Ahmad's study (2003), which found that NGOs could create new social capital through induced collective action. However, their efforts were hampered because of the short period of time it had been in operation and because of weak government support. The government has always participated, to a relatively

limited extent, as a provider and regulator for microfinance in Bangladesh (Hulme and Moore 2006). It is because historically the relationship between the Bangladeshi government and the NGO sector has ranged 'from benign neglect, to co-option, to smear campaigns and repression', depending on the state's 'perception of the balance of power between itself and the NGO sector'.

8.9 CENTRALISATION IN DECENTRALISED MANAGEMENT

The NGOs' 'control',⁴ otherwise known as 'ownership' (Jones and Laverack 2003, 9) over programmes, is one of the key elements of community empowerment. It is comparatively more important to the Bangladeshi NGOs whose funding depends on foreign donations. I stated earlier in this chapter that due to 'donor dependency' most of the NGOs have lost their management control. Either many community leaders and smiths thought that these NGOs were controlled and managed by the donors, who were ignoring the local context or imposing GK, or they were trying to destroy local cultures and skills. Many stakeholders thought of it as a form of 'westernisation'⁵. Many staff members agreed that their management system was the 'personal leadership style'⁶ of administration, where the decision came from the top, and some were neglected whilst others were delayed because of bureaucratic problems. Lewis et al. (2008, 10) argue that the managers play the lead role and all other actors remain silent, whilst the organisations unfold their objectives in a participatory way. That is why the NGOs sometimes did not provide many of their funding and resources as quickly as the field-level NGO staff members wanted. Sometimes, the staff members at field level observed that they did nothing as the central-level staff members could not go beyond the restriction of the central management. One fieldworker of PAB said:

The blacksmiths have been demanding financial support for the last two years, but I could not give any hope to them. There is nothing I can do for them, but I feel they need that. I came to know from the central authorities that they have no funding support for this project except training.

Some other problems were NGOs' lack of communication and turnover of staff. PAB could not run many projects properly as they had no offices at the grass-roots level. Most NGOs utilise a branch and headquarters structure in which branches had limited autonomy from

headquarters (Gauri and Galef 2005, 2045). PAB's local office at Faridpur controlled the project offices of Faridpur, Madaripur and Barisal districts. Some of those were 150 km away from their local office. In some areas, they had partnership NGOs such as GUP, which managed the project for the blacksmiths. However, PAB was unable to convey many of its ideas to GUP properly. Some senior-level staff members of SEED stated that junior staff members could not provide efficient advocacy and counselling services as their educational background was insufficient. It found that more than 90 % of the SEED staff members were from a 'commerce' background, which was dominated by the head who was from this background. I argue that a 'multidisciplinary approach' (with different disciplines) is needed together with 'credibility' based on technical expertise (VanSant 2003). The staff turnover was another problem for Proshika. The annual turnover rate was high during year 3 of phase VI. The lack of promotions, low salaries, expressed dislike of the emphasis on loan collection and uncertainties about Proshika's future were reasons for staff leaving. It was difficult to recruit good people in any low-status, low-pay occupation and very difficult to keep them (Ahmad 2007, 362). Because of this turnover, Proshika failed to respond to the community adequately.

8.10 Limitations of Social Capital and Community Empowerment Approaches

The NGOs' capacities, derived from the theories of social capital and community empowerment, had significant implications for Bangladeshi communities. There were some other areas where these two development approaches failed to generate a substantial amount of positive results. It is acknowledged that the NGOs were not based on these approaches, but their programme outcomes were related to the components and objectives of those approaches. In general, the NGO staff members had little theoretical knowledge about social capital and community empowerment approaches used in the research, though they still practised them. Moreover, the measurement of social capital and community empowerment within the role of NGOs was not easy. This is because the definitions used for both concepts are vague, qualitative and open to subjective interpretation. In addition, many of these approaches have limitations in terms of broad applications in the Bangladeshi community. In some cases, the NGOs did take initiatives, but due to some community problems, the NGOs could not substantiate those in proper ways. Moreover, NGOs' social contributions could not be visualised entirely.

The notion of social capital has been articulated as a concept, which is homogeneous and recognises primarily 'positive' social norms, values and practices. The fact is that the communities were rarely discrete entities, and some communities have greater power to define and legitimise meaning, values, norms and practices than others. Bangladesh and other poor countries are immersed in multidimensional relationships, whether these are hierarchical or reciprocal. Here, the multitransactional ties were strong, both between patrons and their clients as well as more reciprocal within kin groups (Wood 2005, 16). The power relations structured access to community membership and intra- and intercommunity conflict and competition. Putnam refers to the crucial 'bridging function' of social capital, but gives little attention to inequalities in the distribution of income or other economic, physical, environmental and social advantages, which structure relations in the field of community life (McClenaghan 2000). The lack of consideration of power as an important shaping feature of social capital might lead to horizontal associations, while neglecting vertical relationships (Bayat 2005, 10).

Clearly, the traditional associational ties and forms of closure to which Putnam and Coleman refer may well contribute to the reproduction of these advantages and, by implication, the reproduction of disadvantage and exclusion. My findings showed that inequality in terms of loan distribution, distribution of wealth and resources, power relations, social control and decision-making in both communities made it difficult to get the benefits of social networks. It also encouraged a process of 'political mobilisation' to advance demands based on rights, rather than on rebuilding the community through a process of coordination aimed at developing and enhancing social cohesion. This process clearly reinforced the very exclusion that community development activists in disadvantaged communities, such as Mirpur (1) Market and Mostafapur Bazar, were struggling against to those powerful groups (McClenaghan 2000). Sometimes, the social network or relationship pattern was more or less linked with interest-gaining processes, which sometimes stopped the natural flow of social relationships (Purvez 2003) as found through Proshika's political networks.

The findings also confirmed that the NGOs' initiatives towards development created more problems than solutions for powerless people (Kusumahadi 2002, 12), such as blacksmiths and goldsmiths. The prescription of the components of community empowerment approach provided by Laverack (2006) was found to be ambiguous. Laverack listed nine components as 'domains', which the stakeholders felt too many, and some were duplicated. The notion of their feelings about these elements was not positive as they found them complex, long, gloomy, loose and incomplete, and, to some extent, obscure. Moreover, Laverack emphasises the role of small groups in the community, which can improve problem assessment capacity. However, practically, in urban society, the NGO workers said that it was not possible to establish such kinds of groups properly as they faced busy time constraints. I think that Laverack also avoided some important elements; some could be shortened or left out to avoid repetition. For example, the stakeholders could not find the real difference between the 'increase problem assessment capacities' and 'critical awareness about people's needs'. Furthermore, two other elements such as the 'strengthening links to other organisations and people' and 'create equitable relationships with outside agents' looked similar and were repetitious of one another. The reality is that it should be distinguished in such a way that the community people, community leaders and NGO staff members could understand. They could be able to apply them properly.

Secondly and paradoxically, Laverack's approach appears incomplete. I personally feel that there should be added some other important elements such as fulfilling community 'felt needs' and 'real needs'. I found that community participation was not fully secured by increasing people's assessment capacities and critical awareness as they actually felt that they need to fulfil some felt needs immediately, then to address their real needs. It can be called 'social welfare', which is included in the study Kusumahadi (2002). I also think that the elements they used to 'increase access' and 'social control' are two other elements which should be added to Laverack's empowerment approach.

To increase access is also important to build and promote community capacity, and to achieve greater access to all the resources needed. On the other hand, social control is also important to increase control of the environment and all aspects of life, for example, social affairs, economy, politics and technology. The initiatives of the organisational structures and increase control over management could not save the NGOs from bureaucracy and corruption, where there evolved a 'personal type of management'. Ultimately, it failed to improve real participation, mobilise local resources and accommodate outside interventions properly. Significantly, the NGOs failed to improve asset endowments and opportunity structures of the community (Bodja 2006, 26).

Empowerment, it found, was influenced more by commitment rather than by understanding (Kusumahadi 2002). My own assertion is that there needs to be increased access of the smiths to resources and capital, and the NGOs should increase social controls, where their supervision and monitoring systems can work smoothly. The final element 'increase control over programme management' is phrased badly. It should be restated as 'increase self-management'. The concept 'self-management' is more decentralised, understandable and pursuable. This approach would be considered as more convenient for the NGOs as a function of their direct links with grass-roots organisations to gain a better insight into what kind of programmes are needed and desired by the poor themselves. It would also be helpful to grasp comparative advantages and experiences at the micro level be brought to bear on the formulation of alternative and hopefully more successful development strategies.

8.11 CONCLUSIONS

This chapter identified some critical aspects of NGOs' interventions, where social capital and community empowerment approaches would be more feasible for developing countries such as Bangladesh. Both NGOs were using IK and GK through their interventions, though the use of GK is increasing over time. The analysis shows that the NGOs were facing some contradictions to use these knowledge approaches. In many cases, the NGOs were negotiating to integrate these two knowledge approaches to overcome those contradictions. However, the overall assessment of NGOs' capacities was convincing in terms of community awareness, sharing values and norms, introducing new technologies, increasing human rights and problem assessment capacities, responding to financial crisis, encouraging participation, mobilising resources and providing income generation activities.

On the other hand, NGOs' contribution towards collective actions, sharing mutual benefits, developing local leadership, empowering organisational structure and increasing control over management was weak. The NGOs partially failed to create a congenial atmosphere, where the smiths could work properly in the market. They could not build up effective networks with negotiation of GOs, community leaders, community people and with the external donors. In addition, the study found that NGOs' political networks had developed a negative trend towards human services. More or less both NGOs had money laundering tendencies, leadership and management crises, and donor dependency. It is important that both NGOs recognise the spontaneous participation problem from the community people and community leaders. In this connection, NGOs' social trust with community people and GOs became hard-nosed. Moreover, the local peoples' low socio-economic, cultural and political conditions were other barriers to authentic community empowerment of the NGOs.

Notes

- 1. The question of downward accountability is sensitive to power issues and the word 'beneficiaries' is problematic in its own right. The term 'downward' reinforces the idea of power asymmetry. Kilby (2006, 961) cautions that the use of the term 'downward' accountability can exaggerate the weakness of the beneficiary or client and so suppresses the essential ingredient of authority inherent in the accountability relationship.
- 2. ADAB is a unique network of NGOs developed over the last two decades or so and emerged as a model for the development NGOs across the world. It brings together its members working within the broad framework of people-centred sustainable development. One of its central goals is to knit these constituent parts closer together and respond to their needs creatively. It has been playing active roles in strengthening inter-NGO relations, exchanging ideas and experiences, expanding fields of cooperation, developing organisational skills and enhancing functional relations and communication with government, donors and various groups of civil societies. ADAB is registered simultaneously under the Societies Registration Act of 1860 and Foreign Donations Regulation Ordinance of 1978. Currently, ADAB has a membership of 886. Of these, over 231 (as of December 1997) are at the central level and 655 are at the chapter level (SDNP n.d.).
- 3. NPM originated from the crisis of the Keynesian welfare state in the 1970s in developed countries such as the UK, Australia and New Zealand. The use of management techniques and practices drawn mainly from the private sector is increasingly seen as a global phenomenon. NPM reforms have shifted the emphasis from traditional public administration to public management. Key elements include various forms of decentralising management within public services (e.g. the creation of autonomous agencies and devolution of budgets and financial control), increasing use of markets and competition in the provision of public services (e.g. contracting out and other market-type mechanisms) and increasing emphasis on performance, outputs and customer orientation (Larbi 1999).
- 4. Control is generalised to the broader range of organising efforts such as administration, finance and management (Laverack 2006, 5).

- 5. Westernisation is a process whereby non-Western societies come under the influence of Western culture in matters such as industry, technology, law, politics, economics, lifestyle, diet, language, religion or values. Westernisation has been a pervasive and accelerating influence across the world in the last few centuries. It is usually a two-sided process in which Western influences and interests themselves are joined by a wish of at least parts of the affected society to change towards a more westernised society in the hope of attaining Western life or some aspects of it (Encyclopedia of the Modern Middle East and North Africa 2004).
- 6. It is defined as self-confidence backed by conviction and understanding. Another term might be 'personal mastery'—knowing where you stand, where you want to go and what you need to sacrifice to get there. Whatever you call it, personal leadership stems from one's core beliefs and values and is essential to leading others (Heider, n.d.).

References

- Abom, Bill. 2004. Social Capital, NGOs, and Development: A Guatemalan Case Study. Development in Practice 14 (3): 342–353. doi:10.1080/ 0961452042000191187a.
- Ahluwalia, Isher J., and Wahiduddin Mahmud. 2004. Economic Transformation and Social Development in Bangladesh. *Economic and Political Weekly* 39 (36): 4009–4011.
- Ahmad, Mokbul Murshed. 2003. Whose Ideas? Field Workers' on Non-government Organizations' Policies in Bangladesh. *International Development Planning Review* 25 (1): 67–84. doi:http://dx.doi.org/10.3828/idpr.25.1.4.
- Ahmad, Mokbul Murshed. 2006. The 'Partnership' between International NGOs (Non-government Organizations) and Local NGOs in Bangladesh. *Journal of International Development* 18: 629–638. doi:10.1002/jid.1251.
- Ahmad, Mokbul Murshed. 2007. The Career of NGOs Field Workers in Bangladesh. *Non-profit Management and Leadership* 17 (3): 349–365. doi:10.1002/nml.154.
- Albu, Mike, and Alison Griffith. 2005. Mapping the Market: A Framework for Rural Enterprise Development Policy and Practice. Bourton (UK): Practical Action.
- Anthromorphemics. n.d. Patron Client Relationship. California: Department of Anthropology, University of California. http://www.webref.org. Accessed 12 October 2008.
- Arce, Alberto. 2003. Re-approaching Social Development: A Field of Action between Social Life and Policy Processes. *Journal of International Development* 15: 845–861. doi:10.1002/jid.1039.
- Ashman, Darcy, and Carmen Luca Sugawara. 2013. Civil society networks: options for network design. *Nonprofit Management and Leadership* 23(3): 389–406.

133

© The Author(s) 2016

M.R. Islam, NGOs, Social Capital and Community Empowerment in Bangladesh, DOI 10.1007/978-981-10-1747-6

- Asian Development Bank (ADB). 1998. Cooperation between ADB and NGOs. Manila: ADB.
- Asian Development Bank (ADB). 1999. A Study of NGOs Bangladesh. Bangkok: ADB.
- Asian Development Bank (ADB). 2008. ADB and Bangladesh 2008: The fact sheet. Dhaka: ADB.
- Bagci, Cahit. 2007. Historical evolution of NGOs: NGO proliferation in the post-Cold War era. *Avrupa Gunlugu* 4: 299–326.
- Banks, Nicola, and David Hulme. 2012. The Role of NGOs and Civil Society in Development and Poverty Reduction. *BWPI Working Paper 171*. Manchester: The Brooks World Poverty Institute.
- Barnes, Martha, Rebecca Rodger, and Colleen Whyte. 1997. Empowerment through Community Development in Recreation and Leisure. *Journal of Leisurability* 24 (1): 1–7.
- Barr, Julian, and Peter Dixon. 1998. Incorporating Farmers' and Fishers' Knowledge into Natural Resources- Systems Research on the Bangladesh Floodplains. *Tropical Agriculture Association Newsletter* 19 (1): 32–35.
- Baum, David. 2000. Lighting in a Bottle: Proven Lessons for Leading Change. Dearborn: Chicago.
- Bayat, Abdullah. 2005. Defining Social Capital: A Brief Overview of the Key Aspects and Debates. Bellville: University of the Western Cape.
- Bebbington, Anthony. 1997. New States, New NGOs? Crises and Transitions among Rural Development NGOs in the Andean Region. World Development 25 (11): 1755–1765. doi:10.1016/S0305-750X(97)00066-1.
- Bebbington, Anthony. 2004. NGOs and Uneven Development: Geographies of Development Intervention. *Progress in Human Geography* 28 (6): 725–747. doi:10.1191/0309132504ph5160a.
- Bebbington, Anthony. 2005. Donor–NGO Relations and Representations of Livelihood in Non-governmental Aid Chains. World Development 33 (6): 937–950. doi:10.1016/j.worlddev.2004.09.017.
- Bello, Walden, Shea Cunningham, and Li Kheng Po. 1998. A Siamese Strategy: Development and Disintegration in Modern Thailand. London: Zed.
- Bendana, Alejandro. 2006. NGOs and Social Movement a North/South Divide. Geneva: United Nations Research Institute for Social Development (UNRISD).
- Bodja, Sebilu. 2006. A Critical Perspective on Community Empowerment: The Selected NGOs in Ethiopia. Unpublished master's dissertation. Cape Town: University of Western Cape.
- Bornstein, David. 2004. How to Change the World Social Entrepreneurs and the Power of New Idea. Oxford: Oxford University Press.
- Bourdieu, Pierre. 1986. The Forms of Capital. In Handbook for Theory and Research for the Sociology of Education, ed. John Richardson, 241–258. Westport, CT: Greenwood Press.

- Boven, Karian, and Jun Morohashi. 2002. Best Practices Using Indigenous Knowledge. The Hague: UNESCO/MOST and Nuffic.
- Bowen, Glen A. 2008. An Analysis of Citizen Participation in Anti-poverty Programmes. *Community Development Journal* 43 (1): 65–78. doi:10.1093/cdj/bsm011.
- Briggs, John. 2005. The Use of Indigenous Knowledge in Development: Problems and Challenges. *Progress in Development Studies* 5 (2): 99–114. doi:10.1191/1464993405ps105oa.
- Briggs, John, Joanne Sharp, Hoda Yacoub, Nabila Hamed, and Alan Roe. 2007. The Nature of Indigenous Environmental Knowledge Production: Evidence from Bedouin Communities in Southern Egypt. *Journal of International Development* 19: 239–251. doi:10.1002/jid.133.
- Briggs, John, and Johanne Sharp. 2004. Indigenous Knowledge and Development: A Postcolonial Caution. *Third World Quarterly* 25 (4): 661–676. doi:10.1080/01436590410001678915.
- Brinkerhoff, Derick W., and Omar Azfar. 2006. *Decentralization and community empowerment: Does community empowerment deepen democracy and improve service delivery*. Washington, DC: US Agency for International Development, Office of Democracy and Governance.
- Bromideh, Ali Akbar. 2011. The Widespread Challenges of NGOs in Developing Countries. *International NGO Journal* 6: 197–202. doi:10.5897/NGOJ11.014.
- Brown, David. 1994. Seeking the Consensus: Populist Tendencies at the Interface between Research and Consultancy. *Paper presented at the workshop from Consultancy to Research*. Swansea: University of Wales.
- Brown, L. David, and Darcy Ashman. 1996. Participation, Social Capital and Intersectoral Problem-solving: African and Asian Cases. *World Development* 24 (9): 1467–1479. doi:10.1016/0305-750X(96)00053-8.
- Buckland, Jerry. 1998. Social Capital and Sustainability of NGO Intermediated Development Projects in Bangladesh. *Community Development Journal* 33 (3): 236–248. doi:10.1093/cdj/33.3.236.
- Buckland, Jerry. 2004. Globalization, NGOs and Civil Society in Bangladesh. In *Globalization and Community: Canadian Perspectives* ed. Jean-Luc Chodkiewicz and Ramond Wiest, Paper 34. Winnipeg: University of Manitoba.
- Burns, Paul. 1996. Introduction. In *Small business and entrepreneurship*, eds. Burns, Paul, and Jim Dewhurst, 1–7. London: Macmillan Education.
- Burt, Roland S. 1992. Structural Holes: The Social Structure of Competition. Cambridge: Harvard University Press.
- Cahn, Miranda. 2008. Indigenous Entrepreneurship, Culture and Microenterprise in the Pacific Islands: Case Studies from Samoa. *Entrepreneurship & Regional Development* 20 (1): 1–18. doi:10.1080/08985620701552413.
- Cajete, Gregory. 2000. Native Science: Natural Laws of Interdependence. Santa Fe, NM: Clear Light Publishers.

- Carlisle, Sandra. 2006. 'Thinking about 'Culture' as an Influence on Health and Wellbeing.' *Discussion paper 3.* Glasgow: University of Glasgow.
- Carroll, Thomas F. 2001. Social Capital, Local Capacity Building and Poverty Reduction. *Social Development Paper No. 3.* Manila: ADB.
- Chambers, Robert. 2004. Ideas for Development: Reflecting Forwards. *IDS Working Paper 238.* Brighton: Institute of Development Studies (IDS).
- Chaudhury, I. A. 2006. Sustainable Livelihoods through Capacity Building and Enterprise Development, Documenting the Evidence and Lessons Learned. Dhaka: Practical Action Bangladesh.
- Cheng, Yin Cheong. 2003. Local Knowledge and Human Development in Globalization of Education. Keynote Speech Presented at The International Conference on Globalization and Challenges for Education organized by National Institute of Educational Policy and Administration (NIEPA), 2003 New Delhi, India.
- Cheung, Stefan Yan-Leung. 2006. Social Enterprises and Entrepreneurs. *Conference on Social Enterprise*. Hong Kong Convention and Exhibition Centre, City University of Hong Kong, 6 April.
- Clark, John D. 1991. Democratising Development: The Role of Voluntary Organizations. West Hartford (UK): Kumarian Press.
- Cleary, Seamus. 1997. The Role of NGOs under Authoritarian Political Systems. London: Palgrave Macmillan.
- Coleman, James S. 1988. Social Capital in the Creation of Human Capital. *The American Journal of Sociology* 94 (supplementary): S95–S120.
- Coleman, James S. 1990. Foundation of Social Theory. Cambridge: Harvard University Press.
- Columbia Encyclopedia. 2003. Awami League. 6th ed., Columbia: Columbia University Press.
- Cook, Karen Schweers. 2005. Networks, Norms, and Trust: The Social Psychology of Social Capital. *Social Psychology Quarterly* 68 (1): 4–14. doi:10.1177/019027250506800102.
- Correa, Carlos M. 2001. Traditional Knowledge and Intellectual Property- Issues and Options Surrounding the Protection of Traditional Knowledge. *Discussion Paper*. Geneva: Quaker United Nations Office.
- Cosgrave, Serena. 2002. Levels of empowerment: marketers and micro-enterpriselending NGOs in Apopa and Nejapa, El Salvador. *Latin American Perspectives* 29(5): 48–65.
- Craig, Gary, Marjorie Mayo, and Marilyn Taylor. 2000. Globalization from Below: Implication for the Community Development Journal. Community Development Journal 35 (4): 323–335. doi:10.1093/cdj/35.4.323.
- Dale, Ann, Trudeau Fellow, and Jennie Sparkes. 2008. Protecting Ecosystems: Network Structure and Social Capital Formation. Community Development Journal 43 (2): 143–156. doi:10.1093/cdj/bsm007.

- Davis, John. K. 2006. NGOs and Development in Bangladesh: Whose Sustainability Counts? Perth (Australia): Murdoch University.
- Deacon, Bob. 2000. *Globalization and Social Policy: The Threat to Equitable Welfare.* Geneva: United Nations Research Institute for Social Development (UNRISD).
- DeFilippis, James. 2001. The Myth of Social Capital in Community Development. *Housing Policy Debate* 12 (4): 781–806.
- Delors, Jacques. 1996. International commision for education for 21st century. Learning: The tresures within report to the UNESCO. Paris: UNESCO.
- Department for International Development (DFID). 2009. DFID's work in Bangladesh. Dhaka: DFID.
- Devine, Joe. 1996. NGOs: Changing Fashion or Fashioning Change? Occasional Paper. Bath (UK): Centre for Development Studies (CDS), University of Bath.
- Devine, Joe. 2003. The Paradox of Sustainability: Reflections on NGOs in Bangladesh. Annals of the American Academy of Political and Social Science 590: 227–242.
- Dreze, Jean. 2004. Don't forget India's poor. Time Magazine 164(23).
- Dowla, Asif. 2006. In Credit We Trust: Building Social Capital by Grameen Bank in Bangladesh. *Journal of Socio-Economics* 35 (1): 102–122. doi:10.1016/ j.socec.2005.12.006.
- Dudley, Eric. 1993. The Critical Villager: Beyond Community Participation. New York: Routledge.
- Ebrahim, Alnoor. 2001. NGO Behavior and Development Discourse: Cases from Western India. Voluntas: International Journal of Voluntary and Nonprofit Organizations 12 (2): 79–101.
- Ebrahim, Alnoor. 2004. Seeking NGO-donor Partnership for Greater Effectiveness and Accountability Multilateral Investment Fund (MIF) & Sustainable Development Department (SDS). Final Workshop Report. Inter-American Development Bank (IDB).
- Edwards, Michael. 1999. NGO Performance- What Breeds Success? New Evidence from South Asia. *World Development* 27 (2): 361–374. doi:10.1016/S0305750X(98)00125-9.
- Edwards, Michael, and David Hulme. 1995. Non-governmental Organisations: Performance and Accountability: Beyond the Magic Bullet. London/West Hartford, Earthscan: Kumarian.
- Edwards, Michael, and David Hulme. 1996. Too Close for Comfort? The Impact of Official Aid on Non-governmental Organizations. *World development* 24: 961–974. doi:10.1016/0305-750X(96)00019-8.
- Edwards, Michael, David Hulme, and Tina Wallace. 1999. NGOs in a Global Future: Marrying Local Delivery to Worldwide Leverage. *Public Administration and Development* 19: 117–136.

- Eklund, Leena. 1999. From Citizen Participation towards Community Empowerment: An analysis on Health Promotion from Citizen Perspective. Academic Dissertation. Finland: Faculty of Medicine, University of Tampere.
- Encyclopedia of the Modern Middle East and North Africa. 2004. *Westernisation*. USA: Gale Group, Inc.
- Escobar, Arturo. 1995. Encountering Development: The Making and Unmaking of the Third World. Princeton (New Jersey): Princeton University Press.
- Eslake, Saul. 2000. The Drivers of Globalization. Presentation to a Regional Defence Seminar hosted by the Australian Department of Defence Cypress Lakes Resort, New South Wales, 20 November.
- Evans, Peter. 1996. Government Action, Social Capital and Development: Reviewing the Evidence of Synergy. *World Development* 24 (6): 1119–1132. doi:10.1016/0305-750X(96)00021-6.
- Farrington, John, Anthony Bebbington, Kate Wellard, and David Lewis. 1993. Reluctant Partners? Non-governmental Organizations, the State, and Sustainable Agricultural Development. London: Routledge.
- Fatmi, Mohammad N. E., and Mashrur Islam. 2001. Towards a Sustainable Poverty Forming Model the Proshika Approach. In *Challenges for Second Generation Microfinance, Regulations, Supervision and Resource Mobilization* ed. Iffath Sharif and Geoffrey D. Wood. Dhaka, University Press Ltd.
- Fedderke, Johannes, Raphael de Kadt and John Luiz. 1999. Economic Growth and Social Capital: A Critical Reflection. *Theory and Society* 28: 709–745.
- Ferguson, James. 1994. The Anti-politics Machine: Development, Depoliticization, and Bureaucratic Power in Lesotho. Minneapolis: University of Minnesota Press.
- Fisher, William F. 1997. Doing Good? The Politics and Anti-politics of NGO Practices." *Annual Review of Anthropology* 26: 439–464.
- Fox, Jonathan. 1997. The World Bank and Social Capital: Contesting the Concept in Practice. *Journal of International Development* 9: 963–971.
- Free World Academy. 2005. Global Knowledge for a Global World. http://www. freeworldacademy.com/Accessed 9 January 2007.
- Freel, Mark S. 2000. Barriers to Product Innovation in Small Manufacturing Firms. *International Small Business Journal* 18 (2): 60–80. doi:10.1177/ 0266242600182003.
- Friedmann, John. 1996. Rethinking Poverty: Empowerment and Citizen Right. International Social Science Journal 148: 161–172.
- Friedman, Thomas L. 1998. The Lexus and the Olive Tree. New York: Times Books.
- Fruttero, Ana, Varun Gauri. 2005. The Strategic Choices of NGO: Location in Rural Bangladesh'. *The Journal of Development Studies* 41 (5): 759–787. doi:10.1080/00220380500145289.
- Garilao, Emesto D. 1987. Indigenous NGOs as Strategic Institution: Managing the Relationship with Government and Resource Agencies. *World Development* 15 (Supplement): 113–120. doi:10.1016/0305-750X(87)90149-5.

- Gauri, Varun, and Anna Fruttero. 2003. Location Decisions and Non-governmental Organization Motivation: Evidence from rural Bangladesh. Washington D. C.: World Bank.
- Gauri, Varun, and Julia Galef. 2005. NGOs in Bangladesh: Activities, Resources and Governance. *World Development* 33 (12): 2045–2065. doi:10.1016/j. worlddev.2005.05.009.
- Goodall, Jane. 1986. The Chimpanzees of Gombe: Patterns of Behaviour. Boston: Bellknap Press.
- Gorjestani, Nicolas. 2000. Indigenous Knowledge for Development: Opportunities and Challenges. Washington D. C.: World Bank.
- Government of Bangladesh (GoB). 2007. Bangladesh Millennium Development Goals: Mid-term Progress Report 2007. http://www.undp.org.bd/mdgs/ MDG_Mid-term_Progress_Report2007.pdf. Accessed 8 April 2009.
 Government of Bangladesh (GoB). 2008. Constitution of Bangladesh: Part II,
- Government of Bangladesh (GoB). 2008. Constitution of Bangladesh: Part II, Fundamental rights of state policy. Dhaka: GoB. http://www.pmo.gov.bd/ constitution/part2.htm#P2 Accessed 28 March 2009.
- Government of Bangladesh (GoB). 2011. Population Census 2011. Dhaka: Planning Commission: GoB.
- Grant, Ruth W., and Robert O. Keohane. 2005. Accountability and Abuses of Power in World Politics. *American Political Science Review* 99 (1): 29–43.
- Gray, Mel., Karren Healy, and Penny Crofts. 2003. Social Enterprise: Is it the Business of Social Work? *Australian Social Work* 56 (2): 141–154. doi:10.1046/j.0312-407X.2003.00060.x.
- Green, Lara, and Donald Curtis. 2005. Bangladesh: Partnership or Posture in Aid for Development? *Public Administration and Development* 25 (5): 389–398. doi:10.1002/pad.375.
- Grootaert, Christiaan. 1998. Social Capital: The Missing Link. Washington D. C.: World Bank.
- Grootaert, Christiaan, and Thierry Bastelaer. 2001. Understanding and Measuring Social Capital: A synthesis of Findings and Recommendations from the Social Capital Initiatives. Washington D. C.: World Bank.
- Guhathakurta, Meghna. 2004. Developmental Paradigms: Past Debates and Future Options. In *Bangladesh in the New Millennium* ed. Mohiuddin Ahmad, 113–125. Dhaka: Community Development Library.
- Gunter, Bernhard G., and Rolph Van der Hoeven. 2004. The Social Dimension of Globalization: A Review of Literature. *Working paper no. 24.* Geneva: Policy Integration Department, World Commission of Social Dimension of Globalization, International Labour Organization (ILO).
- Hailey, John. 2006. NGO Leadership Development: A Review of the Literature. Oxford: International NGO Training & Research Centre (ITRAC).

- Hailey, John, and Rick James. 2004. Trees die from the top: international perspectives on NGO leadership development. *Voluntas* 15 (4): 343–53. doi:10.1007/s11266-004-1236-8.
- Hamyaran. 2006. Community Empowerment for Sustainable Development in Iran. Beirut (Lebanon): Fifth Mena Development Forum, Iran NGO Resource Centre.
- Haque, M. Shamsul. 2002. The Changing Balance of Power between the Government and NGOs in Bangladesh. *International Political Science Review* 23 (4): 411–435.
- Harcourt, Tim. n.d. What is This Thing Called Globalization? *A Discussion paper*. New South Wales: Economics' the Association Journal of the Economics and Business Educators Association of New South Wales.
- Harrison, Tom. 2007. The Role of Contestation in NGO Partnerships. Journal of International Development 19 (3): 389–400. doi:10.1002/jid.1373.
- Hashemi, Syed. 1996. NGO Accountability in Bangladesh: Beneficiaries, Donors and the State. In *Beyond the Magic Bullet: NGO Performance and Accountability in the Post-Cold War World* ed. Michael Edwards, and David Hulme, 123–131. West Hartford: Kumarian Press.
- Hedrick-Wong, Yuwa, Bosse Kramsjo, and Asgar Ali Sabri. 2001. Experiences and Challenges in Credit & Poverty Alleviation Programmes in Bangladesh: The Case of Proshika. In *Who Needs Credit? Poverty and Finance in Bangladesh* ed. Geoffrey D. Wood, and Iffath A. Sharif, 145–169. Dhaka: University Press Ltd.
- Heider, John. n.d. Enlightened Leadership is Service, not Selfishness. http://www.johnbaldoni.com/download/leadership_chpt1.pdf. Accessed 5 March 2008.
- Hendry, Chris, James Brown, and Robert Deflippi. 2000. Regional Clustering of High Technology-based Firms: Opto-electronics in Three Countries. *Regional Studies* 34 (2): 129–144. doi:10.1080/00343400050006050.
- Hoppers, Catherine Odora. 2004. Indigenous Knowledge Systems: An Invisible Resource in Literacy Education. http://www.transcend.org/t_database/articles.php?ida=281. Accessed 25 August 2005.
- Horton, Douglas, Anastasia Alexaki, Samual Bennett-Lartey, Kim Noële Brice, Dindo Campilan, Fred Carden, et al. 2003. *Evaluating Capacity Development-Experiences from Research and Development Organizations around the World*. Canada: ISNAR/IDRC/CTA.
- Hossain, Abul. 2006. The Changing Local Rural Power Structure: The Elite and NGOs in Bangladesh. *Journal of Health Management* 8 (2): 229–250. doi:10.1177/097206340600800205.
- Howell, Phillipa. 2003. Indigenous Early Warning Indicators of Cyclones: Potential Application of Coastal Bangladesh. http://www.benfieldhrc.org/disaster_stu dies/working_papers/pdfs/workingpaper6.pdf Accessed 23 December 2005
- Huda, Karishma, Sabeel Rahman, and Catherine Guirguis. 2005. Is Social Capital the Missing Link? An Exploratory Study of Gram Shahayak Committee Effectiveness. Dhaka: Bangladesh Rural Advancement Committee (BRAC).
- Hulme, David. 2000. Protecting and Strengthening Social Capital in Order to Produce Desirable Development Outcome. Social Development Systems for Coordinated Poverty Eradication, SD Scope Paper no. 4. Bath: Centre for Development Studies, University of Bath.
- Hulme, David, and Michael Edwards. 1997. NGOs, States and Donors: Too Close for Comfort. Basingstoke/New York: Macmillan/St. Martin's Press.
- Jor Comfort. Basingstoke/ New York: Machinal/ St. Martin's Press.
 Hulme, David, and Karen Moore. 2006. Why Has Micro-finance been a Policy Success in Bangladesh (and Beyond)? Working paper series 041. Manchester: Global Policy Research Group, IDPM, University of Manchester.
 Hunter, Boyd. 2004. Taming the Social Capital in Hydra? Indigenous Poverty, Social Capital Theory and Measurement. Discussion paper no. 261/2004. Canberra, Centre for Aboriginal Economic Policy Research, Australian National University.
- Ingamells, Ann. 2006. Community Development and Community Renewal: Tracing the Workings of Power. *Community Development Journal* 42 (2): 237-250. doi:10.1093/cdj/bsi111.
- International Labour Organization (ILO). 2004. A Fair Globalization Creating Opportunities for All. Geneva: World Commission on the Social Dimension on Globalization, ILO.
- International NGOs training and Research Centre (INTRAC). 2005. Civil society and community development, INTRAC Conference, Amman (Jordan), 18–20 April. Islam, A. K. M. Aminul. 1974. A Bangladesh Village Conflict and Cohesion: An
- Anthropological Study of Politics. Cambridge, Massachusetts (USA): Schenkman Publishing Company.
- Islam, M. Rezaul. 2014a. Improving Development Ownership among the Vulnerable People: Challenges of NGOs' Community Empowerment Projects in Bangladesh. Asian Social Work and Policy Review 8 (3): 193–209. doi:10.1111/aswp.12035.
- Islam, M. Rezaul. 2014b. NGOs' Role for Social Capital and Community Empowerment in Community Development: Experience from Bangladesh. *Asian Social Work and Policy Review* 8 (3): 261–274. doi:10.1111/aswp.12039.
 Islam, M. Rezaul. 2015. NGOs' social capital development practice for social welfare in Bangladesh. *Global Social Welfare* 2(4): 167–176.
- Islam, M. Rezaul 2015a. Non-governmental Organizations and Community Development in Bangladesh. International Social Work. doi:10.1177/ 0020872815574133.
- Islam, M. Rezaul 2015b. NGOs' Social Capital Development Practice for Social Welfare in Bangladesh. Global Social Welfare 2 (4): 167-176. doi:10.1007/ s40609-015-0034-0.
- Islam, M. Rezaul, and Ahmadullah Mia. 2007. The Innovative Elements in Nonformal Education of Bangladesh: Perspective of Income Generating Programmes for Poverty Alleviation. International Journal of Education and Development using Information and Communication Technology (IJEDICT) 3 (3): 89–104.

- Islam, M. Rezaul, Ahmadullah Mia and Nihar Ranjan Sorcar. 2005. Income Generation Perspective in Non-formal Education: NGO Initiatives in Bangladesh. Asia Pacific Journal of Social Work and Development 15 (1): 17–29. doi:10.1080/21650993.2005.9755966.
- Islam, M. Rezaul and William John Morgan. 2012a. Non-governmental Organizations in Bangladesh: Their Contribution to Social Capital Development and Community Empowerment. *Community Development Journal* 47 (3): 369–385. doi:10.1093/cdj/bsr024.
- Islam, M. Rezaul and William John Morgan. 2012b. Agents of Community Empowerment? The Possibilities and Limitations of Non-governmental Organizations in Bangladesh. *Journal of Community Positive Practices* 12 (4): 703–725.
- Islam, M. Rezaul, Siti Hajar Aabu Bakar, and Haris Abd. Wahab. 2013. Local Knowledge in the Lips of Globalization: Uncertainties of Community Participation in NGO Activities. *Revista De Cercetare Si Interventie Sociala* 43: 7–23.
- Jackson, Edward T. 2005. Social Enterprise: Blending Business with Social Justice-What's Possible? Victoria: Faculty of Business, University of Victoria.
- Jackson, Kenneth E., and Tazul Islam. 2005. Regulation of Microfinance NGOs: General Reflections and the Case of Bangladesh. *International Journal of Rural Management* 1 (1), 45–57. doi:10.1177/097306800400100103.
- Jad, Islah. 2007. NGOs: Between Buzzwords and Social Movements. Development in Practice 17 (4–5): 622–629. doi:10.1080/09614520701469781.
- Johnson, Martha. 1992. Lore: Capturing Traditional Environmental Knowledge. Ottawa: Dene Cultural Institute/IDRC).
- Johnson, Susan, and Ben Rogaly. 1997. Microfinance and Poverty Reduction. London: Oxfam.
- Jones, A., and Glenn Laverack. 2003. Building Capable Communities within a Sustainable Livelihoods Approach: Experiences from Central Asia. http://www.livelihoods.org/lessons/docs/kyrgyz_SLLPC.doc (last accessed June 2, 2007).
- Kabeer, Naila. 2005. Is Microfinance a 'Magic Bullet' for Women's Empowerment? Analysis of Findings from South Asia. *Economic and Political Weekly* 29: 4709–4718.
- Kabeer, Naila. 2003. Making rights work for the poor: Nijera Kori and the construction of —collective capabilities in rural Bangladesh, IDS working paper 200. Brighton: Institute of Development Studies (IDS).
- Kahler, David. 2000. Linking Non-formal Education to Development: NGO Experience During the Education for All Decade. Boston: World Education.
- Kamat, Sangeeta. 2004. The Privatization of Public Interest: Theorizing NGO Discourse in a Neo-liberal Era. *Review of International Political Economy* 11 (1): 155–176. doi:10.1080/0969229042000179794.

- Kamata, Yoji. 2000. Indigenous Knowledge, Cultural Empowerment and Alternatives. http://www.adf.jp/main/eng/eng-pdf/5-2(1)IK%20Paper~1. pdf. Accessed 12 November 2005.
- Kay, Alan. 2005. Social Capital, the Social Economy and Community Development. *Community Development Journal* 41 (2): 160–173. doi:10.1093/cdj/bsi045.
- Kelly, Katharine, and Tullio Caputo. 2005. Case Study of Grassroots Community Development: Sustainable, flexible and cost-effective responses to local needs. *Community Development Journal* 41 (2): 234–245. doi:10.1093/cdj/bsi052.
- Keystone. 2006. Downward Accountability to 'Beneficiaries': NGO and donor Perspectives. http://www.keystoneaccountability.org/files/Keystone% 20Survey%20Apr%2006%20Final%20Report.pdf. Accessed 11 May 2008.
- Khan, Shahrukh Rafi. 2006. Learning from South Asian 'Successes': Tapping Social Capital. *South Asia Economic Journal* 7 (2): 157–178. doi:10.1177/139156140600700201.
- Khan, Zarina Rahman. 2001. Decentralized Governance: Trials and Triumphs. In *Bangladesh: Promise and Performance* ed. Rounak Jahan, 107–129. London: Zed Books.
- Khan, Niaz Ahmed, Sukanta Sen, and M. Millat-e Mustafa. 2000. A Primer on the Documentation of Indigenous Knowledge in Bangladesh: The BARCIK's Experience. In *Of Popular Wisdom: Indigenous Knowledge as Practices in Bangladesh* ed. Niaz Ahmed Khan, 15–22. Dhaka: Resource Centre for Indigenous Knowledge (BARCIK).
- Khandker, Shahidur R. 1998. Fighting Poverty with Micro-credit: Experience in Bangladesh. New York: Oxford University Press for the World Bank.
- Khandker, Shahidur R., Hussain A. Samad, and Zahed H. Khan. 1998. Income and Employment Effects of Micro-credit Programmes: Village-level Evidence from Bangladesh. *The Journal of Development Studies* 35: 96–124. doi:10.1080/00220389808422566.
- Kilby, Patrick. 2006. Accountability for Empowerment: Dilemmas Facing Nongovernmental Organizations. *World Development* 34 (6): 951–963. doi:10.1016/j.worlddev.2005.11.009.
- Kilpatrick, Sue, John Field, and Ian Falk. 2001. Social Capital: An Analytical Tool for Exploring Lifelong Learning and Community development. *British Educational Research Journal* 29 (3): 417–432. doi:10.1080/01411920301859.
- Knack, Stephen, and Philip Keefer. 1997. Does Social Capital have an Economic Payoff? A Cross-country Investigation. *The Quarterly Journal of Economics* 112 (4): 1251–1288. doi:10.1162/003355300555475.
- Kooy, Racelle, and Yvonne Vizina. 2006. Indigenous Knowledge. "In Focus." Gatineau, QC: Canadian Cultural Observatory.

- Krishna, Anirudh, and Elizabeth Shrader. 1999. "Social Capital Assessment Tool." Prepared for *Conference on Social Capital and Poverty Alleviation*, 22–24 June. Washington, D.C.: World Bank.
- Kusumahadi, Meth. 2002. Practical Challenges to the Community Empowerment Program Experiences of Satunama Foundation of Yogyakarta Indonesia. http://www.satunama.org/upl/article/en_Meth.Kusumahadi%5C's% 20PRESENTATION.pdf. Accessed 8 January 2016.
- Larbi, George A. 1999. *The New Public Management Approach and Crisis States.* Geneva: United Nations Research Institute for Social Development (UNRISD).
- Layder, Derek. 1987. Key Issues in Structuration Theory. Current Perspectives in Social Theory 8: 25-46.
- Laverack, Glenn. 2006. Using a 'Domains' Approach to Build Community Empowerment. *Community Development Journal* 41 (1): 4–12. doi:10.1093/cdj/bsi038.
- Lee, Julian. 2004. NGO Accountability: Rights and Responsibilities, Programme on NGOs and Civil Society. Geneva: CASIN.
- Leen, Maureen. 2003. The Challenges and Opportunities for International Civil Society in Promoting Ethical Globalisation. Development Studies Association (DSA) Conference on Globalisation and Development, Glasgow (UK), University of Strathclyde, 10–12 September.
- Lewis, David. 2007. Bringing in Society, Culture and Politics: Values and Accountability in Bangladeshi NGO. In *Global Accountabilities: Participation, Pluralism, and Public Ethics* ed. Alnoor Ebrahim, and Edward Weisband, 131–148. Cambridge: Cambridge University Press.
- Lewis, David. 2008. Crossing the boundaries between 'third sector' and state: Life-work histories from the Phillipines, Bangladesh and the UK. *Third World Quarterly* 29(1): 125–141.
- Lewis, David, Dennis Rodgers, and Michael Woolcock. 2008. The fiction of development: Literary representation as a source of authoritative knowledge. *The Journal of Development Studies* 44(2): 198–216.
- Lewis, David, and Najnin Kanji. 2009. Non-governmental Organizations and Development. London: Routledge
- Lin, Nan. 2000. Inequality in social capital. *Contemporary Sociology* 29(6): 785-795.
- Lord, John, and Peggy Hutchison. 1997. Empowerment, Disability and the Community Context. *Rehabilitation Digest* 27 (2): 13–15.
- Ludden, David. 1997. The News of Globalization. http://www.sas.upenn.edu/~ dludden/global2.htm. Accessed 27 August 2005.
- Lyons, Michael, Carin Smuts, and Anthea Stephens, A. 2001. Participation, Empowerment and Sustainability: (How) Do the Links Work? *Urban Studies* 8: 1233–1251. doi:10.1080/00420980120061007.

- Mahmud, Wahid. 2008. Pathways of Social Development in Bangladesh: Surprises and Challenges. Paper presented at the International Conference on Understanding of Social Protection in Asia, held in New Delhi on February 17–20, 2008, jointly organized by The Institute of Social Studies, The Hague, and the Institute of Human Development, New Delhi.
- Mahmud, Wahid, Sadiq Ahmed, and Sandeep Mahajan, S. 2008. Economic Reforms, Growth, and the Governance: The Political Economy Aspects of Bangladesh's Development Surprise. *Working paper 57722*. Washington, D.C.: World Bank.
- Makita, Rie. 2009. New NGO-Elite Relations in Business Development for the Poor in Rural Bangladesh. *Voluntas* 20: 50–70. doi:10.1007/s11266-008-9077-5.
- Malecki, Edward J. 1998. How Development Occurs: Local Knowledge, Social Capital, and Institutional Embeddedness. Paper prepared for presentation at the Meeting of the Southern Regional Science Association, Savannah, Georgia, April.
- Malki, M. 2006. Community development approaches: a concept note. In Community empowerment for sustaianable development in Iran, ed. Hamyaran, 50–59. Fifth Mena Development Forum. Beirut, Lebanon: Iran NGO Resource Centre.
- Mannel, Jeneviève, and Arvid Chowdhury. 2005. An Approach to Communication and Social Change: How Communication Can Build Social Capital for the Ultra Poor. *Working paper*. Dhaka: Bangladesh Rural Advancement Committee (BRAC).
- Mathias, Evelyn. 1995. Framework for Enhancing the Use of Indigenous Knowledge. *IK Monitor* 3 (2).
- McClenaghan, Pauline. 2000. Social Capital Exploring the Theoretical Foundations of Community Education. *British Educational Research Journal* 26 (5): 565–582. doi:10.1080/713651581.
- McCleary, Rachel. M., and Robert J. Barro. 2008. Private Voluntary Organizations Engaged in International Assistance, 1939–2004. Nonprofit and Voluntary Sector Quarterly Journal 37 (3): 512–536. doi:10.1177/ 0899764007313719.
- McNeil, Mary, and Michael Woolcock. 2004. Capacity Enhancement for Social Development: A Strategy for Building on Local Context and Process. Prepared for CESI Technical Review Meeting, Paris, 25–26 February. Washington, D.C.: World Bank.
- Micro-credit Summit. 1997. Position Paper prepared by UNESCO (CAB-97/WS/2), Sheraton Washington Hotel Washington, D.C. 4 February. http://www.unesco.org/most/povmicr2.htm. Accessed 22 April 2007.
- Mir, Monir, and Swapan Kumar Bala. 2015. NGO accountability in Bangladesh: Two contrasting cases. VOLUNTAS: International Journal of Voluntary and Nonprofit Organizations 26(5): 1831–1851.

- Mittelman, James H. 2000. The globalization syndrome: transformation and resistance. Princeton: Princeton University Press.
- Mohan, Giles, and Kristian Stokke. 2000. Participatory Development and Empowerment: The Dangers of Localism. *Third World Quarterly* 21 (2): 247–268.
- Molla, Rafiqul Islam, M. Mahmudul Alam, and Abu N. M. Wahid. 2008. Questioning Bangladesh's Microcredit. *Challenge* 51 (6): 113–121.
- Mondal, Abdul Hye. 2000. Social Capital Formation: The Role of NGO Rural Development Programs in Bangladesh. *Policy Sciences* 33 (3 & 4): 459–475. doi:10.1023/A:1004825609376.
- Morgan, William John. 1987. Considering Antonio Gramsci. *Teaching Politics* 16 (3): 301–307.
- Morgan, William John. 2005. Local Knowledge and Globalization: Are They Compatible. In *Globalization, Education and Culture Shock* ed. Cedric Cullingford, and Stan Gunn, 35–47. London: Ashgate.
- Morgan, William John, and A. C. Tujman. 2000. The challenges of adult education in Europe deconstruction and reconstruction of nation and state. In *Language, Mobility: Contemporary Issues for Adult Education in Europe* ed. Agnieszka Bron and Michael Schemmann, 47–62. Germany, Lit Verlag /London, Global.
- Moser, Caroline O. 1996. Confronting Crisis: A Summary of Household Responses to Poverty and Vulnerability in Four Poor Urban Communities. Environmentally Sustainable Development and Monographs series no. 7. Washington, D.C.: World Bank.
- Mostashari, Ali. 2005. An Introduction to Non-governmental Organizations Management. Iranian Study Group MIT. http://web.mit.edu/isg/ NGOManagement.pdf. Accessed 11 September 2008.
- Murtaza, Niaz. 2012. Putting the Lasts First: The Case for Community-focused and Peer-managed NGO Accountability Mechanisms. VOLUNTUS: International Journal of Voluntary and Nonprofit Organizations 23: 109–125.
- Nahapiet, Janine, and Sumantra Ghoshal. 1998. Social Capital, Intellectual Capital, and the Organizational Advantage. *Academy of Management Review* 23 (2): 242–262.
- Narayan, Deepa. 2002. Empowerment and Poverty Reduction: A Source Book. Washington, D.C.: World Bank.
- Narayan, Deepa, and Michael F. Cassidy. 2001. A Dimensional Approach to Measuring Social Capital: Development and Validation of a Social Capital Inventory. *Current Sociology* 49 (2): 59–102. doi:10.1177/ 0011392101049002006.
- Narayan, Deepa, and Michael Woolcock. 2000. Social Capital: Implications for Development Theory, Research and Policy. *The World Bank Research Observer* 15 (2): 225–249.

- National Statistics. 2005. *Measuring Social Capital*. London: Office for National Statistics.
- Nawaz, Shah. 2004. An Evaluation of Micro-credit as a Strategy to Reduce Poverty: A Case Study of Three Micro-credit Programs in Bangladesh. In *Development on* the Edge ed. Kenneth Jackson, N. Lewis, S. Adams, and M. Mortein, 170–1175. The Fourth Biennial Conference of the Aotearoa New Zealand, 3–5 December.
- Nel, Etienne, Tony Binns, and Nicole Motteux. 2001. Community-based Development, Non-governmental Organizations and Social Capital in Postapartheid South Africa. *Human Geography* 83 (1): 3–13. doi:10.1111/j.0435-3684.2001.00086.x.
- Nelson, Paul. 2002. New Agendas and New Patterns of International NGO Political Action. Voluntas: International Journal of Voluntary and Non-Profit Organanization13 (4): 377–392. doi:10.1023/A:1022062010375.
- Newaz, Ware. 2003. Impact of micro-credit programs of two local NGOs on rural women's lives in Bangladesh. Academic dissertation. Finland: Faculty of Economics and Administration, University of Tampere.
- Newton, Kenneth. 2001. Trust, Social Capital, Civil Society, and Democracy. International Political Science Review 22: 201–214.
- NGO Affairs Bureau (NGOAB). 2009. List of NGOs in Bangladesh as of July 2009. http://www.ngoab.gov.bd/. Accessed 8 August 2009. NGO News (2015). NGOs' list of Bangladesh. http://ngonewsbd.com/ngo-list-
- NGO News (2015). NGOs' list of Bangladesh. http://ngonewsbd.com/ngo-listof-bangladesh/. Accessed 6 January 2016.
- NORAD. 2004. SWAps and Civil Society—the Role of Civil Society Organisations in Zambia's Basic Education Sub-Sector Investment Programme. Oslo, Norway: NORAD.
- Nunnenkamp, Peter, and Hannes Ohler. 2012. How to Attract Donations: The Case of US NGOs in International Development. *Journal of Development Studies* 48 (10): 1522–1535. doi:10.1080/00220388.2012.685720.
- Ohmae, Kenichi. 1990. The Borderless World. London: William Collins.
- Ohmae, Kenichi. 1995. The end of the nation state: the rise of regional economies. New York: Free Press.
- Olate, Rene. 2003. Local Institutions, Social Capital and Capabilities: Challenges for Development and Social Interventions in Latin America. Washington D.C.: Washington University in St Louis.
- Onyx, Jenny, and Paul Bullen. 2000. Measuring Social Capital in Five Countries. The Journal of Applied Behaviour Science 36 (1): 23-42. doi:10.1177/ 0021886300361002.
- Ostrom, Elinor, and T. K. Ahn. 2003. Introduction. In *Foundations of Social Capital* ed. Elinor Ostrom, and T. K. Ahn, xi-xxxix. Cheltenham (UK): Edward Elgar.
- Ottone, Ernesto. 1996. Globalization and Educational Change: Modernism and Citizenship *Prospects*, 26 (2): 231–240. doi:10.1007/BF02195503.

- Paldam, Martin. 2000. Social Capital: One or Many? Definition and Measurement. Journal of Economic Surveys 14: 629–653. doi:10.1111/1467-6419.00127.
- Petras, James. 1997. Imperialism and NGOs in Latin America. http://www.monthlyreview.org/1297petr.htm. Accessed 22 August 2008.
- Phillips, Sarah D. 2005. Will the market set them free? Women, NGOs, and social enterprise in Ukraine. *Human Organization* 64(3): 251–264.
- Portes, Alejandro. 1998. Social Capital: Its Origins and Applications in Modern Sociology. *Annual Review of Sociology* 24: 1–24.
- Portes, Alejandro, and Partica Landolt. 1996. The Downside of Social Capital. *The American Prospect* 26: 18–22.
- Poston, David. 1994. The Blacksmith and the Farmer Rural manufacturing in Sub-Saharan Africa. Brighton: Intermediate Technology Publications.
- Practical Action Bangladesh (PAB). 2005. The Human Resource Policy Manual. Dhaka: PAB.
- Proshika. 2007. Activity Report July 2006-June 2007. Dhaka: Proshika.
- Proshika. 2008. Activity report 2008. Dhaka: Proshika.
- Purvez, Md. Salim Ahmed. 2003. Making Use of Mediating Resources: Social Network of the Extreme Poor in Rural Bangladesh. Dhaka: Proshika.
- Purvez, Md. Salim Ahmed. 2005. Building Support for a Living: The Importance of Social Networks for the Livelihoods of the Poor. In *Making a Living: The Livelihoods of the Rural Poor in Bangladesh* ed. Iqbal Alam Khan, and Janet Seeley, 93–107. Dhaka: University Press Ltd.
- Putnam, Robert. 1993. The Prosperous Community: Social Capital and Public Life. *The American Prospect* 4 (Spring): 35–42.
- Putnam, Robert. 1994. Social Capital and Public Affairs. Bulletin of the American Academy of Arts and Sciences 47 (8): 5–19.
- Putnam, Robert. 1995. Bowling Alone: America's Declining Social Capital. Journal of Democracy 6 (1): 65–78.
- Putnam, Robert. 1998. Foreword for Volume on Social Capital. *Housing Policy Debate* 9 (1): v-viii.
- Putnam, Robert. 2000a. Bowling Alone-the Collapse and Revival of American Community. New York: Simon & Schuster.
- Putnam, Robert. 2000b. Social Capital: Measurement and Consequences. Isuma: Canadian Journal of Policy Research 2 (Spring): 41–51.
- Radcliff, Benjamin. 2001. Politics, Markets, and Life Satisfaction: The Political Economy of Human Happiness. *American Political Science Review* 95 (4): 939–952.
- Rahman, Sabeel. 2006. Development, Democracy and the NGO Sector Theory and Evidence from Bangladesh. *Journal of Developing Societies* 22 (4): 451–473. doi:10.1177/0169796X06072650.
- Rhea, Zane Ma. 2000. Facing the Challenges of Globalisation: Equity, Justice and Diversity. Paper prepared for the Globalisation Theme of the We the People's

Forum, held at the United Nations on 21–26 May. http://www.wilpf.int.ch/economicjustice/globalisation2000.htm. Accessed 7 September 2008.

- Riddell, Roger C. 2007. *Does Foreign Aid Really Work?* Oxford: Oxford University Press.
- Rifkin, Susan B. 2003. A Framework Linking Community Empowerment and Health Equity: It is a matter of CHOICE. *Journal of Health Population Nutrition* 21 (3): 168–180.
- Risal, Subas. 2014. Mismatch between NGO Services and Beneficiaries' Priorities: Examining Contextual Realities. *Development in Practice* 24 (7): 883–896. doi:10.1080/09614524.2014.950190.
- Roberts, Susan M., John Paul Jones, and Oliver Fröhling. 2005. NGOs and the Globalization of Managerialism. *World Development* 33 (11): 1845–1864. doi:10.1016/j.worlddev.2005.07.004.
- Robinson, Guy M. 1997. Community-based Planning: Canada's Atlantic Action Programme (ACAP). *Geographical Journal* 163: 25–37.
- Roue, Marie. 2003. NGOs, Indigenous Peoples and Local knowledge: Issues of Power in the Arena of Biodiversity. *International Social Science Journal* 55 (178): 533–537. doi:10.1111/j.0020-8701.2003.05504002.x.
- Sampson, Steven. 1996. The social life of projects: importing civil society to Albania. In *Civil Society: Challenging Western Models*, eds. Chris Hann and Elizabeth Dunn. London: Routledge.
- Scholte, Jan Aart. 2003. Protecting the rights and addressing the responsibilities of non-governmental organisations. Unpublished paper, Warwick University, Coventry, UK.
- Schuller, Mark. 2007. Seeing Like a 'Failed' NGO: Globalization-s Impacts on State and Civil Society in Haiti. Political and Legal. *Anthropology Review* 30 (1): 67–89.
- Schuller, Tom. 2000. Thinking about Social Capital. http://www.immagic.com/ eLibrary/ARCHIVES/GENERAL/OPENU_UK/O000519S.pdf. Accessed 10 January 2016.
- Schumacher, Ernst Friedrich. 1973. Small is beautiful: economics as if people mattered. London: Harper and Row.
- Scoones, Ian, and John Thompson. 1994. Knowledge, Power and Agriculture towards a Theoretical Understanding. In *Beyond Farmer First* ed. Ian Scoones, and John Thompson, 6–31. London: Intermediate Technology Publications.
- Seeland, Klaus. 2000. Recent Developments in Social and Community Forestry in India, Nepal and Bhutan. In *Community Forestry- A Change for the Better* ed. Forestry Commission/The Countryside Agency. London: Forestry Commission/Landscape Agency.

- Sen, Amartya. 1999. Development as Freedom. Oxford: Oxford University Press.
- Sen, Amartya. 2002. How to Judge Globalism, the American Prospect. http://www.prospect.org/print/V13/1/sen-a.html. Accessed 12 July 2005.
- Sen, Amartya. 2003. The Importance of Basic Education. Speech to the Commonwealth Education Conference, Edinburgh, 28 October.
- Sen, Binayak, Mustafa K. Mujeri, and Quazi Shahabuddin. 2004. Operationazing Pro-poor Growth: Bangladesh as a Case Study. http://siteresources.world bank.org/INTPGI/Resources/342674-1115051237044/oppgbangladesh (Nov).pdf. Accessed 12 April 2009.
- Sillitoe, Paul. 2000. Give Me More Cake: Indigenous Knowledge, Science and the Poorest of the Poor. Bangladesh Resource Centre for Indigenous Knowledge 2nd National Workshop on Documentation and Application of Indigenous Knowledge, Dhaka, 17–18 March.
- Sillitoe, Paul. 2002. Globalizing Indigenous Knowledge. In *Participating in Development: Approaches to Indigenous Knowledge* ed. Alan Bicker, John Pottier, and Paul Sillitoe, 108–138, London: Routledge.
- Small Economic Enterprise Development (SEED). 2006. SEED Programme (Draft Phase Seven). Dhaka: Proshika.
- Stephens, Sidney. 2000. *Handbook for Culturally Responsive Science Curriculum*. Fairbanks: Alaska Native Knowledge Network.
- Stimson, Robert, John Western, Scott Baum and Yolanda Van Gellecum. 2003. Measuring Community Strength and Social Capital. The European Regional Science Association ERSA 2003 Congress Jyväskylä, Finland August 27–30.
- Stubbs, Paul. 2006. Community Development in Contemporary Croatia: Globalization, Neo-liberalisation and NGOI-isation.: In *Revitalising Communities in a Globalising World* ed. Lena Dominelli, 161–174. London: Ashgate.
- Svendsen, Gunnar Lind Haase, and Gert Tinngard Svendsen. 2004. Creation and Destruction of Social Capital: Entrepreneurship, Co-operative Movements and Institutions. Cheltenham (UK): Edward Elgar.
- Tembo, Fletcher. 2004. NGDOs' Role in Building Poor People's Capacity to Benefit from Globalization. Journal of International Development 16: 1023–1037. doi:10.1002/jid.1155.
- Thaman, Konai Helu. 2003. Decolonizing Pacific Studies: Indigenous Perspectives, Knowledge and Wisdom in Higher Education. *The Contemporary Pacific* 15 (1): 1–17.
- Thornton, Paul, Joe Devine, Peter Houtzager, Deborah Wright, and Santi Rozario. 2000. Partners in Development: A Review of Big NGOs in Bangladesh. Dhaka: DFID, Bangladesh.

- Transparency International Bangladesh (TIB). 2007. Problems in Good Governance in the NGO Sector: The Way Forward. Presented in a discussion meeting at CIRDAP Auditorium, Dhaka, Bangladesh, 6 October.Turner, Mark, and David Hulme. 1997. Governance, Administration and
- Turner, Mark, and David Hulme. 1997. Governance, Administration and Development: Making the State Work. London: Palgrave Macmillan.
 Ulleberg, Inger. 2009. The role and impact of NGOs in capacity development from
- Ulleberg, Inger. 2009. The role and impact of NGOs in capacity development from replacing the state to reinvigorating education. Paris: International Institute for Educational Planning, UNESCO.
- Ulvila, Marko, and Farhad Hossain. 2002. Development NGOs and Political Participation of the Poor in Bangladesh and Nepal. Voluntas: International Journal of Voluntary and Nonprofit Organizations 13 (2): 149–63. doi:10.1023/A:1016055705828.
- United Nations Development Programme (UNDP). 1999. Human Development Report 1999 Globalization with a Human Face. New York: UNDP.
- United Nations Development Programme (UNDP). 2006. Strategic Programme Framework 2006-2010, Democratic Governance and Human Rights Cluster. Dhaka: UNDP Bangladesh.
- United Nations. 2004. Bangladesh-. 2004b. *Meeting the Challenges in an Era of Globalization by Strengthening Regional Development Cooperation*. New York: United Nations.
- VanSant, Jerry. 2003. Challenges of local NGO sustainability. Keynote remarks prepared for the USAID/PVC-ASHA Annual PVO Conference, 14 October.
- Vargas, Claudia Maria. 2001. Community Development and Micro-enterprises: Fostering Sustainable Development. Sustainable Development 8 (1): 11-26.
- Vitásková, Anna, and Petr Mateju. 2006. Interpersonal trust and mutually beneficial exchanges: measuring social capital for comparative analyses. *Sociologický časopis/Czech Sociological Review* 42(03): 493–516.
- Vivian, Jessica. 1994. NGOs and Sustainable Development in Zimbabwe: No Magic Bullets. *Development and Change* 25 (1): 167–193. doi:10.1111/ j.1467-7660.1994.tb00513.x.
- Wallace, Tina, and Jenny Chapman. 2003. Is the Way Aid is Disbursed through NGOs Promoting a Development Practice that Addresses Chronic Poverty Well? An Overview of an On-going Research Project. Presented at Staying Poor: Chronic Poverty and Development Policy, International Conference, IDPM, University of Manchester, 7–9 April.
- Warren, Michael. 1996. Indigenous Knowledge, Biodiversity Conservation, and Development. In Sustainable Development in Third World Countries Applied and Theoretical Perspective ed. Valentine Udoh James, 81-88. London: Prager.

- Westergaard, Kristen, and Abul Hossain. 2005. Boringram Revisited: Persistent Power Structure and Agricultural Growth in a Bangladesh Village. Dhaka: University Press Ltd.
- Willetts, Peter. 2002. What is a Non-Governmental Organization? Research Project on Civil Society Networks in Global Governance, City University, London.
- Wood, Geoffrey D. 1997. Who Needs Credit? In *Who Needs Credit?: Poverty and Finance in Bangladesh* ed. Geoffrey Wood and Ifath Sharif, Dhaka: University Press Ltd. & London, Zed Press.
- Wood, Geoffrey D. 2003. Staying Secure, Staying Poor: The Faustian Bargain. World Development 31 (3): 455–473.
- Wood, Geoffrey D. 2005. Introduction: Poverty, Capabilities and Perverse Social Capital: The Antidote to Sen and Putnam? In *Making a Living the Livelihoods of the Rural Poor in Bangladesh* ed. Iqbal Alam Khan and Janet Seeley, 1–18. Dhaka: University Press Ltd.
- Wood, Geoffrey D. 2007. Clashing Values in Bangladesh: NGOs, Secularism and Ummah. *WeD working Paper 31*. ESRS Research Group on Wellbeing in Developing Countries. Bath: University of Bath.
- Woolcock, Michael. 2001. The Place of Social Capital in Understanding Social and Economic Outcomes. *Isuma* 2 (1): 11–17.
- World Bank. 2005. Simplified Implementation Report. Washington D.C.: World Bank.
- World Bank. 2006. Economic and Governance of Non-governmental Organizations in Bangladesh. Dhaka: World Bank Office.
- World Bank. 2008. Whispers to Voices: Gender and Social Transformation in Bangladesh. Dhaka: The World Bank Dhaka Office.
- Wuyts, Marc, Maureen Mackintosh, and Tom Hewitt. 1992. Development policy and public action. London: Oxford University Press in Association with the Open University.
- Yildiz, Nancy, Yusuf Kassam, Catharina van Heel, and Robert Mitchell. 2003. Annual Review 2003, Proshika Kendra, Phase-VI, Social Programme. Dhaka: Proshika.
- Zain, Mohamed, and Siew Imm Ng. 2006. The Impacts of Network Relationships on SME's Internationalization Process. *Thunderbird International Business Review* 48 (2): 183–205. doi:10.1002/tic.20092.
- Zug, Sebastian. 2006. Monga—Seasonal Food Insecurity in Bangladesh-Bringing the Information Together. *The Journal of Social Studies* 111: 21–39.

INDEX

A

Accountability, 2, 5, 17–19, 32–35, 61, 62, 68, 78, 96–97, 101, 109–113, 120, 123, 131n1 Agent, 2, 4, 5, 13, 28, 36, 45, 62, 77, 81, 96, 98–100, 117, 119, 129

С

Capacity, 3, 4, 5, 25, 31, 48, 55, 61-63, 65-68, 71-85, 90-91, 93, 95, 98, 101–103, 109, 110, 112, 119, 122, 125, 127, 129, 130 problem assessment, 62, 85, 90-91, 129, 130 Collective action, 1, 2, 4, 5, 55, 59, 60, 62, 68, 73-75, 78, 79-81, 83-85, 88, 89, 117, 125, 130 Communication, 24, 25, 40, 41, 47, 73, 75–79, 83, 91, 95, 98–100, 126, 131n2 Community participation, 62, 68, 88-89, 129 Conventional role, 109–113 Critical awareness, 62, 98–100, 103, 129

D

Dependency, 18, 32, 41, 61, 63, 74, 103, 120 donor, 5, 35, 36, 65, 85, 101, 103, 107, 120–123, 126 Development community, 8, 33, 35, 47, 48, 52–54, 61, 66, 68–69, 84, 95, 118, 128 debate, 2, 4, 22, 37, 59–61 local, 4, 22, 37, 43–45, 48, 63, 68 ownership, 18, 26 rural, 25, 26, 66, 94

E

Empowerment community, 2–5, 18, 48, 51–69, 87–103, 120 political, 26 Exposure visit, 78, 89, 94 External intervenors, 120–123

F

Foreign donation, 2, 31, 126, 131n2

© The Author(s) 2016 M.R. Islam, NGOs, Social Capital and Community Empowerment in Bangladesh, DOI 10.1007/978-981-10-1747-6

G

Globalisation, 2–4, 37, 39–49, 63–65, 69, 110, 122 Group federation, 76, 85n2, 92 Group work, 91–93

Η

Holistic approach, 30, 95–98, 103 Human capital, 25, 27, 56, 57, 94, 113

I

Income generation programme, 34, 73, 81

K

Knowledge debate, 47, 112 global, 22, 37, 46–48, 65, 78, 91, 99, 108 indigenous, 36, 37, 39–49, 65 local, 42, 64, 66, 79, 108, 113 sharing, 98–100, 103 traditional, 47

L

Leadership, 13, 28, 31, 62, 76, 83, 85, 91–93, 117, 118, 125, 126, 130, 132n6 Local decision-making, 14, 17, 31, 34, 48, 76, 83, 101, 102, 124, 128 Local level planning, 48, 120 Local voice, 43

Μ

Management autonomous, 100–103 centralised, 103 decentralised, 5, 100–103, 122, 126–127, 130 Market channel, 79–81, 84, 92, 125 Market map, 73, 79–81, 84 Micro-credit, 17, 31, 34–35, 61, 72, 91, 96, 99, 109, 112, 123 Micro-finance, 30, 31, 112, 126 Monolithic approach, 106, 111 Mutual benefits, 2, 4, 55, 72–73, 75–80, 84, 85, 118

N

NGO non-profit organization, 8–9, 95 voluntary organization, 8–9, 17, 31

0

Organisational autonomy, 120–123

Р

Paradigm, 4, 5, 22, 28, 37, 40, 46, 48, 63-69,72 development, 4, 22, 46, 72 Participatory planning, 73-75 Participatory rural appraisal, 73, 90, 91,100 Partnership, 16, 18, 27, 33, 34, 67, 78, 92, 94–96, 99, 103, 117, 120-122, 127 Patron-client relationships, 99, 102, 104n4 Poverty, 1, 5, 14, 16, 18–19, 25–30, 33-37, 42, 43, 46, 60, 67, 79, 81, 82, 100, 102, 108, 111, 112, 121 Power relations, 53, 66, 75, 82-83, 85n1, 99, 110, 120, 128 Power structure, 19, 30, 103, 106, 116, 117, 124-126

R

Rapid market survey, 90 Relation matrix, 79–81, 90 Religious leaders, 106, 118, 123

S

Scientific tools, 90–91 Social capital, 1–5, 17, 46, 48, 51–69, 71–85, 93, 105–132 measurement of, 57, 127 Social enterprise, 95–98, 103, 118 Social network, 54, 56, 60, 64, 68, 73, 78, 79, 83, 88, 89, 91, 92, 93–96, 103, 115–117, 124, 125, 128 Social scheme, 97 Social trust, 2, 4, 73, 81–84, 91, 106, 117–119, 120, 121, 125 Sustainable development, 10, 33, 43, 63–66, 72, 81, 88, 112, 131n2 Sustainable livelihood framework, 81, 91 SWOT, 91

U

Underdeveloped countries, 3, 7

v

Village federations, 85n2